© 2013

David Yu-Ber Wang

ALL RIGHTS RESERVED

## **OXIDATIVE ADDITION TO IRIDIUM(I) COMPLEXES**

by

## **DAVID YU-BER WANG**

A dissertation submitted to the

Graduate School-New Brunswick

Rutgers, The State University of New Jersey

In partial fulfillment of the requirements for the degree of

Doctor of Philosophy

In

Chemistry and Chemical Biology

Written under the direction of

Alan S. Goldman

And approved by

New Brunswick, New Jersey

January 2013

# ABSTRACT OF THE DISSERTATION OXIDATIVE ADDITION TO IRIDIUM(I) COMPLEXES By

**DAVID YU-BER WANG** 

**Dissertation Director:** 

Alan S. Goldman

Understanding the fundamental processes underlying the activation of small molecules by metal complexes is a continuing goal of organometallic chemistry. This thesis aims to better understand the processes and interactions of small molecules with organometallic complexes.

DFT calculations were performed on an extensive series of  $(PH_3)_2IrX$  complexes to investigate the thermodynamics of C-H and N-H oxidative addition. Electronic substituent effects were decomposed into their  $\sigma$  and  $\pi$  effects and individually parameterized; the resulting  $\sigma_X$  and  $\pi_X$  parameters were used to fit the energies of oxidative addition of methane to a series of 3- and 4-coordinate Ir(I) complexes. Regression analysis indicates that oxidative addition of methane to 3-coordinate Ir(I) complexes generally is favored by a less  $\sigma$ -donating ligand *trans* to the vacant coordination site.

DFT calculation was used to investigate the oxidative addition of chlorobenzene and fluorobenzene to pincer iridium complexes. Activation of the aryl C-H bonds

ii

proximal to the electron-withdrawing atom is favored. The kinetic barriers were calculated and correlate with the relative steric interactions in the transition state leading to the oxidative addition product.

CO stretching frequencies for a series of Ni(CO)<sub>3</sub>L complexes were calculated using DFT calculations. NBO analysis of these complexes yielded parameters that reflect the extent of backbonding in the  $\pi^*$  orbitals of the CO ligands and polarization of the  $\pi^b$ orbitals of the CO ligands. The effects of a point charge on free CO were calculated.

Experimentally, reactions of small molecules with pincer iridium complexes were investigated. The kinetic barriers for olefin dissociation from (pincer)Ir(olefin) complexes were measured. The (<sup>tBu</sup>POCOP)Ir-catalyzed olefin isomerization of 1-octene was investigated and shows an apparent 2<sup>nd</sup> order dependence on 1-octene concentration.

A synthesis scheme for  ${}^{tBu3Me}PCP$  ligand was developed. While H<sub>2</sub> did not add to  ${}^{tBu4}PCPIr(CO)$ , three isomers were observed when H<sub>2</sub> was added to  ${}^{tBu3Me}PCPIr(CO)$ , and the isomers were assigned using an isotopic labeling experiment and NMR spectroscopy.

The thermodynamic and kinetic products of tropone addition to  $({}^{tBu}PCP)IrH_2$  were synthesized and characterized. The formation of a tropone dimer was investigated.

The product of methyl fluoride addition to (<sup>tBu</sup>PCP)IrH<sub>2</sub> was characterized through NMR spectroscopy, LIFDI mass spectrometry, and elemental analysis.

#### Acknowledgement

I wish to express my sincere thanks to my advisor, Professor Alan S. Goldman, for his help and guidance for the work that led to this dissertation. Alan has been unfailingly supportive during my time at Rutgers and has helped establish a strong mechanistic foundation to my chemical knowledge.

I would like to thank Professor Karsten Krogh-Jespersen, who served on my committee, for his advice, time, and support throughout my years at Rutgers. My computational work would not have been possible without his help.

I would like to thank Professor Kai Hultzsch and Professor Robert Knowles for their valuable time and support in serving on my thesis committee. I would also like to thank Dr. Tom Emge for his assistance with X-ray crystallographic analysis, and Dr. Nagarajan Murali for his assistance with NMR techniques.

I am thankful to my former and current Goldman group colleagues for making our lab a supportive, helpful, and generally happy place to work; it has been a great pleasure to work with them. Among my group members, I would particulary like to thank Dr. Jongwook Choi, Dr. Sabuj Kundu, Dr. Soumik Biswas, Michael Haibach, David Laviska, Yuriy Choliy, and Dr. Xiawei Zhang for their helpful discussions and work that contributed to this thesis.

Finally my deepest gratitude goes to my fiancée Amy and my family – my parents and my brother – who have unfailingly supported me with care and love throughout the most difficult times of graduate school.

## Dedication

To Amy, Mom, Dad, and Frank

## **Table of Contents**

Abstract		ii
Acknowledg	jement	iv
Dedication		V
Table of Co	ntents	vi
List of Table	es	Х
List of Sche	mes	xiii
List of Figu	res	XV
Chapter 1	Introduction to the Thesis	1
1.2	Research Goals of This Thesis	10
	References	13
Chapter 2	A Computational and Experimental Study of Electronic	17
	Factors Affecting 'Oxidative Addition' of C-H and N-H	
	Bonds to Iridium Complexes	
2.1	Introduction	19
2.2	Computational Methods	22
2.3	Parameterization of Electronic Substituent Effects	23
2.4	Oxidative Addition of CH <sub>4</sub> to 3-Coordinate <i>trans</i> -(PH <sub>3</sub> ) <sub>2</sub> IrX	28
2.5	Orbital Rationale	35
2.6	Experimental Verification	40

2.7	Oxidative Addition of Methane to 4-Coordinate Ir(I)	45
	Complexes	
2.8	Oxidative Addition of Ammonia to 3-Coordinate	47
	Ir(I) Complexes	
2.9	Summary	51
2.10	Acknowledgments	52
	References	53
Chapter 3	DFT Computational Study of the Oxidative Addition of	56
	Chlorobenzene and Fluorobenzene to <sup>R</sup> PCPIr	
	Complexes	
3.1	Introduction	57
3.2	Basis Set Determination	58
3.3	Results	60
	3.3.1 Chlorobenzene Activation	60
	3.3.2 Fluorobenzene Activation	65
	3.3.3 Comparison to Experimental Data	67
3.4	Summary	67
	References	68
Chapter 4	Investigation of Bonding and Through-Space Effects on	70
	CO Stretching Frequencies	
4.1	Introduction	71
4.2	Computational Methods	72
	4.2.1 Computational Details	72

	4.2.2 Natural Bond Orbital Analysis	73
4.3	Calculation of Selected Tolman-type Complexes	74
4.4	Perturbation of CO with a Point Charge	81
4.5	Summary	83
	References	84
Chapter 5	Kinetic Study of ( <sup>tBu</sup> PCP)Ir and ( <sup>tBu</sup> POCOP)Ir	86
	Complexes: Olefin Coordination and Isomerization	
5.1	Introduction	87
5.2	Results and Discussion	91
	5.2.1 Kinetics of Olefin Dissociation from	91
	(pincer)Ir(alkene) Complexes	
	5.2.2 ( <sup>tBu</sup> POCOP)Ir-Catalyzed Olefin Isomerization of	95
	1-Octene	
5.3	Summary	98
5.4	Experimental	99
	5.4.1 General Considerations.	99
	5.4.2 EXSY Experiments.	99
	5.4.3 Catalytic 1-Octene Isomerization.	100
	References	101
Chapter 6	Investigations of the ( <sup>tBu3Me</sup> PCP)Ir(CO) Complex:	102
	Synthesis and Oxidative Addition of H <sub>2</sub>	
6.1	Introduction	103
6.2	Revised Synthesis of <sup>tBu3Me</sup> PCPIrH <sub>2</sub>	104

6.3	Addition of H <sub>2</sub> to <sup>tBu3Me</sup> PCPIrCO	106
6.4	Discussion	110
6.5	Summary	111
6.6	Experimental	111
	References	145
Chapter 7	Addition of Tropone to ( <sup>tBu</sup> PCP)IrH <sub>2</sub> : Formation of a	146
	Dimerization Product	
7.1	Introduction	147
7.2	Previous Investigations: Reactions of Ketones with	148
	( <sup>tBu</sup> PCP)Ir	
7.3	Results and Discussion	150
7.4	Summary	153
7.5	Experimental	154
	References	156
Chapter 8	Synthesis and Characterization of ( <sup>tBu</sup> PCP)Ir(Me)(F)	157
8.1	Introduction	158
8.2	Results and Discussion	159
8.3	Conclusion	162
8.4	Experimental	162
	References	168

## List of Tables

Table 2.1	Calculated Electronic Substituent Parameters ( $\sigma_X$ , $\pi_X$ )	26
Table 2.2	Effect of the Ir-CO Bond Distance on the $CO(\pi^*)$ Orbital	27
	Occupancy of <b>2-F</b>	
Table 2.3	Calculated Reaction Energies of CH <sub>4</sub> Addition to (PH <sub>3</sub> ) <sub>2</sub> IrX	30
	for Different Equatorial Plane Geometries	
Table 2.4	Regression Equations Using Restricted Geometries for 3-	33
	$Y_{Me}$ , 3- $Y_X$ , and 3- $T_{transPH3}$	
Table 2.5	Regression Equations Using Restricted Geometries for (a)	34
	<b>3-</b> $T_{X-apical}$ and (b) <b>3-</b> $T_{cisPH3}$	
Table 2.6	Calculated $\sigma_X$ and $\pi_X$ Parameters for Complexes 4-7	42
Table 2.7	Calculated Energies of Benzene and Methane Oxidative	43
	Addition to Complexes 4-7	
Table 2.8	Calculated Kinetic Parameters for Reductive Cleavage of	44
	Benzene for Complexes 4-7	
Table 2.9	Experimentally Determined Kinetic Parameters for the	45
	Reductive Elimination of Benzene for Complexes 4-7	
Table 2.10	Regression Equations Based on Constrained Geometries	46
Table 2.11	Calculated $\Delta E_X$ for NH <sub>3</sub> Oxidative Addition and	49
	Coordination to (PH <sub>3</sub> ) <sub>2</sub> IrX	

Table 2.12	Regression Equations for the Oxidative Addition and	50
	Coordination of Ammonia to trans-(PH <sub>3</sub> ) <sub>2</sub> IrX Using	
	Constrained Geometries	
Table 3.1	Basis Set Combinations Used for DFT Calculations	58
Table 3.2	Calculated Parameters for the Oxidative Addition of	59
	Chlorobenzene to <sup>tBu</sup> PCPIr to Yield the ortho, trans-Isomer	
Table 3.3	Calculated $\Delta G$ for the Various Complexes in the Reaction	62
	Coordinate for the Oxidative Addition of Chlorobenzene to	
	<sup>tBu</sup> PCPIr, <sup>Me</sup> PCPIr, and <sup>H</sup> PCPIr	
Table 3.4	Calculated $\Delta G$ for the Various Complexes in the	66
	Reaction Coordinate for the Oxidative Addition of	
	Fluorobenzene to <sup>tBu</sup> PCPIr, <sup>Me</sup> PCPIr, and <sup>H</sup> PCPIr	
Table 4.1	Calculated and Experimental $v_{CO}$ Values and NBO	75
	Analysis for Ni(CO) <sub>3</sub> L Complexes	
Table 4.2	NBO Data for Selected Ni(CO) <sub>3</sub> L Complexes	78
Table 6.1	Crystal Data and Structure Refinement for 2	119
Table 6.2	Atomic Coordinates and Equivalent Isotropic Displacement	120
	Parameters for 2	
Table 6.3	Bond Lengths (Å) and Angles (°) for <b>2</b>	121
Table 6.4	Anisotropic Displacement Parameters for 2	123
Table 6.5	Hydrogen Coordinates and Isotropic Displacement	124
	Parameters for 2	

Table 6.6	Torsion Angles (°) for <b>2</b>	126
Table 6.7	Crystal Data and Structure Refinement for 3	129
Table 6.8	Atomic Coordinates and Equivalent Isotropic Displacement	130
	Parameters for <b>3</b>	
Table 6.9	Bond Lengths (Å) and Angles (°) for $\bf 3$	131
Table 6.10	Anisotropic Displacement Parameters for <b>3</b>	134
Table 6.11	Hydrogen Coordinates and Isotropic Displacement	135
	Parameters for <b>3</b>	
Table 6.12	Torsion Angles (°) for <b>3</b>	136
Table 6.13	Crystal Data and Structure Refinement for 4	138
Table 6.14	Atomic Coordinates and Equivalent Isotropic Displacement	139
	Parameters for 4	
Table 6.15	Bond Lengths (Å) and Angles (°) for 4	140
Table 6.16	Anisotropic Displacement Parameters for 4	142
Table 6.17	Hydrogen Coordinates and Isotropic Displacement	143
	Parameters for 4	
Table 6.18	Torsion Angles (°) for <b>4</b>	144

## List of Schemes

Scheme 1.1	Mechanistic aspects of $H_2$ addition to Wilkinson's catalyst	2
Scheme 1.2	Mechanism of Photochemical Catalytic Cyclohexane	5
	Dehydrogenation by RhCl(PMe <sub>3</sub> ) <sub>2</sub> CO	
Scheme 1.3	Mechanism of Alkane Metathesis	8
Scheme 1.4	Oxidative Addition of Various Bonds to PCP Pincer	9
	Iridium Complexes	
Scheme 1.5	Oxidative Addition of C-O and C-F Bonds Through an	9
	Initial C-H Oxidative Addition Step	
Scheme 2.1	(a) Trans Influence of an X Ligand on the $M-L_T$ Distance.	24
	(b) $\pi$ -Back-Bonding of Electron Density from a Metal to	
	Vacant $\pi^*$ Antibonding Orbitals of CO	
Scheme 2.2	Orientation of $sd^2$ -Hybridized Orbitals and $d_{xy}$ in 3-	37
	$T_{transPH3}$ , 3- $Y_X$ , and 3- $Y_{Me}$	
Scheme 2.3	Several Ir(III) Phenyl Hydride Complexes and the Relative	41
	$\sigma$ -Donating Abilities of the Ligand at the Position <i>trans</i> to	
	the Phenyl Moiety	
Scheme 2.4	Formation and Subsequent Reactivity of (PCP)Ir Amido	48
	Hydride Complexes	
Scheme 3.1	Oxidative Addition of Chlorobenzene to <sup>tBu</sup> PCPIr to Yield	59
	the <i>ortho</i> , <i>trans</i> -Isomer	

Scheme 3.2	Representative Reaction Coordinate Diagram for the	61
	Oxidative Addition of Chlorobenzene to <sup>tBu</sup> PCPIr	
Scheme 5.1	Schematic Illustration of the Two Mechanism Classes	87
	Proposed for Olefin Isomerization: (a) "Hydride"	
	Mechanism, (b) $\pi$ -Allyl Mechanism	
Scheme 5.2	Tandem Catalytic Alkane Metathesis with Pincer Iridium	89
	Complexes and Olefin Metathesis Catalysts	
Scheme 5.3	Schematic $\pi$ -Allyl Pathway for 1-Alkene Isomerization	90
	with Dissociative Olefin Exchange	
Scheme 5.4	Alternative $\pi$ -Allyl Pathway for 1-Alkene Isomerization,	97
	Associative Displacement of 2-Alkene by 1-Alkene.	
Scheme 6.1	Revised Synthesis Scheme for $^{tBu3Me}PCPIrH_2(8)$	106
Scheme 7.1	Reaction of 3-Pentanone with 1	148
Scheme 7.2	Reaction of Cyclohexanone and 3-Methylcyclohexanone	149
	with 1	
Scheme 7.3	Addition of Cycloheptanone to 1	150
Scheme 7.4	Possible Mechanism for Tropone Dimerization	152
Scheme 7.5	Addition of 1-Iodopropane to 7	153
Scheme 8.1	General Reaction of Oxidative Addition and Reductive	158
	Elimination	
Scheme 8.2	Oxidative Addition of $CH_3F$ to Complex 1	160
Scheme 8.3	Proposed Mechanism for Oxidative Addition of $CH_3F$ to	162
	(PCP)Ir.	

# List of Figures

Figure 1.1	Protonated <sup>tBu</sup> PCP ligand	6
Figure 1.2	Synthesized variations of the PCP ligand, shown	7
	coordinated to Ir	
Figure 1.3	Various pincer ligand motifs	7
Figure 2.1	Complexes used for the determination of electronic	25
	substituent effect parameters ( $\sigma_X$ , $\pi_X$ )	
Figure 2.2	Three general geometries appropriate for trans-	31
	$(PH_3)_2 Ir(X)(H)(CH_3)$	
Figure 2.3.	2-Variable linear regression fit of calculated $\Delta E_X$ to $\sigma_X$ and	32
	$\pi_X$ parameters for the constrained, lowest energy	
	conformers of CH <sub>4</sub> addition to <i>trans</i> -(PH <sub>3</sub> ) <sub>2</sub> IrX	
Figure 2.4	The HOMO(-8) orbital of <i>trans</i> - $(PH_3)_2$ Ir(F)(H)(Me) in the	39
	$3-T_{transPH3}$ geometry showing how polarization of the metal	
	sd <sup>2</sup> -hybridized orbital may create favorable overlap with	
	the $\pi$ -system of ligand X	
Figure 3.1	Isomers of the oxidative addition of chlorobenzene to	60
	PCPIr	
Figure 3.2	Abbreviated DFT-optimized geometries of (a) ortho, trans-	63
	Cl and (b) ortho, cis-Cl complexes	
Figure 3.3	Transition state interactions leading to (a) the ortho, cis-	64
	isomer and (b) the ortho, trans-isomer	

Figure 4.1	Resonance forms of CO	71
Figure 4.2	Plot of calculated $v_{CO}$ vs. average CO $\pi^*$ orbital occupancy	76
	for Ni(CO) <sub>3</sub> L complexes	
Figure 4.3	Plot of calculated $v_{CO}$ vs. average percentage of CO $\pi^b$	77
	orbital from the C atom for Ni(CO) <sub>3</sub> L complexes	
Figure 4.4	Two-variable linear regression fit of calculated CO	79
	stretching frequency to polarization and backbonding	
	parameters for for Ni(CO) <sub>3</sub> L complexes	
Figure 4.5	Plot of average percentage of CO $\pi^{b}$ orbital from the C	81
	atom vs. average CO $\pi^*$ orbital occupancy for Ni(CO) <sub>3</sub> L	
	complexes	
Figure 4.6	Plot of calculated $v_{CO}$ vs. the average percentage of the CO	82
	$\pi^{b}$ orbitals lying on the carbon atom for free CO in the	
	presence of a point charge.	
Figure 5.1	Catalytic isomerization of 1-octene to internal octenes by 1	90
	(5 mM) at 125 °C in <i>n</i> -octane solvent.	
Figure 5.2	Eyring plot for the rate of dissociative 1-hexene exchange	92
	of ( <sup>tBu</sup> PCP)Ir(1-hexene) with free 1-hexene in <i>p</i> -xylene- $d_{10}$	
	in the temperature interval 20 °C to 46 °C	
Figure 5.3	Eyring plot for the rate of dissociative trans-2-hexene	93
	exchange of ( <sup>tBu</sup> PCP)Ir( <i>trans</i> -2-hexene) with free trans-2-	
	hexene in mesitylene- $d_{12}$ in the temperature interval -17 °C	
	to 6 °C	

Figure 5.4	Eyring plot for the rate of dissociative 1-hexene exchange	94
	of ( <sup><math>^{IBu}</math></sup> POCOP)Ir(1-hexene) with free 1-hexene in <i>p</i> -xylene-	
	$d_{10}$ in the temperature interval 107 °C to 126 °C	
Figure 5.5	Eyring plot for the rate of dissociative trans-2-hexene	95
	exchange of ( <sup>tBu</sup> POCOP)Ir( <i>trans</i> -2-hexene) with free trans-	
	2-hexene in <i>p</i> -xylene- $d_{10}$ in the temperature interval 57 °C	
	to 79 °C	
Figure 5.6	Catalytic isomerization of 1-octene to internal octenes by $2$	96
	(5 mM) at 125 °C	
Figure 5.7	2 <sup>nd</sup> order plot of the catalytic isomerization of 1-octene to	98
	internal octenes by 2 (5 mM) at 125 °C	
Figure 6.1	Isomers of <sup>tBu3Me</sup> PCPIr(CO)(H) <sub>2</sub> complexes	107
Figure 6.2	<sup>1</sup> H NMR spectrum of H <sub>2</sub> addition to <sup>tBu3Me</sup> PCPIr(CO)	108
	and <sup>tBu3Me</sup> PCPIr( <sup>13</sup> CO)	
Figure 6.3	1D NOESY spectra of <sup>tBu3Me</sup> PCPIr(CO) with the selective	109
	excitation of hydride (a) $H_a$ and (b) $H_b$	
Figure 6.4	ORTEP diagram of compound <b>2</b>	118
Figure 6.5	ORTEP diagram of compound <b>3</b>	128
Figure 6.6	ORTEP diagram of compound 4	137
Figure 7.1	Crystal structures of CO adduct 9 and tropone dimer 10	150
Figure 8.1	<sup>1</sup> H NMR signal of <b>2</b> (Ir-CH <sub>3</sub> ) with no decoupling and with	164
	<sup>31</sup> P decoupling	

Figure 8.2	$^{19}$ F NMR signal of <b>2</b> without decoupling (red) and with $^{31}$ P	165
	decoupling (black)	
Figure 8.3	<sup>19</sup> F NMR signal of <b>2</b> : simulated with Spinworks software	165
	and observed NMR data	
Figure 8.4	$^{19}$ F{ $^{31}$ P} NMR signal of <b>2</b> : simulated with Spinworks	166
	software and observed NMR data	
Figure 8.5	<sup>31</sup> P NMR signal of <sup>13</sup> C-labeled <b>2</b> , synthesized from the	166
	reaction with <sup>13</sup> CH <sub>3</sub> F, simulated with Spinworks software	
	and observed NMR data	
Figure 8.6	LIFDI-MS spectrum of <b>2</b>	167

XV111	
27.4.111	

## Chapter 1

# Introduction to the Thesis

Selective bond activation using metal complexes has been a fundamental transformation of organometallic chemistry and is vital to many metal-catalyzed transformations. These bond activation processes typically involve oxidative addition to a metal center. Oxidative addition with a mononuclear metal complex implies a formal two electron oxidation of the metal and involves the cleavage of an X-Y bond with the concomitant formation of M-X and M-Y bonds.<sup>1</sup> Given its fundamental role in bond activation, many reports have focused on oxidative addition and its microscopic reverse, reductive elimination, in the activation and functionalization of small molecules.<sup>2-12</sup>

The history of activation of small molecules by metals extends more than sixty years. In 1955, Halpern reported the catalytic hydrogenation of dichromic anion  $(Cr_2O_7^{2^-})$  by catalytic Cu(II) and proposed in subsequent work that Cu(II) heterolytically cleaves the H<sub>2</sub> bond to give  $[Cu-H]^+$  and H<sup>+</sup> (eq. 1).<sup>13,14</sup> In 1962, Vaska reported the oxidative addition of H<sub>2</sub> to his eponymous complex, *trans*-(PPh<sub>3</sub>)<sub>2</sub>Ir(Cl)(CO)(PPh<sub>3</sub>)<sub>2</sub> (eq. 2),<sup>15</sup> and the following year he reported its use in the catalytic hydrogenation of ethylene and acetylene.<sup>16</sup> The mechanism and utility of the addition of H<sub>2</sub> to Vaska's complex has been extensively studied.<sup>17-28</sup> In 1965, Wilkinson reported the oxidative addition of H<sub>2</sub> to RhCl(PPh<sub>3</sub>)<sub>3</sub> (Wilkinson's catalyst) for the catalytic hydrogenation of olefins and acetylenes.<sup>29</sup> Extensive investigations into the mechanistic aspects of the oxidative

addition of  $H_2$  to Wilkinson's catalyst showed that addition of  $H_2$  occurs predominantly after initial dissociation of a PPh<sub>3</sub> ligand to give the highly active 14-electron complex RhCl(PPh<sub>3</sub>)<sub>2</sub> (Scheme 1.1).<sup>30,31</sup>

$$Cu^{2+} + H_2 \iff [Cu-H]^+ + H^+$$
(1)

$$\begin{array}{c} PPh_{3} & PPh_{3} \\ | \\ Cl - lr - CO + H_{2} & \hline Cl - lr - H \\ | \\ PPh_{3} & OC & PPh_{3} \end{array}$$

$$(2)$$

Scheme 1.1. Mechanistic aspects of H<sub>2</sub> addition to Wilkinson's catalyst.<sup>30,31</sup>



The activation of C-H bonds by transition metal complexes was first reported by Chatt and Davidson in 1965, where  $Ru(0)(dmpe)_2$  (dmpe = 1,2bis(dimethylphosphino)ethane) was found to activate a C-H bond of naphthalene to yield a naphthyl hydride complex (eq. 3).<sup>32</sup> Similarly, Green and co-workers reported the oxidative addition of benzene following the photolysis of H<sub>2</sub> from Cp<sub>2</sub>WH<sub>2</sub> (eq. 4).<sup>33-35</sup>



While the oxidative addition of alkanes was thought to be more difficult to achieve than arenes despite the lower C-H bond strength of alkanes, Bergman<sup>36</sup> and Graham<sup>37</sup> independently reported the isolation and characterization of iridium alkyl hydride complexes in 1982. Bergman reported that the irradiation of Cp\*Ir(PMe<sub>3</sub>)(H)<sub>2</sub> in cyclohexane and neopentane led to loss of H<sub>2</sub> and formation of the respective alkyl hydride complexes (eq. 5).<sup>36</sup> Graham reported the formation of a neopentyl hydride complex following photolytic loss of CO from Cp\*Ir(CO)<sub>2</sub> (eq. 6).<sup>37</sup> In 1984, Jones reported arene and alkane oxidative addition to Rh analogous to Bergman's work.<sup>38</sup>



Investigations into the selectivity of alkane oxidative addition found a surprising thermodynamic preference for activation of the stronger C-H bonds of alkanes. Independent investigations by Bergman,<sup>39</sup> Bercaw,<sup>40</sup> and Jones<sup>41</sup> showed a clear preference for C-H activation of alkanes at the strongest C-H bonds (i.e.  $1^{\circ} > 2^{\circ} > 3^{\circ}$ ) despite widely differing metal-ligand complexes. By contrast, hydrogen atom abstraction by radical reactions occurs preferentially at the more substituted position.<sup>42</sup> While the selectivity of alkane oxidative addition to metal complexes was initially ascribed in part to steric effects,<sup>39</sup> more recent work have shown that metal-X bond strength correlates with H-X bond strength, implying an electronic basis for selectivity.<sup>40,41,43,49</sup>

Oxidative addition of C-H bonds is particularly important in catalytic alkane dehydrogenation. While dehydrogenation is an enthalpically uphill process ( $\Delta H \sim 30$  kcal/mol),<sup>50,51</sup> the process is positive entropically. As a result, some systems under elevated temperatures can overcome the enthalpic barrier and achieve catalytic turnover. To achieve catalytic dehydrogenation under less forcing condition, the dehydrogenation reaction can be coupled with the thermodynamically downhill hydrogenation reaction of a sacrificial hydrogen acceptor, often *tert*-butylethylene (tbe) or norbornene. In 1979, Crabtree reported the first example of transfer dehydrogenation in which [IrH<sub>2</sub>(acetone)<sub>2</sub>(PPh<sub>3</sub>)<sub>2</sub>] [BF<sub>4</sub>] reacted with cycloalkanes in conjunction with tbe to yield cyclic diene iridium complexes (eq. 7).<sup>52</sup> In 1987, Crabtree reported catalytic alkane dehydrogenation and acceptorless photochemical conditions.<sup>53</sup> In 1989, our group reported the first example of efficient alkane functionalization, the photochemical dehydrogenation of alkanes catalyzed by RhCl(PMe<sub>3</sub>)<sub>2</sub>(CO) (Scheme 1.2).<sup>54,55</sup>



Scheme 1.2. Mechanism of photochemical catalytic cyclohexane dehydrogenation by RhCl(PMe<sub>3</sub>)<sub>2</sub>CO.<sup>54,55</sup>



More recently, many groups have investigated the use of pincer ligand metal complexes to effect reactions with small molecules. The "pincer" ligand was first introduced by Moulton and Shaw in 1976 to describe complexes bearing  $\kappa^3$ -2,6- (<sup>t</sup>Bu<sub>2</sub>PCH<sub>2</sub>)<sub>2</sub>C<sub>6</sub>H<sub>3</sub> (<sup>tBu</sup>PCP), a meridionally-bound, benzene-based, tridentate ligand (Figure 1.1).<sup>56</sup> Since then, the term "pincer" has been extended to any meridionally-bound tridentate ligand, and many different pincer ligand motifs have been reported.<sup>57-63</sup>

Figure 1.1. Protonated <sup>tBu</sup>PCP ligand.



Iridium<sup>64-66</sup> and rhodium<sup>67</sup> pincer complexes were investigated as catalysts toward alkane dehydrogenation. While the rhodium pincer complexes displayed low activity for catalytic alkane dehydrogenation,<sup>67</sup> iridium pincer complexes (<sup>tBu</sup>PCP)IrH<sub>2</sub> and (<sup>iPr</sup>PCP)IrH<sub>2</sub> were found to exhibit high rates and turnover number for both acceptorless and transfer dehydrogenation of cyclic alkanes.<sup>64-66,68</sup> More significantly, regioselective catalytic alkane dehydrogenation was reported by Goldman in 1999 in which (<sup>iPr</sup>PCP)IrH<sub>2</sub> selectively dehydrogenates *n*-alkanes to  $\alpha$ -olefins (eq. 8).<sup>69</sup>



Many modifications have been made to the pincer ligand first introduced by Moulton and Shaw in an effort to improve catalytic dehydrogenation activity and to investigate new reactivity. The phosphine substituents have been modified such that methyl, isopropyl, adamantyl, and trifluromethyl groups have replaced one or more of the t-butyl groups (Figure 1.2).<sup>62,68-73</sup> Substitution of OMe,<sup>74-76</sup> CO<sub>2</sub>Me,<sup>74</sup> and NMe<sub>2</sub><sup>77</sup> groups at the *para* position of the aryl ring of the PCP ligand have been reported (Figure 1.2). Replacement of the methylene linkers with nitrogen<sup>60</sup> or oxygen<sup>57,78,79</sup> atoms has been reported (Figure 1.3). The benzene ring has also been modified to give new ligand motifs such as a pyridyl-PNP ligand,<sup>58,80-82</sup> an amido-PNP ligand,<sup>59,83</sup> and a PSiP ligand (Figure 1.3).<sup>61</sup>

Figure 1.2. Synthesized variations of the PCP ligand, shown coordinated to Ir.



Figure 1.3. Various pincer ligand motifs.



Our group has reported the use of pincer iridium complexes in the activation of a wide variety of small molecules. As noted before in the context of dehydrogenation, the oxidative addition of alkanes has been reported,<sup>69</sup> and this reaction is utilized as part of multistep reactions like alkane metathesis (Scheme 1.3)<sup>84,85</sup> and dehydroaromatization (eq. 9).<sup>86</sup> Activation of other hydrocarbons, arenes<sup>87</sup> and alkynes,<sup>88</sup> was also reported (Scheme 1.4). Bonds other than C-H bonds have also been reported by our group to oxidatively add to pincer iridium complexes. Aniline<sup>89</sup> and ammonia<sup>90</sup> have been reported to undergo N-H oxidative addition (Scheme 1.4). The oxidative addition of C-

 $O^{91}$  and C-F<sup>92</sup> bonds have also been reported, although the mechanism for both reactions proceeds through an initial C-H activation step (Scheme 1.5).



Scheme 1.3. Mechanism of alkane metathesis.



Scheme 1.4. Oxidative addition of various bonds to PCP pincer iridium complexes.



Scheme 1.5. Oxidative addition of C-O and C-F bonds through an initial C-H oxidative addition step.



## 1. 2 Research Goals of This Thesis

Understanding the fundamental processes underlying the activation of small molecules by metal complexes is a continuing goal of organometallic chemistry. This thesis aims to better understand the processes and interactions of small molecules with organometallic complexes. The remaining chapters of this thesis can be organized into two parts. The first part comprises Chapters 2 through 4 and covers work that primarily utilizes DFT calculations. The second part comprises Chapters 5 through 8 and covers my experimental work.

Chapter 2 presents a combined computational and experimental study of electronic factors affecting C-H and N-H bond addition to Ir(I) complexes. DFT calculations were performed on an extensive series of  $(PH_3)_2$ IrX complexes to investigate the thermodynamics of C-H and N-H oxidative addition. Electronic substituent effects were decomposed into their  $\sigma$  and  $\pi$  effects and individually parameterized; the resulting  $\sigma_X$  and  $\pi_X$  parameters were used to fit the energies of oxidative addition of methane to a series of 3- and 4-coordinate Ir(I) complexes. Regression analysis indicates that oxidative addition of methane to 3-coordinate Ir(I) complexes generally is favored by a less  $\sigma$ -donating ligand *trans* to the vacant coordination site, although this preference is reversed in the special case where the geometry of the 5-coordinate product is square pyramidal with the X ligand in the apical position.

Chapter 3 presents a computational study into the oxidative addition of chlorobenzene and fluorobenzene to <sup>H</sup>PCPIr, <sup>Me</sup>PCPIr, and <sup>tBu</sup>PCPIr. The thermodynamics of addition were found to favor activation of the C-H bond proximal to the electron-withdrawing Cl atom ( $\Delta G$ : *para* addition > *meta* addition > *ortho* addition).

The kinetic barriers were also calculated and correlated with the relative steric interactions in the transition state leading to the oxidative addition product. Both calculations for thermodynamic and kinetic preferences were found to be in accord with experimental results.

Chapter 4 describes an investigation into the contributions of backbonding and through-space effects on the stretching frequency of metal-bound CO. CO stretching frequencies for a series of Ni(CO)<sub>3</sub>L complexes were calculated using DFT calculations. NBO analysis of these complexes yielded parameters that reflected backbonding in the  $\pi^*$  orbitals of the CO ligands and polarization of the  $\pi^b$  orbitals of the CO ligands. We found that a linear two-variable model fit the data better than the traditional one-variable model of solely back-bonding effects. A close look at rotamers of anisolyl phosphine complexes highlighted the dominant effect of polarization in some cases. In the course of our investigation, we also calculated the effects of a point charge on free CO, which was consistent with previous work.

Chapter 5 presents experimental work relevant to the mechanism of olefin isomerization. The reaction kinetics of olefin coordination to (<sup>tBu</sup>PCP)Ir and (<sup>tBu</sup>POCOP)Ir were quantified and discussed in the context of olefin isomerization. Additionally, the (<sup>tBu</sup>POCOP)Ir-catalyzed olefin isomerization of 1-octene was investigated and found to show an apparent 2<sup>nd</sup> order dependence on 1-octene concentration.

Chapter 6 describes investigations into the synthesis and reactivity of <sup>tBu3Me</sup>PCPIr(CO). A revised synthesis scheme for the synthesis of the <sup>tBu3Me</sup>PCP ligand was developed. The revised synthesis scheme was found to be amenable for the

synthesis of a variety of unsymmetrical pincer ligands. With the <sup>tBu3Me</sup>PCP ligand, addition of H<sub>2</sub> to <sup>tBu3Me</sup>PCPIr(CO) was investigated. Addition of H<sub>2</sub> was not observed to add to <sup>tBu4</sup>PCPIr(CO). However, when H<sub>2</sub> was added to <sup>tBu3Me</sup>PCPIr(CO), three isomers were observed and assigned using an isotopic labeling experiment and NMR spectroscopy.

Chapter 7 discusses the reactivity of tropone with (<sup>tBu</sup>PCP)IrH<sub>2</sub>. The thermodynamic and kinetic products have been synthesized and characterized. One of the tropone addition adducts was previously found to catalyze the dimerization of tropone to give a fused tricyclic dihydrodicycloheptafuranol; the formation of this compound was investigated.

Chapter 8 details the addition of methyl fluoride's  $C(sp^3)$ -F bond to an iridium center via the initial, reversible cleavage of a C-H bond. The structure was characterized through NMR spectroscopy, LIFDI mass spectrometry, and elemental analysis.

### References

- (1) Hartwig, J. F. Organotransition Metal Chemistry: From Bonding to Catalysis; University Science Books: Sausalito, CA, 2010.
- (2) Johnson, S. A.; Hatnean, J. A.; Doster, M. E. Prog. Inorg. Chem. 2012, 57, 255.
- (3) Ozerov, O. V. Chem. Soc. Rev. 2009, 38, 83.
- (4) Zeni, G.; Larock, R. C. Chem. Rev. 2006, 106, 4644.
- (5) Braun, T. Angew. Chem., Int. Ed. 2005, 44, 5012.
- (6) Deutsch, P. P.; Eisenberg, R. Chem. Rev. **1988**, 88, 1147.
- (7) Robinson, S. D. Organomet. Chem. 1978, 7, 348.
- (8) Stille, J. K.; Lau, K. S. Y. Acc. Chem. Res. 1977, 10, 434.
- (9) Esteruelas, M. A.; Lopez, A. M. Organometallics 2005, 24, 3584.
- (10) Jones, W. D. Acc. Chem. Res. 2003, 36, 140.
- (11) Brown, J. M.; Cooley, N. A. Chem. Rev. 1988, 88, 1031.
- (12) Halpern, J. Acc. Chem. Res. **1982**, 15, 332.
- (13) Halpern, J.; Peters, E. J. Chem. Phys. 1955, 23, 605.
- (14) Halpern, J.; MacGregor, E. R.; Peters, E. J. Phys. Chem. 1956, 60, 1455.
- (15) Vaska, L.; DiLuzio, J. W. J. Am. Chem. Soc. 1962, 84, 679.
- (16) Vaska, L.; Rhodes, R. E. J. Am. Chem. Soc. 1965, 87, 4970.
- (17) James, B. R.; Memon, N. A. Can. J. Chem. 1968, 46, 217.
- (18) Chock, P. B.; Halpern, J. J. Am. Chem. Soc. 1966, 88, 3511.
- (19) Eberhardt, G. G.; Vaska, L. J. Catal. 1967, 8, 183.
- (20) Yamaguchi, M. Kogyo Kagaku Zasshi 1967, 70, 675.
- (21) Bennett, M. A.; Milner, D. L. J. Amer. Chem. Soc. 1969, 91, 6983.
- (22) Strohmeier, W.; Onoda, T. Z. Naturforsch. B 1969, 24, 1493.
- (23) Strohmeier, W.; Onoda, T. Z. Naturforsch. B 1969, 24, 461.
- (24) Strohmeier, W.; Onoda, T. Z. Naturforsch. B 1969, 24, 515.
- (25) Strohmeier, W. J. Organomet. Chem. 1971, 32, 137.
- (26) Strohmeier, W.; Fleischmann, R. J. Organomet. Chem. 1971, 29, C39.
- (27) Strohmeier, W.; Fleischmann, R.; Onoda, T. J. Organomet. Chem. 1971, 28, 281.
- (28) Ugo, R.; Pasini, A.; Fusi, A.; Cenini, S. J. Amer. Chem. Soc. 1972, 94, 7364.
- (29) Young, J. F.; Osborn, J. A.; Jardine, F. H.; Wilkinson, G. Chem. Commun. 1965, 131.
- (30) Halpern, J.; Wong, C. S. J. Chem. Soc., Chem. Commun. 1973, 629.
- (31) Goodman, J.; Grushin, V. V.; Larichev, R. B.; Macgregor, S. A.; Marshall, W. J.; Roe, D. C. J. Am. Chem. Soc. **2010**, 132, 12013.
- (32) Chatt, J.; Davidson, J. M. J. Chem. Soc. 1965, 843.
- (33) Green, M. L. H.; Knowles, P. J. J. Chem. Soc. D 1970, 1677.
- (34) Green, M. L. H.; Knowles, P. J. J. Chem. Soc. A 1971, 1508.
- (35) Giannotti, C.; Green, M. L. H. J. Chem. Soc., Chem. Commun. 1972, 1114.
- (36) Janowicz, A. H.; Bergman, R. G. J. Am. Chem. Soc. 1982, 104, 352.
- (37) Hoyano, J. K.; Graham, W. A. G. J. Am. Chem. Soc. 1982, 104, 3723.
- (38) Jones, W. D.; Feher, F. J. J. Am. Chem. Soc. 1984, 106, 1650.
- (39) Buchanan, J. M.; Stryker, J. M.; Bergman, R. G. J. Am. Chem. Soc. **1986**, 108, 1537.

- (40) Bryndza, H. E.; Fong, L. K.; Paciello, R. A.; Tam, W.; Bercaw, J. E. J. Am. Chem. Soc. 1987, 109, 1444.
- (41) Northcutt, T. O.; Wick, D. D.; Vetter, A. J.; Jones, W. D. J. Am. Chem. Soc. 2001, 123, 7257.
- (42) Hass, H. B.; McBee, E. T.; Weber, P. Ind. Eng. Chem. 1936, 28, 333.
- (43) Bennett, J. L.; Wolczanski, P. T. J. Am. Chem. Soc. 1994, 116, 2179.
- (44) Bennett, J. L.; Wolczanski, P. T. J. Am. Chem. Soc. 1997, 119, 10696.
- (45) Jones, W. D.; Hessell, E. T. J. Am. Chem. Soc. 1993, 115, 554.
- (46) Wick, D. D.; Jones, W. D. Organometallics **1999**, *18*, 495.
- (47) Clot, E.; Besora, M.; Maseras, F.; Megret, C.; Eisenstein, O.; Oelckers, B.; Perutz, R. N. Chem. Commun. 2003, 490.
- (48) Evans, M. E.; Burke, C. L.; Yaibuathes, S.; Clot, E.; Eisenstein, O.; Jones, W. D. J. Am. Chem. Soc. 2009, 131, 13464.
- (49) Choi, G.; Morris, J.; Brennessel, W. W.; Jones, W. D. J. Am. Chem. Soc. 2012, 134, 9276.
- (50) Crabtree, R. H.; Mellea, M. F.; Mihelcic, J. M.; Quirk, J. M. J. Am. Chem. Soc. 1982, 104, 107.
- (51) Afeefy, H. Y.; Liebman, J. F.; Stein, S. E., "Neutral Thermochemical Data" In NIST Chemistry Webbook, NIST Standard Reference Database Number 69; Linstrom, P. J., Mallard, W. G., Eds. Gaithersburg, MD, http://webbook.nist.gov/chemistry/, (retrieved September 28, 2012).
- (52) Crabtree, R. H.; Mihelcic, J. M.; Quirk, J. M. J. Am. Chem. Soc. 1979, 101, 7738.
- (53) Burk, M. J.; Crabtree, R. H. J. Am. Chem. Soc. 1987, 109, 8025.
- (54) Maguire, J. A.; Boese, W. T.; Goldman, A. S. J. Am. Chem. Soc. 1989, 111, 7088.
- (55) Rosini, G. P.; Soubra, S.; Vixamar, M.; Wang, S.; Goldman, A. S. J. Organomet. *Chem.* **1998**, 554, 41.
- (56) Moulton, C. J.; Shaw, B. L. J. Chem. Soc., Dalton Trans. 1976, 1020.
- (57) Göttker-Schnetmann, I.; White, P.; Brookhart, M. J. Am. Chem. Soc. 2004, 126, 1804
- (58) Ben-Ari, E.; Gandelman, M.; Rozenberg, H.; Shimon, L. J. W.; Milstein, D. J. *Am. Chem. Soc.* **2003**, *125*, 4714.
- (59) Fan, L.; Parkin, S.; Ozerov, O. V. J. Am. Chem. Soc. 2005, 127, 16772.
- (60) Benito-Garagorri, D.; Bocokić, V.; Mereiter, K.; Kirchner, K. Organometallics 2006, 25, 3817.
- (61) Morgan, E.; MacLean, D. F.; McDonald, R.; Turculet, L. J. Am. Chem. Soc. 2009, 131, 14234.
- (62) Adams, J. J.; Lau, A.; Arulsamy, N.; Roddick, D. M. *Organometallics* **2011**, *30*, 689.
- (63) Gibson, V. C.; Redshaw, C.; Solan, G. A. Chem. Rev. 2007, 107, 1745.
- (64) Gupta, M.; Hagen, C.; Flesher, R. J.; Kaska, W. C.; Jensen, C. M. Chem. Commun. 1996, 2083.
- (65) Gupta, M.; Hagen, C.; Kaska, W. C.; Cramer, R. E.; Jensen, C. M. J. Am. Chem. Soc. 1997, 119, 840.
- (66) Xu, W.-w.; P. Rosini, G.; Krogh-Jespersen, K.; S. Goldman, A.; Gupta, M.; M. Jensen, C.; C. Kaska, W. *Chem. Commun.* **1997**, 2273.

- (67) Wang, K.; Goldman, M. E.; Emge, T. J.; Goldman, A. S. J. Organomet. Chem. 1996, 518, 55.
- (68) Liu, F.; Goldman, A. S. Chem. Commun. 1999, 655.
- (69) Liu, F.; Pak, E. B.; Singh, B.; Jensen, C. M.; Goldman, A. S. J. Am. Chem. Soc. 1999, 121, 4086.
- Kundu, S.; Choliy, Y.; Zhuo, G.; Ahuja, R.; Emge, T. J.; Warmuth, R.; Brookhart, M.; Krogh-Jespersen, K.; Goldman, A. S. *Organometallics* 2009, 28, 5432.
- (71) Punji, B.; Emge, T. J.; Goldman, A. S. Organometallics 2010, 29, 2702.
- (72) Adams, J. J.; Arulsamy, N.; Roddick, D. M. Organometallics 2011, 30, 697.
- (73) Adams, J. J.; Arulsamy, N.; Roddick, D. M. Organometallics 2012, 31, 1439.
- (74) Krogh-Jespersen, K.; Czerw, M.; Zhu, K.; Singh, B.; Kanzelberger, M.; Darji, N.; Achord, P. D.; Renkema, K. B.; Goldman, A. S. J. Am. Chem. Soc. 2002, 124, 10797.
- (75) Zhu, K.; Achord, P. D.; Zhang, X.; Krogh-Jespersen, K.; Goldman, A. S. J. Am. Chem. Soc. 2004, 126, 13044.
- (76) Ray, A.; Kissin, Y. V.; Zhu, K.; Goldman, A. S.; Cherian, A. E.; Coates, G. W. J. Mol. Catal. A 2006, 256, 200.
- Huang, Z.; Brookhart, M.; Goldman, A. S.; Kundu, S.; Ray, A.; Scott, S. L.;
   Vicente, B. C. Adv. Synth. Catal. 2009, 351, 188.
- (78) Göttker-Schnetmann, I.; Brookhart, M. J. Am. Chem. Soc. 2004, 126, 9330.
- (79) Göttker-Schnetmann, I.; White, P. S.; Brookhart, M. Organometallics **2004**, *23*, 1766.
- (80) Ben-Ari, E.; Cohen, R.; Gandelman, M.; Shimon, L. J. W.; Martin, J. M. L.; Milstein, D. *Organometallics* **2006**, *25*, 3190.
- (81) Ben-Ari, E.; Leitus, G.; Shimon, L. J. W.; Milstein, D. J. Am. Chem. Soc. 2006, 128, 15390.
- (82) Khaskin, E.; Iron, M. A.; Shimon, L. J. W.; Zhang, J.; Milstein, D. J. Am. Chem. Soc. 2010, 132, 8542.
- (83) Zhu, Y.; Fan, L.; Chen, C.-H.; Finnell, S. R.; Foxman, B. M.; Ozerov, O. V. Organometallics 2007, 26, 6701.
- (84) Goldman, A. S.; Roy, A. H.; Huang, Z.; Ahuja, R.; Schinski, W.; Brookhart, M. *Science* **2006**, *312*, 257.
- (85) Haibach, M. C.; Kundu, S.; Brookhart, M.; Goldman, A. S. Acc. Chem. Res. 2012, 45, 947.
- (86) Ahuja, R.; Punji, B.; Findlater, M.; Supplee, C.; Schinski, W.; Brookhart, M.; Goldman, A. S. *Nat. Chem.* **2011**, *3*, 167.
- (87) Zhang, X.; Kanzelberger, M.; Emge, T. J.; Goldman, A. S. J. Am. Chem. Soc. 2004, 126, 13192.
- (88) Ghosh, R.; Zhang, X.; Achord, P.; Emge, T. J.; Krogh-Jespersen, K.; Goldman, A. S. J. Am. Chem. Soc. 2007, 129, 853.
- (89) Kanzelberger, M.; Zhang, X.; Emge, T. J.; Goldman, A. S.; Zhao, J.; Incarvito, C.; Hartwig, J. F. J. Am. Chem. Soc. 2003, 125, 13644.
- (90) Zhao, J.; Goldman, A. S.; Hartwig, J. F. Science 2005, 307, 1080.
- (91) Choi, J.; Choliy, Y.; Zhang, X.; Emge, T. J.; Krogh-Jespersen, K.; Goldman, A. S. *J. Am. Chem. Soc.* **2009**, *131*, 15627.

(92) Choi, J.; Wang, D. Y.; Kundu, S.; Choliy, Y.; Emge, T. J.; Krogh-Jespersen, K.; Goldman, A. S. *Science* **2011**, *332*, 1545.

# Chapter 2

# A Computational and Experimental Study of Electronic Factors Affecting 'Oxidative Addition' of C-H and N-H Bonds to Iridium Complexes

### Abstract

A combined computational and experimental study of electronic factors affecting C-H and N-H bond addition to Ir(I) complexes has been carried out. DFT calculations were performed on an extensive series of (PH<sub>3</sub>)<sub>2</sub>IrX complexes to investigate the thermodynamics of C-H and N-H oxidative addition. Electronic substituent effects were decomposed into their  $\sigma$  and  $\pi$  effects and individually parameterized; the resulting  $\sigma_X$ and  $\pi_X$  parameters were used to fit the energies of oxidative addition of methane to a series of 3- and 4-coordinate Ir(I) complexes. Regression analysis indicates that oxidative addition of methane to 3-coordinate Ir(I) complexes generally is favored by a less  $\sigma$ -donating ligand *trans* to the vacant coordination site, although this preference is reversed in the special case where the geometry of the 5-coordinate product is square pyramidal with the X ligand in the apical position. The oxidative addition of methane to 4-coordinate Ir(I) was found to be analogous in its electronic preferences to the 3coordinate case. Rates of reductive elimination from a series of isoelectronic Ir(III) phenyl hydride complexes were measured and found to be consistent with the computational analysis. The regression analysis was extended to investigate NH<sub>3</sub>
oxidative addition to 3-coordinate Ir(I) complexes. While more  $\sigma$ -donating ligands are unfavorable for N-H oxidative addition (analogous to methane oxidative addition), they are even more unfavorable for NH<sub>3</sub> coordination. An orbital rationale is proposed for the results obtained.

# 2.1 Introduction

Oxidative addition is a fundamental transformation of metal complexes and is a key step in various processes including catalytic dehydrogenation<sup>1</sup> and arylation reactions.<sup>2</sup> Oxidative addition implies a formal two-electron oxidation of a metal center and involves the cleavage of an X-Y bond with concomitant formation of metal-X and metal-Y bonds. As suggested by its name and according to textbooks and conventional wisdom, oxidative addition is commonly thought to be favored by more electron-rich metal centers that can facilitate the formal two-electron loss during this transformation.<sup>3-5</sup> Consequently, there have been reports of electron-donating ligands around a metal center promoting oxidative addition.<sup>6,7</sup> Conversely, there are also reports of the use of electron-withdrawing ligands around a metal center to promote reductive elimination,<sup>8-10</sup> the microscopic reverse of oxidative addition.

The addition of C-H and N-H bonds to metal centers form highly desirable subsets of oxidative addition reactions. Many reports of C-H oxidative addition to latemetal complexes, particularly Ir(I) and Rh(I) complexes, can be found.<sup>11-14</sup> While our group has primarily investigated reactions involving 3-coordinate Ir(I) and Rh(I) complexes,<sup>1</sup> many reports of additions to analogous 4-coordinate complexes covering a wide breadth of ligand motifs have been published.<sup>1,5</sup>

There are only a few recent examples of N-H oxidative addition of ammonia  $(NH_3)$ .<sup>15-20</sup> Among the reports of NH<sub>3</sub> oxidative addition, our group reported that a change in the type of pincer ligand from an aryl-PCP type to an aliphatic-PCP ligand results in a change in thermodynamic selectivity from 4-coordinate NH<sub>3</sub> coordination (aryl-PCP type) to the 5-coordinate amido hydride product (aliphatic-PCP type).<sup>15</sup> This

change in selectivity was attributed to increased electron density on the metal center afforded by the more electron-donating aliphatic-PCP ligand, a hypothesis consistent with the conventional wisdom regarding oxidative addition.

While the concept of oxidative addition being oxidative at the metal seems intuitively appealing, there have been reports that suggest that oxidative addition may not be oxidative at all with respect to the metal and, correspondingly, that electron-donating ligands do not necessarily favor oxidative addition. Saillard and Hoffman cautioned against interpreting the oxidative addition formalism as a literal picture of oxidation at the metal; rather, these authors proposed that a better description of oxidative addition involved the formation of covalent bonds in which electron density in ligand-localized and metal-localized orbitals moves in both ligand-to-metal and metal-to-ligand directions during the course of the reaction.<sup>21</sup> Similarly, investigations by Low and Goddard<sup>22</sup> and Koga and Morokuma<sup>23</sup> concluded that covalency, rather than oxidation, is a better description of oxidative addition products.

Various theoretical calculations of oxidative addition to 3-coordinate Ir(I) and Rh(I) complexes have also suggested that electron-donating ligands do not necessarily favor the reaction. Cundari calculated that the energy of methane oxidative addition to *trans*-(PH<sub>3</sub>)<sub>2</sub>IrX (X = H, Cl) was significantly more favorable for (PH<sub>3</sub>)<sub>2</sub>IrCl ( $\Delta$ H = -41.6 kcal/mol) than (PH<sub>3</sub>)<sub>2</sub>IrH ( $\Delta$ H = -12.8 kcal/mol).<sup>24</sup> Similarly, we reported that oxidative addition of H<sub>2</sub> to *trans*-(PH<sub>3</sub>)<sub>2</sub>MX complexes (X = Cl, Ph, M = Rh, Ir) was significantly more favorable for X = Cl than X = Ph ( $\Delta\Delta$ E = 24.3 kcal/mol and 31.1 kcal/mol for M = Rh and Ir, respectively).<sup>25</sup> In both cases, the more strongly electron-donating ligands (i.e. H<sup>-</sup> and Ph<sup>-</sup>) result in *less* favorable oxidative addition than the weakly donating Cl<sup>-</sup> ligand.

In a substituent effect study on oxidative addition of H-H and C-Y (Y = H, C, and F) bonds to  $(PH_3)_2IrX$  (X = H, F, Cl, and Ph) complexes, we noted that  $\pi$ -donation by the ancillary ligand X favored the addition reaction, whereas  $\sigma$ -donation from X was unfavorable.<sup>26</sup> We concluded that, at least for the limited set of substrates considered, substituent effects in oxidative addition reactions were dominated by individual orbital interactions rather than transfer of charge.

Other groups have reported similar counter-intuitive results in which oxidative addition is promoted by less electron-donating ligands. Su and Chu reported C-F oxidative addition of CH<sub>3</sub>F to (PH<sub>3</sub>)<sub>2</sub>MX (X = Cl, H, Me; M = Rh, Ir);<sup>27</sup> Macgregor reported NH<sub>3</sub> oxidative addition to (PH<sub>3</sub>)<sub>2</sub>MX (X = Cl, H; M = Rh, Ir);<sup>28</sup> and Diggle, Macgregor, and Whittlesey reported the oxidative addition of H<sub>2</sub>, CH<sub>4</sub>, and C<sub>2</sub>H<sub>4</sub> to Ru(CO)(IR)<sub>3-n</sub>(PH<sub>3</sub>)<sub>n</sub> (R = H = 1,3-imidazol-2-ylidene and R = Me = 1,3-dimethylimidazol-2-ylidene).<sup>29</sup> In each case, the less electron-donating ligands were calculated to favor the oxidative addition reaction.

Thus, to date, no comprehensive review of electronic effects on the thermodynamics of oxidative addition has been performed. Here, we report a systematic approach through DFT calculations to probe the electronic factors that influence the thermodynamics of C-H (methane) and N-H (ammonia) oxidative addition to both 3- and 4-coordinate Ir(I) complexes. Computed electronic substituent effects have been decomposed into their  $\sigma$ - and  $\pi$ -components and individually parameterized. The resulting parameters are used to investigate ligand characteristics favorable for the oxidative addition of methane to a series of 3- and 4-coordinate Ir(I) complexes. Rates of benzene reductive elimination from a series of isoelectronic Ir(I) phenyl hydride

complexes have been experimentally determined to verify the conclusions drawn from the electronic structure calculations. The factors that influence the thermodynamics of ammonia oxidative addition were investigated through calculations analogously to methane. Finally, an orbital rationale is proposed for the results obtained.

# **2.2** Computational Methods

All calculations used DFT methodology<sup>30</sup> as implemented in the Gaussian 03 and 09 series of computer programs.<sup>31,32</sup> We employed the Perdew-Burke-Ernzerhof exchange-correlation functional (PBE) in all calculations.<sup>33</sup> For the model complexes (1-3), we applied the SDD relativistic, small-core effective potential and corresponding basis set to the Ir atom<sup>34</sup> and all-electron, valence triple- $\zeta$  plus polarization and diffuse function 6-311+G(d,p) basis sets to all non-metal atoms.<sup>35-38</sup> For the calculations of actual complexes 4-7 and related species, we used the LANL2DZ relativistic, small-core effective potential and the LANL2TZ basis set augmented by a diffuse d-type function (exponent = 0.07645) for Ir<sup>39,40</sup> and 6-311+G(d,p) basis sets for atoms bonded to Ir or contained in substrate molecules (e.g. CH<sub>4</sub>, NH<sub>3</sub>, or benzene); all other atoms were treated with valence double- $\zeta$  plus diffuse function 6-31+G basis sets. All complexes underwent geometry optimization, and stationary points were subjected to normal mode analysis. Electronic population indeces were derived from the Natural Bond Orbital (NBO) analysis scheme developed by Weinhold et al.<sup>41,42</sup> and calculated using NBO 5.G and Gaussian 03.43

#### **2.3 Parameterization of Electronic Substituent Effects**

Electronic substituent effects on oxidative addition of C-H and N-H bonds to Ir complexes were determined from a series of comparative calculations on small model complexes. To investigate solely the electronic substituent effects, we chose to minimize steric contributions by focusing on complexes that have a (PH<sub>3</sub>)<sub>2</sub>IrX substructure, with the  $PH_3$  groups held mutually *trans* (X = various monodentate ligands). Provided that the X ligand and any additional ligands are not unusually large, we can assume that steric interactions with the PH<sub>3</sub> groups are negligible and that perturbations to the calculated reaction energies between complexes with different X ligands are strictly attributable to electronic effects. The electronic effects of various X ligands were further separated into  $\sigma$  and  $\pi$  effects and separately parameterized yielding  $\sigma_X$  and  $\pi_X$  parameters. The X ligands investigated feature a range of  $\sigma$ -type and  $\pi$ -type electronic properties: strongly  $\sigma$ donating ligands (H, Li, Ph, Me), weakly  $\sigma$ -donating ligands (pyrrolide), strongly  $\sigma$ donating and  $\pi$ -donating ligands (O<sup>2-</sup>, BH<sub>3</sub><sup>2-</sup>), strongly  $\sigma$ -donating and  $\pi$ -accepting ligands (BH<sub>2</sub>), weakly  $\sigma$ -donating and  $\pi$ -donating ligands (NH<sub>2</sub>, F, OH, OMe), and weakly  $\sigma$ -donating and  $\pi$ -accepting ligands (NH<sub>3</sub>, NO<sub>2</sub>). The orientations of some ligands (BH<sub>2</sub>, NH<sub>2</sub>, NO<sub>2</sub>, OH, OMe, Ph, and pyrrolide) were also constrained to be coplanar with or orthogonal to the plane of the molecule.

*Trans* influence was used to isolate the  $\sigma$  effects of different X ligands. The *trans* influence is a ground-state phenomenon in which the strength of a ligand-metal bond inversely affects the length of the ligand-metal bond located *trans* to it;<sup>44-46</sup> that is, if a particular X ligand forms a comparatively strong and short bond with the metal center, then a bond situated *trans* to it will tend to be correspondingly weak and long (Scheme

2.1a). If the *trans* ligand ( $L_T$ ) bonds solely to a metal center through  $\sigma$  interactions, then the *trans* influence produced by an X ligand directly reflects the effective  $\sigma$ -donating ability of X.

**Scheme 2.1.** (a) *Trans* Influence of an X Ligand on the M-L<sub>T</sub> Distance. (b)  $\pi$ -Backbonding of Electron Density from a Metal to Vacant  $\pi^*$  Antibonding Orbitals of CO.



Thus, the geometries of a series of *trans*-(PH<sub>3</sub>)<sub>2</sub>Ir(NH<sub>3</sub>)(X) complexes (**1-X**) were optimized, and the resulting equilibrium Ir-N bond distances were measured (Figure 2.1). It was assumed that NH<sub>3</sub> exhibits exclusively  $\sigma$ -type bonding interactions with the Ir atom; consequently, variations in Ir-N bond distances were attributed to variations in the  $\sigma$ -donation characteristics of X. The complexes were constrained to rigorously maintain square planar geometries (angles between ligands equal to 90°) to eliminate unusual perturbations to the geometry about the metal center. In addition, the orientation of certain X ligands (e.g. BH<sub>2</sub><sup>-</sup>) were constrained to be coplanar with or orthogonal to the plane of the molecule to control for differences in X ligand orientation.<sup>47</sup> All other geometrical variables were allowed to freely optimize. The differences in the Ir-N bond distances between complexes constrained (as described above) and unconstrained are relatively small, and thus the applied geometry constraints exert only a minor effect on the measured Ir-N bond distances.<sup>48</sup>

**Figure 2.1.** Complexes used for the determination of electronic substituent effect parameters ( $\sigma_x$ ,  $\pi_x$ ).



The  $\sigma_X$  parameter was referenced relative to the Ir-N bond distance for 1-H, as shown in equation 1. Ligands with a positive  $\sigma_X$  parameter have engendered elongations in the Ir-N bonds *trans* to X and are hence considered stronger  $\sigma$  donors than hydride, whereas ligands with a negative  $\sigma_X$  value are considered weaker  $\sigma$  donors than hydride. The series of calculated ligands and corresponding derived  $\sigma_X$  parameters are shown in Table 2.1. The series follows what one would traditionally expect: strong  $\sigma$ -donating ligands such methyl and phenyl exert large *trans* influences comparable to that of hydride are found to have  $\sigma_X$  values near zero (-0.015 and -0.012, respectively), while weakly  $\sigma$ donating substituents like fluoride and hydroxide possess relatively more negative  $\sigma_X$ values (-0.146 and -0.110, respectively). As expected, hydride is among the most strongly  $\sigma$ -donating ligands, as only BH<sub>3</sub><sup>2-</sup>, O<sup>2-</sup>, Li<sup>-</sup>, and BH<sub>2</sub><sup>-</sup> are calculated to have positive  $\sigma_X$  values, indicating more strongly  $\sigma$ -donating properties than hydride.

$$\sigma_{\rm X} = (\text{Ir-N bond distance}; \text{Å})_{\rm X} - (\text{Ir-N bond distance}; \text{Å})_{\rm H}$$
 (1)

X ligand	σχ	$\pi_X$
BH₂ <sup>-</sup> , ⊥	0.115	-0.038
BH₂ <sup>-</sup> , ∥	0.139	-0.036
BH3 <sup>2-</sup>	0.235	0.114
CF <sub>3</sub> <sup>-</sup>	-0.033	-0.031
CH <sub>3</sub> <sup>-</sup>	-0.015	0.018
F <sup>-</sup>	-0.146	0.076
H	0.000	0.000
Li	0.128	0.010
NH <sub>2</sub> <sup>-</sup> , <sup>⊥</sup> 1	-0.072	0.037
NH <sub>2</sub> ,    <sup>2</sup>	-0.072	0.072
NH₃	-0.153	-0.064
NO <sub>2</sub> <sup>-</sup> , ⊥	-0.081	0.001
$NO_2^-$ ,	-0.071	-0.023
0 <sup>2-</sup>	0.023	0.312
OH <sup>-</sup> , ⊥	-0.110	0.064
ОН⁻, ∥	-0.113	0.083
OMe <sup>-</sup> , ⊥	-0.106	0.059
OMe⁻, ∥	-0.113	0.072
Ph <sup>-</sup> , ⊥	-0.012	0.002
pyrrolide, ⊥	-0.108	0.017
= coplanar with P-Ir-P axis		
$\perp$ = orthogonal to P-Ir-P axis		
<sup>1</sup> H-N-Ir-P dihedrals set to 58.5°		
<sup>2</sup> H-N-Ir-P dihedrals set to 31.5°		

**Table 2.1.** Calculated Electronic Substituent Parameters ( $\sigma_X$ ,  $\pi_X$ ).

We gauge  $\pi$ -effects from consideration of metal( $d_{\pi}$ )-ligand( $p_{\pi}$ ) backbonding using a series of 4-coordinate *trans*-(PH<sub>3</sub>)<sub>2</sub>Ir(CO)(X) complexes **2-X** (Figure 2.1). The extent to which electron density is transferred from occupied metal d-orbitals into formally vacant CO( $\pi^*$ ) orbitals (Scheme 2.1b)<sup>5</sup> was quantified via NBO analyses to determine the occupancies of Lewis-type bonding and antibonding orbitals.<sup>41,42</sup>

First, the geometries of **2-X** were fully optimized without any imposed symmetry constraints, and the *average* Ir-CO bond distance was found to be 1.86796 Å. Geometries of **2-X** were then constrained to be rigorously square planar in the same manner as **1-X** described above and, additionally, the Ir-CO bond distance was constrained to the length 1.86796 Å. This latter constraint was imposed due to the concern that the *trans*-influence of ligand X would affect the Ir-CO bond distance and influence, accordingly,  $Ir(d_{\pi})$ -CO( $\pi^*$ ) orbital overlap. A change in overlap due to variation in the Ir-CO bond distance affects the  $\pi^*$  bond occupancies independent of the  $\pi$  donating or accepting characteristics of ligand X, thus introducing a  $\sigma$ -type component to the occupancy values of the virtual CO( $\pi^*$ ) orbitals. The effect is illustrated in Table 2.2, where varying the Ir-CO distance of complex **2-F** changes the CO( $\pi^*$ ) bond occupancy. By constraining the C-O bond distance to a common value for all complexes, this effect is eliminated, and the differences in orbital occupancies reflect therefore solely differences in the  $\pi$ -donating and accepting abilities of ligands X.

d(lr-CO) (Å)	CO(π*) orbital occupancy
1.850	0.585
1.855	0.582
1.860	0.579
1.865	0.576
1.86796 <sup>a</sup>	0.574
1.870	0.573
1.875	0.569
1.880	0.566
1.885	0.563
1.890	0.560

**Table 2.2.** Effect of the Ir-CO Bond Distance on the  $CO(\pi^*)$  Orbital Occupancy of **2-F**.

<sup>a</sup>Average C-O bond length from fully optimized **2-X** geometries, see text.

The total  $CO(\pi^*)$  orbital occupancy is the sum of the NBO populations in the two orthogonal CO( $\pi^*$ ) antibonding orbitals. Analogously to the  $\sigma_X$  parameter, the  $\pi_X$ substituent parameter is referenced to the  $CO(\pi^*)$  occupancy for 2-H and defined as shown in equation 2. Ligands with a positive  $\pi_{\rm X}$  parameter are considered  $\pi$  donors, whereas ligands with a negative  $\pi_X$  value are considered  $\pi$  acceptors. The trend in the  $\pi_X$ values (Table 2.1) is as expected:  $\pi$  donating ligands like amido and methoxide have positive  $\pi_X$  values (both 0.072), while a  $\pi$  accepting ligand like BH<sub>2</sub><sup>-</sup> has a negative  $\pi_X$ value (-0.036). The relative orientation of the ligands, while not significantly influencing  $\sigma_X$  values, does have a more significant effect on the value of the  $\pi_X$  parameter. The amido ligand shows very different  $\pi_X$  values when oriented coplanar with the P-Ir-P axis (0.072) or orthogonal to the axis (0.037). Additionally, while ligands like NH<sub>3</sub> and BH<sub>3</sub> are not expected to interact significantly in a hyperconjugative  $\pi$  sense with a metal d orbital, the charge that the ligand imparts on the complexes affects the effective  $\pi$ characteristics of the metal center,<sup>49</sup> and as such, they are effectively considered as a  $\pi$ acceptor and  $\pi$  donor, respectively.

 $\pi_{\rm X} = ({\rm CO} \ \pi^* \text{ bond occupancy; e})_{\rm X} - ({\rm CO} \ \pi^* \text{ bond occupancy; e})_{\rm H}$  (2)

#### 2.4 Oxidative Addition of CH<sub>4</sub> to 3-Coordinate trans-(PH<sub>3</sub>)<sub>2</sub>IrX

The energy of oxidative addition of methane to  $(PH_3)_2IrX$  was calculated for a variety of X ligands ( $\Delta E_X$ , Table 2.3). Structures for the 3-coordinate reactants and the 5-coordinate products were optimized both with and without imposed geometry restrictions. When the geometries were restricted, the P-Ir-P angles were fixed at 180°,

and the remaining ligands were held rigorously perpendicular to the P-Ir-P axis. Selected X ligands were constrained to be coplanar with or orthogonal to the P-Ir-P axis, and the Ir-C-H angles of the methyl group of the 5-coordinate products were set to tetrahedral values ( $109.47^{\circ}$ ). We found that the unconstrained geometries of the 5-coordinate products fell into three general categories: square pyramidal and two types of distorted trigonal bi-pyramidal.<sup>50-52</sup> These geometries were then standardized into three model geometries that are described by the local coordination in the equatorial plane: T-shape (square pyramidal, **3-T**<sub>transPH3</sub>), Y<sub>Me</sub> shape (where the X-Ir-H angle is set to 72°, **3-Y**<sub>Me</sub>), and  $Y_X$  shape (where the Me-Ir-H angle is set to 72°, 3- $Y_X$ )(Figure 2.2). For both constrained and unconstrained geometries, the computed data (reaction energies, bond distances and orbital populations) was treated in the same manner. The numerical fits of the regression analyses were better when constrained geometries were used, but, overall, the same conclusions may be drawn from either set of geometries and regression analyses. Because the geometry variations of specific complexes are perturbations to the calculated reaction energies independent of the underlying electronic factors of the X ligands, and since the overall conclusions remain the same regardless of geometry restrictions, only the geometry constrained data will be presented below.

$X - M + CH_4 \longrightarrow X - M < H$			
	ΔΕ	 Ev (kcal/m	ol)
X ligand	 Y <sub>Mo</sub>	Y <sub>Y</sub>	Т
BH,∵,⊥	3.8	14.8	15.2
BH <sub>2</sub> <sup>-</sup> ,	-2.0	32.6	18.6
BH3 <sup>2-</sup>	0.3	9.3	15.9
CF <sub>3</sub> <sup>-</sup>	-3.4	3.9	-2.5
CH <sub>3</sub> <sup>-</sup>	-4.9	-4.6	-1.3
F <sup>.</sup>	-15.1	-35.1	-28.8
H	-6.9	1.7	-0.7
Li⁼	-5.4	14.6	15.0
$NH_{2}^{-}, \perp^{1}$	-6.3	-10.4	-8.5
NH₂ <sup>-</sup> , ∥ <sup>2</sup>	-10.2	-24.6	-14.4
$NH_3$	-16.1	-28.9	-32.7
NO₂ <sup>-</sup> , ⊥	-3.6	-6.6	-10.9
NO2 <sup>-</sup> ,	-2.4	-0.5	-6.5
0 <sup>2-</sup>	0.3	-23.1	-7.5
OH <sup>-</sup> , ⊥	-12.2	-22.8	-18.4
OH <sup>™</sup> , ∥	-10.2	-30.6	-20.9
OMe <sup>-</sup> , ⊥	-9.3	-19.4	-15.8
OMe <sup>-</sup> ,	-9.0	-28.4	-20.2
Ph <sup>-</sup> , ⊥	-1.6	2.9	1.6
pyrrolide, ⊥	-9.0	-14.7	-16.3
= coplanar with P-Ir-P axis			
$\perp$ = orthogonal to P-Ir-P axis			
<sup>1</sup> H-N-Ir-P dihedrals set to 58.5°			
<sup>2</sup> H-N-Ir-P dihedrals set to 31.5°			

**Table 2.3.** Calculated Reaction Energies ( $\Delta E_X$ , kcal/mol) of CH<sub>4</sub> Addition to (PH<sub>3</sub>)<sub>2</sub>IrX for Different Equatorial Plane Geometries, cf. Figure 2.2.<sup>a</sup>

<sup>a</sup>For each ligand X, the lowest calculated  $\Delta E_X$  is in bold font.





We propose that the differences in reaction energies ( $\Delta\Delta E$ ) arise from differences in  $\sigma$  and  $\pi$  effects exerted by the X ligands. A 2-variable linear regression was performed in which the  $\sigma_X$  and  $\pi_X$  parameters were simultaneously fit to the data set comprised of the lowest calculated reaction energy for each X ligand (bold font values shown in Table 2.3; Figure 2.3). Inspection of the sign and magnitude of the coefficients obtained in the regression equation reveals which X ligand characteristics are favorable for oxidative addition. The large positive coefficient for the  $\sigma_X$  parameter (86.1) indicates that *weaker*  $\sigma$  donating ligands favor oxidative addition. The large negative coefficient for the  $\pi_X$ parameter indicates that oxidative addition is favored by  $\pi$ -donating ligands (-60.8). Given that conventional wisdom suggests that oxidative addition is generally favored by more electron-donating ligands, the result obtained for the  $\sigma_X$  parameter was surprising, since it clearly suggests that oxidative addition should not be regarded simply as an oxidative event at the metal.

**Figure 2.3.** 2-Variable linear regression fit of calculated  $\Delta E_X$  to  $\sigma_X$  and  $\pi_X$  parameters for the constrained, lowest energy conformers of CH<sub>4</sub> addition to *trans*-(PH<sub>3</sub>)<sub>2</sub>IrX.



While the above regression analysis utilizes the lowest reaction energy for each X ligand, each singular geometry case ( $Y_{Me}$ ,  $Y_X$ ,  $T_{transPH3}$ ) may, of course, be treated separately. Although the use of constrained geometries results in higher  $\Delta E_X$  values, it does allow for the direct comparison of ligands with controlled geometries. Computed  $\Delta E_X$  values for each of the three constrained geometry cases are presented in Table 2.3, and the resulting regression equations are given in Table 2.4. In all three cases ( $Y_{Me}$ ,  $Y_X$ ,  $T_{transPH3}$ ), the  $\sigma_X$  parameter coefficient is large and positive, indicating that oxidative addition is favored by less  $\sigma$ -donating ligands, independent of geometry. The  $\pi_X$  parameter coefficient varies, from negative in the case of  $Y_X$  to close to or effectively zero for  $T_{transPH3}$  and  $Y_{Me}$ .

	regression equation	std. dev. of σ effects (kcal/mol)	std. dev. of π effects (kcal/mol)
$ \begin{array}{ c c c } \hline PH_3 & PH_3 & PH_3 \\ X - \begin{matrix} I \\ I \\ PH_3 \\ PH_3 \\ H_3 \\ H$	$\Delta E_{x} = (37 \pm 8)\sigma_{x} + (2 \pm 10)\pi_{x} + (-5.2 \pm 0.9)$ $R^{2} = 0.570$	4.0	0.2
$ \begin{array}{ c c c c } & & & & & & PH_3 \\ X - I_r & + CH_4 & \longrightarrow & X - I_r < H_3 & Y_X \text{ shape} \\ PH_3 & & PH_3 & \alpha = 72^{\circ} \\ PH_3 & & PH_3 & \alpha = 72^{\circ} \end{array} $	$\Delta E_{X} = (143 \pm 15)\sigma_{X} + (-106 \pm 19)\pi_{X} + (-0.5 \pm 1.7)$ $R^{2} = 0.876$	15.3	8.5
$\begin{array}{ c c c } \hline PH_3 & PH_3 & PH_3 \\ X - Ir & + CH_4 & \longrightarrow H_3C - Ir & X & Tshape \\ I & PH_3 & PH_3 & PH_3 \\ 1 & 3-T & \end{array}$	$\Delta E_x = (131 \pm 10)\sigma_x + (-34 \pm 13)\pi_x + (-2.0 \pm 1.2)$ $R^2 = 0.911$	14.0	2.8

Table 2.4. Regression Equations Using Restricted Geometries for  $3-Y_{Me}$ ,  $3-Y_X$ , and  $3-T_{transPH3}$ .

Because the  $\sigma_X$  and  $\pi_X$  parameters refer to widely different properties, it was unclear from casual inspection of a particular regression equation what the relative contributions of  $\sigma$  and  $\pi$  effects were on the overall energy of oxidative addition over the set of X ligands. To provide a sense of relative contributions, standard deviations were calculated for both  $\sigma$  and  $\pi$  effects by first multiplying the sets of individual computed  $\sigma_X$ and  $\pi_X$  parameters (Table 2.1) by the regression coefficients for the respective parameters (Table 2.4). The resulting two sets of energies correspond to the ( $\sigma$ ,  $\pi$ ) energy contributions for each ligand. The standard deviation of each set was then calculated, reflecting the relative significance of  $\sigma$  and  $\pi$  effects for that particular oxidative addition geometry (Y<sub>Me</sub>, Y<sub>X</sub>, T<sub>transPH3</sub>).

Across all three geometry types,  $\sigma$  effects are more significant for the energy of oxidative addition than  $\pi$  effects. For the **3-T**<sub>transPH3</sub> and **3-Y**<sub>Me</sub> geometries, the standard deviation of  $\sigma$  effects are significantly larger than the standard deviation of  $\pi$  effects (14.0 and 4.0 kcal/mol vs. 2.8 and 0.2 kcal/mol, respectively), suggesting that the energy of

oxidative addition is predominantly governed by  $\sigma$  effects. While the standard deviation of  $\sigma$  effects remains larger than the standard deviation of  $\pi$  effects for the 3-Y<sub>x</sub> geometry (15.3 vs. 8.5 kcal/mol),  $\pi$  effects more significantly influence the energy of oxidative addition in this geometry than in the other geometries.

In the rare cases when the X ligand exerts a stronger *trans* influence than hydride or methyl,<sup>16,50,53</sup> the 5-coordinate complex may adopt a square pyramidal geometry in which the X ligand is found in the apical position as opposed to an equatorial position (**3**- $T_{X-apical}$ ). In the regression equation for this geometry, a reversal of preference for  $\sigma$ effects is observed (Table 2.5a). The  $\sigma_X$  parameter coefficient is negative, suggesting that a more  $\sigma$ -donating ligand is favorable for oxidative addition. Like the **3**- $Y_{Me}$ geometry, the standard deviation of  $\sigma$  effects for **3**- $T_{X-apical}$  is significantly larger than the standard deviation of  $\pi$  effects (4.1 vs. 0.1 kcal/mol, respectively), suggesting that the energy of oxidative addition in the **3**- $T_{X-apical}$  geometry is predominantly governed by  $\sigma$ effects.

**Table 2.5.** Regression Equations Using Restricted Geometries for (a) the X-apical Square Pyramidal Geometry  $(3-T_{X-apical})$  and (b) the Oxidative Addition of Methane to cis-(PH<sub>3</sub>)<sub>2</sub>IrX (3-T<sub>cisPH3</sub>).

		regression equation	std. dev. of σ effects (kcal/mol)	std. dev. of π effects (kcal/mol)
(a)	$\begin{array}{c} PH_3 & PH_3 \\ X-Ir & + CH_4 & \longrightarrow H_3C-Ir -H \\ PH_3 & X-PH_3 \\ 1 & X-apical Square Pyramida \end{array}$	$\Delta E_{X} = (-38 \pm 7)\sigma_{X} + (1 \pm 18)\pi_{X} + (2.3 \pm 0.9)$ $R^{2} = 0.643$	4.1	0.1
(b)	$\begin{array}{c} PH_{3} & PH_{3} \\ H_{3}P-I_{r} & + CH_{4} & \longrightarrow & H_{3}P-I_{r}-CH_{3} \\ X & H_{X} \\ cis-(PH_{3})_{2}IrX \end{array}$	$\Delta E_{X} = (0 \pm 14)\sigma_{X} + (33 \pm 18)\pi_{X} + (-16 \pm 2)$ $R^{2} = 0.241$	0.0	3.1

Given the variety of pincer ligand motifs,<sup>54</sup> we were also interested in the energetic effect of varying the ligand occupying the position *cis* to the vacant coordination site. Consequently, the energy of oxidative addition of CH<sub>4</sub> to *cis*-(PH<sub>3</sub>)IrX complexes was calculated. The 5-coordinate structures were restricted to a square pyramidal geometry with the hydride located in the apical position (3-T<sub>cisPH3</sub>) and a 2variable linear regression was performed analogously to the treatment for the trans- $(PH_3)_2$ IrX complexes described above (Table 2.5b). The coefficient of determination for the regression is very low ( $R^2 = 0.241$ ), suggesting that the energy of oxidative addition does not correlate significantly with  $\sigma$  or  $\pi$  effects of the X ligand in this geometry. To the extent that the  $\sigma$  and  $\pi$  effects correlate with the energy of oxidative addition, the  $\sigma_X$ parameter coefficient is zero, suggesting that with that ligand position only  $\pi$  effects correlate with the energy of oxidative addition. As a practical consequence in ligand design, this result suggests that modification of a pincer ligand at the positions cis to the vacant coordination site should not strongly influence the thermodynamics of oxidative addition to 3-coordinate Ir(I); instead, the position of the pincer ligand *trans* to the vacant coordination site strongly influences the thermodynamics of oxidative addition.

# **2.5 Orbital Rationale**

These results can be rationalized from a consideration of orbital interactions in both the reactant 3-coordinate  $d^8$  Ir(I) complexes and the product 5-coordinate  $d^6$  Ir(III) complexes. In the 3-coordinate *trans*-(PH<sub>3</sub>)<sub>2</sub>IrX complexes, the  $d^8$  Ir(I) atom can be visualized as having filled  $d_{xy}$ ,  $d_{xz}$ , and  $d_{yz}$  orbitals and a filled  $d_{z2}$ -hybridized orbital that lies on the axis perpendicular to the plane containing the P, X, and Ir atoms (cf. Scheme 2.2).

In the 5-coordinate d<sup>6</sup> Ir(III) complexes, the d<sub>xy</sub>, d<sub>xz</sub>, and d<sub>yz</sub> orbitals remain filled, and two d orbitals are vacant. A traditional analysis of the orbital interactions involves  $\sigma$ bonding of the PH<sub>3</sub> groups, the X ligand, and the Me group with the d<sub>x2-y2</sub> orbital on Ir. In that orbital picture, the X ligand and the PH<sub>3</sub> groups interact equivalently with the d<sub>x2y2</sub> orbital, and no distinction is made between the positions *cis* and *trans* to the Me group. However, given the differences in the regression equations for the **3-T**<sub>transPH3</sub> and **3-T**<sub>cisPH3</sub> geometries (cf. Tables 2.4 and 2.5), it is necessary to invoke Landis's representation of sd-hybridized metal orbitals in order to differentiate the two ligand positions.<sup>45,55</sup> Here, three sd<sup>2</sup>-hybridized orbitals arise from the combination of the 5d<sub>x2-</sub> y2, 5d<sub>z2</sub>, and 6s orbitals on Ir. One of these sd<sup>2</sup>-hybridized orbital is oriented along the P-Ir-P axis, while the other two are oriented in the plane perpendicular to the P-Ir-P axis; for the **3-T**<sub>transPH3</sub> geometry, the sd<sup>2</sup>-hybridized orbitals are oriented along the Ir-X, Ir-P, and Ir-H axes.

If we examine  $\sigma$  interactions in the plane containing X, Ir, H, and Me, then the orbital picture can be simplified to contain just three orbitals: a filled d<sub>xy</sub> orbital and two empty sd<sup>2</sup>-hybridized orbitals that, when considered together, resemble the d<sub>x2-y2</sub> orbital in the shapes of the orbital lobes (Scheme 2.2). In the case of the square pyramidal geometry (**3-T**<sub>transPH3</sub>), the X, Me, and H groups interact directly with the two empty sd<sup>2</sup>-hybridized orbitals. Because the  $\sigma$  orbital of the X ligand "shares" one of the sd<sup>2</sup> orbitals with the Me group *trans* to it (3c-4e interaction), a less  $\sigma$ -donating X ligand will be favorable for oxidative addition.

Scheme 2.2. Orientation of  $sd^2$ -hybridized Orbitals and  $d_{xy}$  in 3-T<sub>transPH3</sub>, 3-Y<sub>X</sub>, and 3-Y<sub>Me</sub>.<sup>a</sup>



<sup>a</sup>The depictions of  $3-Y_X$  and  $3-Y_{Me}$  have coordinate axes that are rotated 45 degrees counterclockwise relative to that shown for  $3-T_{transPH3}$ .

The orbital picture is different for the  $Y_X$  and  $Y_{Me}$  shapes. For the  $Y_X$  geometry, the Me and H groups, which are strong *trans*-influence ligands, interact maximally with the two empty sd<sup>2</sup>-hybrid orbitals. The  $\sigma$  orbital of the X ligand then interacts directly with the filled  $d_{xy}$  orbital, an interaction that again is favored by a less  $\sigma$ -donating X ligand. For the  $Y_{Me}$  shape, the Me group and the X ligand switch places. The  $\sigma$  orbital of the X ligand is now oriented to interact primarily with the empty sd<sup>2</sup>-hybridized orbital, although orbital overlap is reduced because the ligand is positioned ca. 9° off the axis. Additionally, the Me group has a slight interaction with the filled  $d_{xy}$  orbital, although the geometry limits the angular overlap. Taken together, oxidative addition is slightly favored by less  $\sigma$ -donating X ligands in the  $Y_{Me}$  geometry, though not to the extent encountered in the **3-T**<sub>transPH3</sub> or **3-Y**<sub>X</sub> geometries.

While the  $3-T_{X-apical}$  geometry is similar to the  $3-T_{transPH3}$  geometry with the only difference being the positions of the X ligand and hydride in the 5-coordinate complex,

the regression equation for the 3- $T_{X-apical}$  geometry suggests a reversal in preference for  $\sigma$  effects; the coefficient of the  $\sigma_X$  parameter is negative (Table 2.5a), suggesting that more  $\sigma$ -donating ligands now are favorable for oxidative addition. In this geometry, the  $\sigma$  orbital of the X ligand would not need to "share" an sd<sup>2</sup>-hybridized orbital with a strong *trans*-influence ligand like Me or hydride as is the case for the 3- $T_{transPH3}$  geometry. Additionally, the favorability of more  $\sigma$ -donating ligands could arise in part from the antibonding interaction of X with the torus of the filled 5d<sub>z2</sub> orbital in the 3-coordinate complex.

The proposed orbital picture also correlates with the conclusions drawn from the coefficient of the  $\pi_X$  parameter. In the 3-coordinate *trans*-(PH<sub>3</sub>)<sub>2</sub>IrX complexes, the  $\pi$ -type orbitals would interact with the corresponding filled off-axis d orbital (d<sub>xy</sub>, d<sub>xz</sub>, or d<sub>yz</sub>). However, in the 5-coordinate d<sup>6</sup> Ir(III) complexes, the  $\pi$  interactions are not as simple. For the **3-Y**<sub>X</sub> geometry, the  $\pi$  interactions of X in the plane containing X, Ir, H, and Me would be with the empty sd<sup>2</sup>-hybridized orbitals. Additionally, the  $\pi$  interactions in the plane containing the PH<sub>3</sub> groups would be with a rotated filled d-orbital, resulting in a decrease in the angular overlap integral by ca. 30% from its optimal value. Both of these orbital effects would result in oxidative addition being strongly favored by  $\pi$  donating ligands, which corresponds nicely with the obtained  $\pi_x$  coefficient from the **3-Y**<sub>x</sub> regression equation, which is large and negative (-106, Table 2.4).

For the 3-T<sub>transPH3</sub> geometry, the X-Ir  $\pi$  interactions would be approximately the same as those in the 3-coordinate complexes (overlap with two filled d orbitals), suggesting little to no influence of  $\pi$  effects on the energy of oxidative addition. However, the  $\pi_X$  coefficient of the 3-T<sub>transPH3</sub> regression equation is relatively small and negative (-34, Table 2.4), suggesting that oxidative addition should be favored by slightly  $\pi$  donating ligands. This result could arise from  $\pi$  interactions between a  $\pi$ -type orbital on X and the vacant sd<sup>2</sup>-hybridized orbital oriented along the Ir-H axis. In the 3-coordinate complex, this type of interaction would be entirely non-bonding due to the symmetry of the orbitals. However, mixing of the 1s orbital of the hydride with this sd<sup>2</sup>-hybridized orbital would polarize the sd<sup>2</sup>-hybridized orbital such that net overlap with the  $\pi$ -type orbital of X could occur. Examination of the relevant MO for the *trans*-(PH<sub>3</sub>)<sub>2</sub>Ir(F)(H)(Me) complex in the **3-T**<sub>transPH3</sub> geometry reveals this  $\pi$ -type interaction (Figure 2.4).

**Figure 2.4.** The HOMO(-8) orbital of *trans*-(PH<sub>3</sub>)<sub>2</sub>Ir(F)(H)(Me) in the **3-T**<sub>transPH3</sub> geometry (isovalue = 0.015) showing how polarization of the metal sd<sup>2</sup>-hybridized orbital may create favorable overlap with the  $\pi$ -system of ligand X.



In contrast, the regression equation for the 3- $T_{X-apical}$  geometry suggests little to no influence of  $\pi$  effects on oxidative addition; the coefficient of the  $\pi_X$  parameter is effectively zero within the error of the regression (1 ± 18, Table 2.5). The  $\pi$  interactions in the 5-coordinate complex are expected to be similar to those in the 3- $T_{transPH3}$ . However, the polarization of the sd<sup>2</sup>-hybridized orbital that arose from mixing with the 1s orbital of the hydride would be attenuated by equivalent mixing with the  $\sigma$  orbital of the Me group. As a result, the interaction between the  $\pi$ -type orbital on the X ligand and the vacant sd<sup>2</sup>-hybridized orbital oriented along the Me-Ir-H axis remains a non-bonding due to orbital symmetry.

In the 3-Y<sub>Me</sub> geometry, the  $\pi$  interactions would be again approximately the same as those in the 3-coordinate complexes (albeit with slightly decreased overlap due to the 9° off-axis location of the X ligand). Consequently, the expected result that  $\pi$  effects have little to no influence on the energy of oxidative addition is reflected in the computed  $\pi_X$  coefficient that is zero within the error of the regression (2 ± 10, Table 2.4).

# 2.6 Experimental Verification

A series of iridium complexes were investigated computationally and experimentally to further investigate the unexpected result that oxidative addition to 3coordinate Ir(I) complexes should be favored by less  $\sigma$ -donating ligands. Four isoelectronic 5-coordinate Ir(III) phenyl hydride complexes have been previously reported in the literature: (PCP)Ir(H)(Ph) (4)(PCP =  $\kappa^3$ -C<sub>6</sub>H<sub>3</sub>-2,6-(CH<sub>2</sub>P<sup>t</sup>Bu<sub>2</sub>)<sub>2</sub>)<sup>56</sup>, (<sup>Am</sup>PNP)Ir(H)(Ph) (5) (<sup>Am</sup>PNP =  $\kappa^3$ -bis(2-diisopropylphosphino-4methylphenyl)amide)<sup>57,58</sup>, (<sup>Py</sup>PNP)Ir(H)(Ph)<sup>+</sup>PF<sub>6</sub><sup>-</sup> (6) (<sup>Py</sup>PNP =  $\kappa^3$ -C<sub>5</sub>H<sub>3</sub>N-2,6 $(CH_2P^tBu_2)_2)^{59,60}$ , and *trans*- $(P^iPr_3)_2$ -Ir(C1)(H)(Ph) (7).<sup>61,62</sup> Aside from the differences in the phosphine steric bulk and the variations in the P-Ir-P angles of the pincer-ligand frameworks, the series of complexes feature equatorial ligands with varying  $\sigma$ -donating abilities (Scheme 2.3). Complex **4** possesses a strongly  $\sigma$ -donating aryl carbon, while complex **7** has a less  $\sigma$ -donating chloride. Complexes **5** and **6** feature an amido nitrogen and a pyridyl nitrogen, respectively, and are hence expected to have intermediate  $\sigma$ -donating abilities.

**Scheme 2.3.** Several Ir(III) Phenyl Hydride Complexes and the Relative  $\sigma$ -donating Abilities of the Ligand at the Position *trans* to the Phenyl Moiety.



The  $\sigma_X$  and  $\pi_X$  parameters for the X ligands in complexes 4-7 were calculated from the corresponding 4-coordinate (ligand)Ir(NH<sub>3</sub>) and (ligand)Ir(CO) complexes (Table 2.6). The values of the  $\sigma_X$  parameter were consistent with the expected  $\sigma$ -donating properties of these complexes, whereas the  $\pi_X$  parameter values indicate the X ligands in complexes 4, 5, and 7 to be moderate  $\pi$ -donors (4 < 5 < 7) and cationic 6 to be neither  $\pi$ accepting or donating. With these  $\sigma_X$  and  $\pi_X$  parameters,  $\Delta E$  for the oxidative addition of methane were calculated from the regression equation presented in Table 4 for the Y<sub>X</sub> geometry. The calculated  $\Delta E_{OA}$  values become more favorable with increasingly negative  $\sigma_X$  parameters, except for the two PNP-pincer complexes. Complex 5 ( $\pi_X =$ 0.088) is calculated to be a stronger  $\pi$ -donor than complex 6 ( $\pi_X = 0.008$ ). This difference in  $\pi$  contribution results in oxidative addition being more favorable to complex 5 than complex 6, despite complex 5 being a stronger  $\sigma$ -donor. Actual DFT calculations (see Computational Methods) for the energies of benzene and methane oxidative addition (Table 2.7) are consistent with the regression-equation-derived trend (Table 2.6) of increasingly favorable oxidative addition across the series in Scheme 3 ( $\Delta E_{OA}$  : 4 > 6 > 5 > 7).

	σχ	πχ	Calculated $\Delta E_{OA}$ for $Y_X$ Geometry
4	0.017	0.070	-5.4
5	-0.097	0.088	-23.7
6	-0.107	0.008	-16.8
7	-0.137	0.106	-31.4

**Table 2.6.** Calculated  $\sigma_X$  and  $\pi_X$  Parameters for Complexes 4-7.

M + benzene $\rightarrow$ M(H)(Ph)				
M(H)(Ph)	ΔE	ΔH	ΔS	ΔG
4	-11.7	-11.9	-43.7	1.2
5	-37.9	-38.4	-37.3	-27.2
6	-30.8	-31.1	-41.3	-18.7
7	-42.1	-41.7	-49.6	-26.9
M + methane $\rightarrow$ M(H)(CH <sub>3</sub> )				
M(H)(Ph)	ΔE	ΔH	ΔS	ΔG
4	-2.1	-2.4	-34.3	7.8
5	-28.6	-28.7	-33.3	-18.8
6	-23.6	-23.7	-30.9	-14.5
7	-35.4	-35.1	-33.5	-25.1

 Table 2.7.
 Calculated Energies of Benzene and Methane Oxidative Addition to

 Complexes 4-7.<sup>a</sup>

<sup>a</sup>Units for  $\Delta E$ ,  $\Delta H$ , and  $\Delta G$  are kcal/mol; units for  $\Delta S$  are eu. Standard state for  $\Delta G$  is T = 298.15 K and P = 1 atm for each species participating in the reaction.

The transition states for oxidative C-H cleavage of benzene were located via DFT calculations, and the calculated kinetic parameters for reductive coupling are given in Table 2.8. The  $\Delta E^{\ddagger}$  and  $\Delta H^{\ddagger}$  values for reductive coupling from complexes **5** and **6** were found to be lower than the thermodynamic barriers for reductive elimination. This suggests that the overall kinetic barriers to reductive elimination for complexes **5** and **6** correspond to the transition state for  $\sigma$ -coordination of a benzene C-H bond. Since coordination of benzene to the 3-coordinate Ir(I) complexes is expected to be a barrierless or near-barrierless process on the potential energy surface, the overall kinetic barrier for the reductive elimination of benzene from complexes **5** and **6** can be approximated by the thermodynamic barriers shown in Table 2.7.

$M(H)(Ph) \rightarrow M + benzene$				
M(H)(Ph)	ΔE <sup>‡</sup>	$\Delta H^{\ddagger}$	∆S <sup>‡</sup>	ΔG <sup>‡</sup>
4	14.1	13.9	-3.2	14.8
5	25.3	20.3	-22.7	27.1
6	16.8	16.4	-3.0	17.3
7	33.3	32.1	0.7	31.9

 Table 2.8.
 Calculated Kinetic Parameters for Reductive Cleavage of Benzene for

 Complexes 4-7.

Direct experimental determination of the oxidative addition energies is difficult because the 3-coordinate complexes leading to the oxidative addition products are not directly observable. However, the microscopically reverse process - reductive elimination – can be observed, and the rates can be measured (Table 2.9). Although caution must be taken when drawing thermodynamic conclusions from kinetic data, the kinetic barrier to oxidative addition is expected to be small relative to the much larger thermodynamic differences. EXSY NMR experiments were performed with complex 4 at a range of temperatures from -54 °C to -27 °C and yielded activation parameters for reductive elimination of 13.2 kcal/mol and -1.7 eu for  $\Delta H^{\ddagger}$  and  $\Delta S^{\ddagger}$ , respectively. The rate of reductive elimination for complexes 5 was measured at a range of temperatures from 85 °C to 114 °C by monitoring the rate of disappearance of the hydride signal upon exchange with toluene- $d_8$ ; the enthalpic and entropic contributions to the barrier were determined to be 31.2 kcal/mol and 6.3 eu, respectively. The rate of reductive elimination for complex 6 at 84 °C was measured to be 1.3 x  $10^{-4}$  s<sup>-1</sup>, which corresponds to a kinetic barrier ( $\Delta G^{\ddagger}$ ) of 27.4 kcal/mol. A conservative upper limit for the rate of benzene loss from complex 7 at 120 °C was previously reported to be 5 x  $10^{-5}$  s<sup>-1</sup>, corresponding to a kinetic barrier ( $\Delta G^{\ddagger}$ ) of 30.9 kcal/mol.<sup>62</sup>

**Table 2.9.** Experimentally Determined Kinetic Parameters for the Reductive Elimination

 of Benzene for Complexes 4-7.

$M(H)(Ph) \rightarrow M + benzene$				
М	$\Delta H^{\ddagger}$	∆S <sup>‡</sup>	∆G <sup>‡</sup> (at 84 °C)	ΔG <sup>‡</sup> (120 °C)
4	13.2 (±0.9)	-1.7 (±3.7)	13.8 <sup>ª</sup>	13.9 <sup>a</sup>
5	31.2 (±1.9)	6.3 (±5.1)	29.0 <sup>ª</sup>	28.7 <sup>a</sup>
6	$k = 1.25 \times 10^{-4} s^{-1} (84 °C)$		27.4	27.5 <sup>b</sup>
7	k ≤ 5.0 x 10	0⁻⁵ (120 °C)	≥ 31.2 <sup>°</sup>	≥ 30.9
$\Delta H^{\ddagger}$ and $\Delta G^{\ddagger}$ values given in kcal/mol. $\Delta S^{\ddagger}$ values given in eu. <sup>a</sup> Extrapolated values. <sup>b</sup> Calculated with $\Delta S^{\ddagger} = -1.7$ eu. <sup>c</sup> Calculated with $\Delta S^{\ddagger} = 6.3$ eu.				

The trend for the relative rates of reductive elimination correlates well with the expected trends from both DFT calculation (Tables 2.7 and 2.8) as well as from the regression equation (Table 2.6). Complex 4 bearing the strongly  $\sigma$ -donating aryl ligand was found to have the smallest barrier to reductive elimination, while complex 7 bearing the weakly  $\sigma$ -donating chloride ligand was found to have the largest barrier. Complexes 5 and 6 that have ligands of relatively intermediate  $\sigma$ -donating ligands were found to have intermediate rates of reductive elimination (vs. complexes 4 and 7).

### 2.7 Oxidative Addition of Methane to 4-Coordinate Ir(I) Complexes

We find that the principles of *trans* influence effects can also be more generally applied to 4-coordinate Ir(I) complexes. Oxidative addition to 4-coordinate 16-electron

Ir(I) complexes is well known in the literature.<sup>13,63-67</sup> Upon applying the same 2-variable linear regression analysis method to the oxidative addition of methane to 4-coordinate Ir(I) complexes, three regression equations were obtained, corresponding to the three possible 6-coordinate isomeric complexes (Table 2.10). The regression analysis of the two oxidative addition reactions that proceed in a *cis* fashion yield regression equations that have positive  $\sigma_X$  coefficients and negative  $\pi_X$  coefficients, which is consistent with the analysis of oxidative addition to 3-coordinate Ir(I). Analogously to the **3-T** square pyramidal geometry, the ligands of the 6-coordinate oxidative addition products may be considered again as interacting with sd<sup>2</sup>-hybridized orbitals that are oriented along the three metal-ligand bond axes. Less  $\sigma$ -donating X ligands favor oxidative addition, due to the presence of a strong *trans*-influence ligand (hydride or methyl) *trans* to the X ligand in the 6-coordinate product (3c-4e bonding interaction).

**Table 2.10.** Regression Equations Based on Constrained Geometries.

	regression equation	std. dev. of σ effects (kcal/mol)	std. dev. of π effects (kcal/mol)
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	$\Delta E_{x} = (44 \pm 7)\sigma_{x} + (-20 \pm 9)\pi_{x} + (-1 \pm 1)$ $R^{2} = 0.686$	4.5	1.5
$\begin{array}{c} \begin{array}{c} PH_3 & PH_3\\ X - I_r - NH_3 + CH_4 \longrightarrow X - I_r \stackrel{M \subset H_3}{\longrightarrow} \operatorname{cis} \operatorname{addition}\\ I_{PH_3} & H_3 N^{-} I_{PH_3} \end{array}$	$\Delta E_x = (52 \pm 10)\sigma_x + (-31 \pm 13)\pi_x + (1 \pm 1)$ $R^2 = 0.683$	5.7	2.5
$\begin{array}{c} P_{H_3} & P_{H_3} \\ I & I_1 & \\ X & I_1 & \\ P_{H_3} & P_{H_3} + CH_4 \longrightarrow X & I_3 \\ H_3 & P_{H_3} & \\ H_3 & P_{H_3} & P_{H_3} \end{array}$	$\Delta E_x = (-32 \pm 6)\sigma_x + (28 \pm 8)\pi_x + (2 \pm 1)$ $R^2 = 0.676$	3.4	2.2

The opposite preference is observed when oxidative addition proceeds in a *trans*fashion. The  $\sigma_X$  coefficient is negative (-32), while the  $\pi_X$  coefficient is positive (28). This reversal of preference, however, is still consistent with the orbital rationale given above for oxidative addition to 3-coordinate Ir(I) complexes; in the present case, the analogous 5-coordinate complex is the X-apical square pyramidal geometry. In the 6coordinate case, the NH<sub>3</sub> group is positioned *trans* to the X ligand throughout the transformation, akin to the vacant coordination site in the analogous 3-/5-coordinate case. Again, we would not expect less  $\sigma$ -donating ligands to favor oxidative addition, and, indeed, the 2-variable linear regression nicely mirrors the 5-coordinate case. The coefficient for the  $\sigma_X$  parameter is again small and negative (Table 2.10), suggesting that a strong  $\sigma$  donor slightly favors oxidative addition.

# 2.8 Oxidative Addition of Ammonia to 3-Coordinate Ir(I) Complexes

In addition to C-H oxidative addition, we were interested in investigating the electronic effects influencing N-H addition, in particular NH<sub>3</sub> oxidative addition. There are relatively few examples of oxidative addition of ammonia that yield an amido hydride complex,<sup>15-20</sup> and only two examples of N-H oxidative addition of ammonia to a single, well-defined, late transition metal center have been reported – one from our group in 2005<sup>15</sup> and one by Turculet and co-workers in 2009.<sup>16</sup> In 2005, we reported that aliphatic PCP pincer Ir complex **8** reacts with ammonia and a base to give amido hydride **9** at 25 °C (Scheme 2.4).<sup>15</sup> However, we found that when the aryl PCP pincer Ir amido hydride complex **10** was synthesized, it underwent isomerization to the 4-coordinate NH<sub>3</sub> coordination complex **11** at 25 °C. We had proposed that the more electron-donating aliphatic PCP pincer ligand of complex **9** should favor oxidative addition more than the aryl PCP pincer ligand of complex **10** in accordance with the conventional wisdom that a

more electron-rich metal center should favor oxidative addition. This hypothesis seemed to be confirmed by the recent report from the Turculet group in which oxidative addition of ammonia is observed with an iridium pincer complex featuring a highly  $\sigma$ -donating P-Si-P ligand.<sup>16</sup> However, this reasoning is co*ntrary* to the conclusions of the work reported above for methane oxidative addition, where less  $\sigma$ -donating ligands should favor oxidative addition; thus, ammonia oxidative addition was investigated in the same manner as the methane addition presented above.





The reaction energies for oxidative addition of NH<sub>3</sub> to a variety of 3-coordinate Ir(I) complexes were calculated (Table 2.11),<sup>68</sup> and a 2-variable linear regression was performed to fit the  $\sigma_X$  and  $\pi_X$  parameters to the reaction energies (Table 2.12). We found that the coefficient of the  $\sigma_X$  parameter was positive (68.4), again indicating that oxidative addition is favored by less  $\sigma$ -donating ligands. Although this result appeared

contrary to the observed reactivity of complexes 9 and 10, N-H oxidative addition of ammonia was then considered in the context relative to the commonly-favored  $\rm NH_3$  coordination.<sup>69,70</sup>

X ligand	$\Delta E_{NH3}$ oxidative addition	$\Delta E_{NH3}$ coordination		
BH <sub>2</sub> <sup>-</sup> , ⊥	-7.3	-12.8		
BH₂ <sup>-</sup> , ∥	-15.6	-12.5		
BH <sub>3</sub> <sup>2-</sup>	-9.1	-0.2		
CF <sub>3</sub>	-21.5	-25.2		
CH <sub>3</sub> <sup>-</sup>	-20.1	-21.8		
F	-35.4	-39.3		
H	-22.9	-21.6		
Lī	-12.9	-4.3		
NH <sub>2</sub> <sup>-</sup> , ⊥ <sup>1</sup>	-24.2	-26.4		
NH₂ <sup>-</sup> , ∥ <sup>2</sup>	-19.2	-28.1		
NH <sub>3</sub>	-40.9	-53.1		
NO <sub>2</sub> <sup>-</sup> , ⊥	-21.2	-26.5		
NO <sub>2</sub> <sup>-</sup> ,	-22.1	-28.0		
0 <sup>2-</sup>	-7.5	-11.9		
OH <sup>-</sup> , ⊥	-27.9	-32.4		
OH⁻, ∥	-27.8	-32.8		
OMe⁻, ⊥	-27.3	-30.0		
OMe⁻, ∥	-26.9	-32.3		
Ph <sup>-</sup> , ⊥	-18.0	-21.3		
pyrrolide, ⊥	-29.6	-33.4		
= coplanar with P-Ir-P axis				
$\perp$ = orthogonal to P-Ir-P axis				
1 1	<sup>1</sup> H-N-Ir-P dihedrals set to 58.5°			
<sup>2</sup> H-N-Ir-P dihedrals set to 31.5°				

Table 2.11. Calculated  $\Delta E_X$  for NH<sub>3</sub> Oxidative Addition and Coordination to  $(PH_3)_2IrX$ .

std dev of std dev of σ effects regression equation  $\pi$  effects (kcal/mol) (kcal/mol)  $\Delta E_{X} = (68 \pm 9)\sigma_{X} + (27 \pm 13)\pi_{X} + (-21 \pm 1)$ 7.3 2.2  $R^2 = 0.777$ PH<sub>3</sub> PH<sub>3</sub> I I I Ir + NH<sub>3</sub>→→ X−Ir−NH<sub>3</sub> NH<sub>3</sub> coordination PH<sub>3</sub> PH<sub>3</sub>  $\Delta \mathsf{E}_{X} = (105 \pm 9)\sigma_{X} + (30 \pm 12)\pi_{X} + (-23 \pm 1)$ 11.2 2.4  $R^2 = 0.897$ 

**Table 2.12.** Regression Equations for the Oxidative Addition and Coordination of Ammonia to *trans*-(PH<sub>3</sub>)<sub>2</sub>IrX Using Constrained Geometries.

Reaction energies for the coordination of NH<sub>3</sub> to a variety of 3-coordinate Ir(I) complexes were calculated (Table 2.11), and a 2-variable linear regression was performed (Table 2.12). The  $\pi_X$  coefficient in the regression equation for NH<sub>3</sub> coordination (30) was equivalent within the error limits of the regression to the  $\pi_{\rm X}$ coefficient of the regression equation for  $NH_3$  oxidative addition (27). This suggests that  $\pi$  effects exert no significant control over the favorability of NH<sub>3</sub> coordination vs. oxidative addition. By contrast, a large difference is seen in the magnitude of the  $\sigma_X$ coefficients of the two regression equations. The coefficient of the  $\sigma_X$  parameter in the regression equation for NH<sub>3</sub> coordination was again positive (104.8), indicating that coordination is favored by less  $\sigma$ -donating ligands. However, the  $\sigma_X$  coefficient for the NH<sub>3</sub> coordinating regression equation has a larger magnitude than the  $\sigma_X$  coefficient for N-H oxidative addition (68.4). This suggests that while a more  $\sigma$ -donating ligand is unfavorable for oxidative addition, it is even more unfavorable for  $NH_3$  coordination. As a result, complexes with a more  $\sigma$ -donating ligand like complex 9 or Turculet's PSiPpincer Ir complex disfavor  $NH_3$  coordination relative to N-H oxidative addition, to such an extent that the amido hydride complex is observed.

### 2.9 Summary

DFT calculations were performed on a series of Ir(I) complexes to investigate the electronic factors that affect the thermodynamics of C-H and N-H oxidative addition. Electronic substituent effects were decomposed into their  $\sigma$  and  $\pi$  effects and parameterized, and these parameters were used to fit the energies of oxidative addition of methane to a series of 3- and 4-coordinate Ir(I) complexes. Regression analysis indicates that oxidative addition to 3-coordinate Ir(I) complexes is generally favored in most cases by a less  $\sigma$ -donating ligand trans to the vacant coordination site, although this preference reverses in the special case where the geometry of the 5-coordinate product is square pyramidal with the X ligand in the apical position. The oxidative addition of methane to 4-coordinate Ir(I) was found to be analogous in its electronic preferences to the 3-coordinate Coordinate Ir(I) was found to be analogous in its electronic preferences to the 3-coordinate Coordinate Coordinate

Given that conventional wisdom suggests that oxidative addition is favored by more electron-donating ligands, the surprising result that less  $\sigma$ -donating ligands are usually favorable for oxidative addition to Ir(I) suggests that oxidative addition should not be regarded simply as an oxidative event at the metal. Instead, an orbital rationale was proposed which invokes sd<sup>2</sup>-hybridized orbitals and the strong role of *trans* influence. Importantly, rates of reductive elimination from a series of isoelectronic Ir(III) phenyl hydride complexes were measured and found to be fully consistent with the computational analysis.

The regression analysis was extended to investigate  $NH_3$  oxidative addition to 3coordinate Ir(I) complexes. While a more  $\sigma$ -donating ligand is unfavorable for oxidative addition (analogous to methane oxidative addition), it is even more unfavorable for  $NH_3$  coordination. This result explains the observed reactivity in which a more  $\sigma$ -donating ligand disfavors NH<sub>3</sub> coordination relative to N-H oxidative addition and hence affords the amido hydride complex.

# 2.10 Acknowledgments

We thank O. Eisenstein for thoughtful discussions regarding this work.

# References

- (1) Choi, J.; MacArthur, A. H. R.; Brookhart, M.; Goldman, A. S. Chem. Rev. 2011, 111, 1761.
- (2) *Modern Arylation Methods*; Ackermann, L., Ed.; Wiley-VCH Verlag GmbH & Co. KGaA: Weinheim, 2009.
- (3) Cotton, F. A.; Wilkinson, G.; Murillo, C. A.; Bochmann, M. *Advanced Inorganic Chemistry*; 6th ed.; Wiley-Interscience: New York, 1999.
- (4) Crabtree, R. H. *The Organometallic Chemistry of the Transition Metals*; 4th ed.; John Wiley & Sons, Inc.: Hoboken, NJ, 2005.
- (5) Hartwig, J. F. Organotransition Metal Chemistry: From Bonding to Catalysis; University Science Books: Sausalito, CA, 2010.
- (6) Su, M.-D.; Chu, S.-Y. J. Phys. Chem. A 1998, 102, 10159.
- (7) Su, M.-D.; Chu, S.-Y. *Inorg. Chem.* **1998**, *37*, 3400.
- (8) Hartwig, J. F. Inorg. Chem. 2007, 46, 1936.
- (9) Ariafard, A.; Yates, B. F. J. Organomet. Chem. 2009, 694, 2075.
- (10) Yamashita, M.; Cuevas Vicario, J. V.; Hartwig, J. F. J. Am. Chem. Soc. 2003, 125, 16347.
- (11) *Activation and Functionalization of C-H Bonds*; Goldberg, K. I.; Goldman, A. S., Eds.; American Chemical Society: Washington, DC, 2004.
- (12) Lyons, T. W.; Sanford, M. S. Chem. Rev. 2010, 110, 1147.
- (13) Choi, J.; Goldman, A. S. Top. Organomet. Chem. 2011, 34, 139.
- (14) Crabtree, R. H. J. Organomet. Chem. 2004, 689, 4083.
- (15) Zhao, J.; Goldman, A. S.; Hartwig, J. F. Science 2005, 307, 1080.
- (16) Morgan, E.; MacLean, D. F.; McDonald, R.; Turculet, L. J. Am. Chem. Soc. 2009, 131, 14234.
- (17) Nakajima, Y.; Kameo, H.; Suzuki, H. Angew. Chem., Int. Ed. 2006, 45, 950.
- (18) Casalnuovo, A. L.; Calabrese, J. C.; Milstein, D. Inorg. Chem. 1987, 26, 971.
- (19) Khaskin, E.; Iron, M. A.; Shimon, L. J. W.; Zhang, J.; Milstein, D. J. Am. Chem. Soc. 2010, 132, 8542.
- (20) Hillhouse, G. L.; Bercaw, J. E. J. Am. Chem. Soc. 1984, 106, 5472.
- (21) Saillard, J. Y.; Hoffmann, R. J. Am. Chem. Soc. 1984, 106, 2006.
- (22) Low, J. J.; Goddard, W. A. J. Am. Chem. Soc. 1986, 108, 6115.
- (23) Koga, N.; Morokuma, K. J. Phys. Chem. 1990, 94, 5454.
- (24) Cundari, T. R. J. Am. Chem. Soc. 1994, 116, 340.
- (25) Xu, W.-w.; P. Rosini, G.; Krogh-Jespersen, K.; S. Goldman, A.; Gupta, M.; M. Jensen, C.; C. Kaska, W. *Chem. Commun.* **1997**, 2273.
- (26) Krogh-Jespersen, K.; Goldman, A. S. In *Transition State Modeling for Catalysis*; Truhlar, D. G., Morokuma, K., Eds.; American Chemical Society: 1999; *721*, pp 151.
- (27) Su, M.-D.; Chu, S.-Y. J. Am. Chem. Soc. 1997, 119, 10178.
- (28) Macgregor, S. A. Organometallics 2001, 20, 1860.
- (29) Diggle, R. A.; Macgregor, S. A.; Whittlesey, M. K. Organometallics 2004, 23, 1857.
- (30) Parr, R. G.; Yang, W. *Density-Functional Theory of Atoms and Molecules*; University Press: Oxford, 1989.
- (31)Gaussian 03, Revision C.02, Frisch, M. J.; Trucks, G. W.; Schlegel, H. B.; Scuseria, G. E.; Rob, M. A.; Cheeseman, J. R.; Jr., J. A. M.; Vreven, T.; Kudin, K. N.; Burant, J. C.; J. M. Millam; Iyengar, S. S.; Tomasi, J.; V. B.; B. M.; M. Cossi; G. Scalmani; N. Rega; G. A. Petersson; H. Nakatsuji; M. Hada; M. Ehara; K. Toyota; Fukuda, R.; , J. H.; , M. I.; T. Nakajima; Y. Honda; O. Kitao; Nakai, H.; , M. K.; X. Li, J. E. K.; H. P. Hratchian; J. B. Cross; V. Bakken; C. Adamo; J. Jaramillo; R. Gomperts; R. E. Stratmann; O. Yazyev; A. J. Austin; R. Cammi; C. Pomelli, J.; W. Ochterski; P. Y. Ayala; K. Morokuma; G. A. Voth; P. Salvador; J. J. Dannenberg; Zakrzewski, V. G.; , S. D.; A. D. Daniels; M. C. Strain; O. Farkas; D. K. Malick; A. D. Rabuck; K. Raghavachari; J. B. Foresman; J. V. Ortiz; Q. Cui; A. G. Baboul; S. Clifford; J. Cioslowski; B. B. Stefanov; G. Liu; A. Liashenko; P. Piskorz; I. Komaromi; R. L. Martin; D. J. Fox; T. Keith; Al-Laham, M. A.; , C. Y. P.; A. Nanayakkara; M. Challacombe; P. M. W. Gill; B. Johnson; W. Chen; M. W. Wong; C. Gonzalez; Pople, J. A., Gaussian, Inc., Wallingford, CT.
- (32) Gaussian 09, Revision A.02, Frisch, M. J.; Trucks, G. W.; Schlegel, H. B.; Scuseria, G. E.; Robb, M. A.; Cheeseman, J. R.; Scalmani, G.; Barone, V.; Mennucci, B.; Petersson, G. A.; Nakatsuji, H.; Caricato, M.; Li, X.; Hratchian, H. P.; Izmaylov, A. F.; Bloino, J.; Zheng, G.; Sonnenberg, J. L.; Hada, M.; Ehara, M.; Toyota, K.; Fukuda, R.; Hasegawa, J.; Ishida, M.; Nakajima, T.; Honda, Y.; Kitao, O.; Nakai, H.; Vreven, T.; Montgomery, Jr., J. A.; Peralta, J. E.; Ogliaro, F.; Bearpark, M.; Heyd, J. J.; Brothers, E.; Kudin, K. N.; Staroverov, V. N.; Kobayashi, R.; Normand, J.; Raghavachari, K.; Rendell, A.; Burant, J. C.; Iyengar, S. S.; Tomasi, J.; Cossi, M.; Rega, N.; Millam, N. J.; Klene, M.; Knox, J. E.; Cross, J. B.; Bakken, V.; Adamo, C.; Jaramillo, J.; Gomperts, R.; Stratmann, R. E.; Yazyev, O.; Austin, A. J.; Cammi, R.; Pomelli, C.; Ochterski, J. W.; Martin, R. L.; Morokuma, K.; Zakrzewski, V. G.; Voth, G. A.; Salvador, P.; Dannenberg, J. J.; Dapprich, S.; Daniels, A. D.; Farkas, Ö.; Foresman, J. B.; Ortiz, J. V.; Cioslowski, J.; Fox, D. J., Gaussian, Inc., Wallingford, CT.
- (33) Perdew, J. P.; Burke, K.; Ernzerhof, M. Phys. Rev. Lett. 1996, 77, 3865.
- (34) Dolg, M.; Wedig, U.; Stoll, H.; Preuss, H. J. Chem. Phys. 1987, 86, 866.
- (35) Ditchfield, R.; Hehre, W. J.; Pople, J. A. J. Chem. Phys. 1971, 54, 724.
- (36) Hariharan, P. C.; Pople, J. A. Mol. Phys. 1974, 27, 209.
- (37) Krishnan, R.; Binkley, J. S.; Seeger, R.; Pople, J. A. J. Chem. Phys. 1980, 72, 650.
- (38) Clark, T.; Chandrasekhar, J.; Spitznagel, G. W.; Schleyer, P. v. R. J. Comput. Chem. 1983, 4, 294.
- (39) Hay, P. J.; Wadt, W. R. J. Chem. Phys. 1985, 82, 299.
- (40) Roy, L. E.; Hay, P. J.; Martin, R. L. J. Chem. Theory Comput. 2008, 4, 1029.
- (41) Reed, A. E.; Curtiss, L. A.; Weinhold, F. Chem. Rev. 1988, 88, 899.
- (42) Carpenter, J. E.; Weinhold, F. *THEOCHEM* **1988**, *46*, 41.
- (43) NBO 5.G, Glendening, E. D.;Badenhoop, J. K.; Reed, A. E.; Carpenter, J. E.; Bohmann, J. A.; Morales, C. M.; Weinhold, F., Theoretical Chemistry Institute, University of Wisconsin, Madison, 2001.

- (44) Collman, J. P.; Hegedus, L. S.; Norton, J. R.; Finke, R. G. *Principles and Applications of Organotransition Metal Chemistry*; 2nd ed.; University Science Books: Sausalito, CA, 1987.
- (45) Landis, C. R.; Firman, T. K.; Root, D. M.; Cleveland, T. J. Am. Chem. Soc. 1998, 120, 1842.
- (46) Huheey, J. E. *Inorganic Chemistry*; 3rd ed.; Harper and Row: New York, 1983.
- (47) One exception was NH<sub>2</sub>, which is non-planar. The H-N-Ir-P dihedral angles were set to 58.5° and 31.5° for "perpendicular" and "parallel" orientations, respectively.
- (48) The largest differences are found in complexes that have a highly distorted unconstrained geometry, typically arising from the introduction of an unconventional substituent (e.g. the unconstrained 1-Li geometry has an optimized P-Ir-P angle of 127°).
- (49) Goldman, A. S.; Krogh-Jespersen, K. J. Am. Chem. Soc. 1996, 118, 12159.
- (50) Riehl, J. F.; Jean, Y.; Eisenstein, O.; Pelissier, M. Organometallics 1992, 11, 729.
- (51) Rachidi, I. E. I.; Eisenstein, O.; Jean, Y. New J. Chem. 1990, 14, 671.
- (52) Jean, Y.; Eisenstein, O. *Polyhedron* **1988**, *7*, 405.
- (53) Sola, E.; García-Camprubí, A.; Andrés, J. L.; Martín, M.; Plou, P. J. Am. Chem. Soc. **2010**, *132*, 9111.
- (54) Choi, J.; MacArthur, A. H. R.; Brookhart, M.; Goldman, A. S. *Chem. Rev.* 2011, *111*, 1761 and references therein.
- (55) Frenking, G.; Froehlich, N. Chem. Rev. 2000, 100, 717.
- (56) Kanzelberger, M.; Singh, B.; Czerw, M.; Krogh-Jespersen, K.; Goldman, A. S. J. *Am. Chem. Soc.* **2000**, *122*, 11017.
- (57) Fan, L.; Parkin, S.; Ozerov, O. V. J. Am. Chem. Soc. 2005, 127, 16772.
- (58) Zhu, Y.; Fan, L.; Chen, C.-H.; Finnell, S. R.; Foxman, B. M.; Ozerov, O. V. *Organometallics* **2007**, *26*, 6701.
- (59) Ben-Ari, E.; Gandelman, M.; Rozenberg, H.; Shimon, L. J. W.; Milstein, D. J. *Am. Chem. Soc.* **2003**, *125*, 4714.
- (60) Ben-Ari, E.; Cohen, R.; Gandelman, M.; Shimon, L. J. W.; Martin, J. M. L.; Milstein, D. *Organometallics* **2006**, *25*, 3190.
- (61) Werner, H.; Höhn, A.; Dziallas, M. Angew. Chem., Int. Ed. Engl. 1986, 25, 1090.
- (62) Rosini, G. P.; Wang, K.; Patel, B.; Goldman, A. S. *Inorg. Chim. Acta* **1998**, *270*, 537.
- (63) Deutsch, P. P.; Eisenberg, R. Chem. Rev. 1988, 88, 1147.
- (64) Janowicz, A. H.; Bergman, R. G. J. Am. Chem. Soc. 1982, 104, 352.
- (65) Bergman, R. G. Science **1984**, 223, 902.
- (66) Vaska, L.; DiLuzio, J. W. J. Am. Chem. Soc. 1962, 84, 679.
- (67) Vaska, L. Acc. Chem. Res. 1968, 1, 335.
- (68) The geometries for these complexes were constrained; the unconstrained complexes were also calculated. The regression equations differ slightly, but the conclusions drawn remain the same regardless of geometry constraints.
- (69) Bursten, B. E.; Chen, S.; Chisholm, M. H. J. Organomet. Chem. 2008, 693, 1547.
- (70) Averkiev, B. B.; Truhlar, D. G. Catal. Sci. Tech. 2011, 1, 1526.

## Chapter 3

## DFT Computational Study of the Oxidative Addition of Chlorobenzene and Fluorobenzene to <sup>R</sup>PCPIr Complexes

#### Abstract

The oxidative addition reactions of chlorobenzene and fluorobenzene to <sup>H</sup>PCPIr, <sup>Me</sup>PCPIr, and <sup>tBu</sup>PCPIr were calculated. The thermodynamics of addition favor activation of the C-H bond proximal to the electron-withdrawing Cl atom ( $\Delta$ G: *para* addition > *meta* addition > *ortho* addition). The kinetic barriers were also calculated and correlated with the relative steric interactions in the transition state leading to the oxidative addition product. Both calculations for thermodynamic and kinetic preferences were found to be in accord with experimental results.

#### **3.1 Introduction**

Regioselective C-H bond activation is an important area of investigation in organometallic chemistry. Many groups have utilized various strategies for achieving regioselective C-H bond activation.<sup>1,2</sup> In a common strategy that was popularized by Murai, precoordination of a metal to a functional group of an arene can lead to the preferential activation of arene C-H bonds at positions *ortho* to the functional group.<sup>3</sup>

The Milstein and Ozerov groups have reported regioselective C-H bond activation of haloarenes by PNP-type pincer iridium complexes.<sup>4,5</sup> Although Milstein has attributed the regioselectivity primarily to haloatom coordination to the metal center,<sup>6</sup> it is likely that electronic effects of the haloatom on the neighboring C-H bonds of the arene also play a role in the observed regioselectivity.<sup>7,8</sup> An electronic basis for regioselectivity follows principles first set forth by Bergman, which indicates the key role electronegativity has on the selectivity of C-H bonds by metal complexes.<sup>9,10</sup>

Our group has likewise been interested in understanding the mechanism of C-H activation to metal complexes, in particular to <sup>tBu</sup>PCPIr pincer complexes (<sup>tBu</sup>PCP =  $\kappa^3$ -C<sub>6</sub>H<sub>3</sub>-2,6-(CH<sub>2</sub>P<sup>t</sup>Bu<sub>2</sub>)<sub>2</sub>).<sup>11-13</sup> The low-valent, 14-electron complex is highly active to oxidative addition of C-H bonds and has been utilized for dehydrogenation and small molecule activation.<sup>14-18</sup> In analogy to the reports from Milstein and Ozerov, our group has been interested in the kinetic and thermodynamic selectivity of haloarene addition to t<sup>tBu</sup>PCPIr. While the experimental work has been performed by David Laviska,<sup>19</sup> this chapter will detail computational investigations into this reaction.

#### **3.2 Basis Set Determination**

All calculations used DFT methodology<sup>20</sup> as implemented in the Gaussian 03 and 09 series of computer programs.<sup>21,22</sup> We employed the Perdew–Burke–Ernzerhof exchange-correlation functional (PBE) in all calculations.<sup>23</sup> We applied the SDD relativistic, small-core effective potential and corresponding basis set to the Ir atom.<sup>24</sup> A survey of various basis set combinations was conducted for non-metal atoms to assess the optimal basis set for this study.<sup>25-27</sup> In the interest of computational speed, the basis sets for atoms more distant from the metal center were generally made smaller than for atoms closer to the metal center. The various basis set combinations can be seen in Table 3.1.

	Basis Set 1	Basis Set 2	Basis Set 3
ir	sdd	sdd	sdd
atoms bonded to Ir	6-31+G(d)	6-311+G(d,p)	6-311+G(d,p)
remaining PCP ligand C's	6-31G <sup>ª</sup>	6-31G <sup>ª</sup>	6-31G
PCP ligand H's	STO-3G	STO-3G	STO-3G
CIPh's H's, remaining C's, and Cl	6-31G <sup>b</sup>	6-311+G(d,p)	6-311+G(d,p)
number of basis functions	382	512	560

**Table 3.1**. Basis set combinations used for DFT calculations.

	Basis Set 4	Basis Set 5	Basis Set 6
ir	sdd	sdd	sdd
atoms bonded to Ir	6-311+G(d,p)	6-311+G(d,p)	6-311+G(d,p) <sup>c</sup>
remaining PCP ligand C's	6-31G	6-31+G	6-31+G
PCP ligand H's	6-31G	6-31+G	6-31+G
CIPh's H's, remaining C's, and Cl	6-311+G(d,p)	6-311+G(d,p)	6-311++G(d,p)
number of basis functions	603	695	700

	Basis Set 7	Basis Set 8	Basis Set 9
ir	sdd	sdd	sdd
atoms bonded to Ir	6-311++G(d,p)	6-311+G(d,p)	6-311+G(d,p)
remaining PCP ligand C's	6-31++G	6-31+G(d)	6-31+G(d,p)
PCP ligand H's	6-31++G	6-31+G(d)	6-31+G(d,p)
CIPh's H's, remaining C's, and Cl	6-311++G(d,p)	6-311+G(d,p)	6-311+G(d,p)
number of basis functions	743	810	939

<sup>&</sup>lt;sup>a</sup>STO-3G for tBu primary C's. <sup>b</sup>STO-3G for H's. <sup>c</sup>6-311++G(d,p) for hydride.

The energy of chlorobenzene oxidative addition to <sup>tBu</sup>PCPIr to yield the *ortho,trans*-isomer (Scheme 3.1) was calculated for each basis set. As can be seen in Table 3.2, increasing basis set size from Basis Set 1 to Basis Set 5 significantly influences  $\Delta E$ ,  $\Delta H$ , and  $\Delta G$ . However, further increases in basis set size beyond that of Basis Set 5 do not appear to significantly impact the calculated energies, despite requiring considerably longer computational time. As a result, Basis Set 5 was used as the basis set for all further calculations.

**Scheme 3.1**. Oxidative addition of chlorobenzene to <sup>tBu</sup>PCPIr to yield the *ortho*, *trans*-isomer.



**Table 3.2**. Calculated parameters for the oxidative addition of chlorobenzene to <sup>tBu</sup>PCPIr

 to yield the *ortho*, *trans*-isomer.

	ΔE (kcal/mol)	ΔH (kcal/mol)	∆S (eu)	ΔG (kcal/mol)
Basis Set 1	-18.3	-19.1	-49.1	-4.5
Basis Set 2	-28.7	-31.3	-49.5	-16.5
Basis Set 3	-15.9	-16.5	-50.1	-1.6
Basis Set 4	-12.2	-12.3	-48.3	2.1
Basis Set 5	-9.8	-9.9	-47.5	4.3
Basis Set 6	-9.8	-9.8	-47.4	4.3
Basis Set 7	-9.9	-10.0	-47.5	4.2
Basis Set 8	-9.8	-9.8	-47.4	4.3
Basis Set 9	-9.8	-9.9	-47.1	4.1

#### 3.3 Results

#### 3.3.1 Chlorobenzene Activation

Addition of chlorobenzene to PCPIr can result in five possible aryl hydride isomers that vary in the location of the Cl atom and the relative orientation of the Cl atom and the hydride (Fig. 3.1). Addition is thought to proceed via initial C-H coordination to form a  $\sigma$ -complex, followed by oxidative cleavage to yield the aryl hydride product (Scheme 3.2).<sup>6</sup> The barrier to formation of the  $\sigma$ -complex is generally considered to be very small or non-existent; this TS was not calculated. Similarly, the energies of C-H coordination are similar among the isomers, with the exception of the *ortho*, *trans* isomer, which could not be located. Geometry optimization of the *ortho*, *trans* isomer results in the migration of the Ir from the side of the Cl atom substituent to the opposite side, which corresponds to the *ortho*, *cis* isomer. It is unclear from computation whether the potential energy surface for the *ortho*, *cis* C-H coordination complex is extremely shallow, or whether that intermediate does not exist and formation of the *ortho*, *trans* isomer proceeds from the C-H complex that is more generally assigned to the *ortho*, *cis* isomer.

Figure 3.1. Isomers of the oxidative addition of chlorobenzene to PCPIr.





The energies for the addition of chlorobenzene to <sup>H</sup>PCPIr, <sup>Me</sup>PCPIr, and <sup>tBu</sup>PCPIr were calculated (Table 3.3). The thermodynamics of chlorobenzene addition to <sup>H</sup>PCPIr and <sup>Me</sup>PCPIr favor activation of the C-H bond proximal to the electron-withdrawing Cl atom ( $\Delta G$ : *para* addition > *meta* addition > *ortho* addition), analogous to previously reported results.<sup>10,28,29</sup> The difference in energies between the *meta* isomers is very small ( $\Delta\Delta G < 0.3$  kcal/mol), which is consistent with the Cl being directed away from the metal center and incurring minimal interactions with the metal center and PCP ligand. By contrast, the energy difference between the *ortho* isomers is considerably larger ( $\Delta\Delta G$ > 4.4 kcal/mol), consistent with interactions of the Cl atom occurring with the ligand substituents or the Ir atom. For the *ortho* isomers, the *ortho*, *cis*-isomer is higher in energy than the *ortho*, *trans*-isomer, due to unfavorable steric interactions between the bulky Cl atom and the hydride. Additionally, the *ortho*, *trans*-isomer may benefit from interactions between the Ir atom and the Cl atom.

However, the calculated  $\Delta G$  values for <sup>tBu</sup>PCPIr do not follow this trend, contrary to experimental results.<sup>19</sup> While the calculated  $\Delta G$  values for the *meta* and *para* isomers of chlorobenzene addition to <sup>tBu</sup>PCPIr follow the same trends observed for <sup>H</sup>PCPIr and <sup>Me</sup>PCPIr, the values for the *ortho* isomers for <sup>tBu</sup>PCPIr are much larger than expected, such that they are no longer the thermodynamically favored products. This discrepancy likely arises due to the calculations overstating the steric interactions between the Cl atom and the phosphine substituents of the *ortho* isomers, which are largest for <sup>tBu</sup>PCPIr.

**Table 3.3**. Calculated  $\Delta G$  (kcal/mol) for the various complexes in the reaction coordinate for the oxidative addition of chlorobenzene to <sup>tBu</sup>PCPIr, <sup>Me</sup>PCPIr, and <sup>H</sup>PCPIr.

		ΔG			
R = tBu	R4PCPIr + CIPh	C-H $\sigma$ -complex	TS (barrier to OA)	aryl hydride	barrier to RE
2-CIPh-Cltrans	0.0	not located	23.4	4.2	19.2
2-CIPh-Clcis	0.0	11.7	18.1	5.5	12.6
3-CIPh-Cltrans	0.0	10.3	17.1	2.9	14.2
3-CIPh-Clcis	0.0	11.1	16.9	3.3	13.6
4-CIPh	0.0	11.4	18.1	4.2	13.9
R = Me	R4PCPIr + CIPh	C-H σ-complex	TS	aryl hydride	barrier to RE
2-CIPh-Cltrans	0.0	not located	7.5	-10.4	17.9
2-CIPh-Clcis	0.0	2.7	3.6	-6.0	9.6
3-CIPh-Cltrans	0.0	3.2	2.1	-4.5	6.6
3-CIPh-Clcis	0.0	3.4	2.3	-4.2	6.5
4-CIPh	0.0	3.5	2.7	-2.8	5.5
R = H	R4PCPIr + CIPh	C-H σ-complex	TS	aryl hydride	barrier to RE
2-CIPh-Cltrans	0.0	not located	4.6	-7.9	12.5
2-CIPh-Clcis	0.0	1.7	3.1	-2.7	5.8
3-CIPh-Cltrans	0.0	0.3	1.4	-1.2	2.6
3-CIPh-Clcis	0.0	0.4	1.2	-1.3	2.5
4-CIPh	0.0	0.7	1.8	-0.4	2.2

ΔG

The differences in steric interactions between the *ortho*, *cis*-isomer and the *ortho*, *trans*-isomer can be seen in the DFT calculated geometries of the 5-coordinate aryl hydride complexes (Fig. 3.2). The *ortho*, *trans*-isomer displays a square pyramidal geometry. The aryl group of the *ortho*, *trans*-isomer displays normal coordination; the angles of the two carbons (C2, C6) ortho to the ipso carbon (C1) relative to the Ir atom are essentially identical (124.1° and 123.3°, respectively). By contrast, the *ortho*, *cis*-isomer displays rotation of the aryl group consistent with the presence of steric interactions between the chlorine atom and the hydride. The aryl group displays rotation about the ipso carbon (C1) such that the Cl atom is found farther away from the hydride. The angles Ir-C2-C6 (129.5°) and Ir-C1-C6 (117.9°) clearly illustrate this rotation. Additionally, the distance between the hydride and the chlorine atom (2.588 Å) is smaller than the sum of the van der Waals radii for hydrogen and chlorine (2.95 Å), suggesting steric interactions between the atoms.<sup>30</sup>

**Figure 3.2**. Abbreviated DFT-optimized geometries of *ortho*, *trans*-Cl (**a**) and *ortho*, *cis*-Cl (**b**) complexes. Selected distances (Å) and angles (°): (**a**) C<sub>PCP,ipso</sub>-Ir-H, 89.0; H-Ir-C1, 89.0; Ir-C1-C2, 124.1; Ir-C1-C6, 123.3; Ir-Cl, 3.441. (**b**) C<sub>PCP,ipso</sub>-Ir-H, 100.8; H-Ir-C1, 84.7; Ir-C1-C2, 129.5; Ir-C1-C6, 117.9; Ir-C1, 2.588.



Kinetically, the transition state leading to the *ortho*, *cis*-isomer is lower in energy than the transition state leading to the *ortho*, *trans*-isomer. This result can be rationalized by examining the approach trajectory of the metal center to the ortho C-H bond. As shown in Figure 3.3, a trajectory in which the Ir atom approaches the C-H bond on the side opposite to the Cl substituent will lead to the *ortho*, *cis*-isomer; conversely, a trajectory in which the Ir atom approaches the C-H bond on the same side as the Cl substituent will lead to the *ortho*, *trans*-isomer. In TS<sub>ortho,cis</sub>, the Cl atom is directed farther away from the metal center than in TS<sub>ortho,trans</sub>, which results in a decrease in unfavorable steric interactions between the Cl atom and the phosphine substituents of the PCP ligand. Comparing <sup>H</sup>PCPIr, <sup>Me</sup>PCPIr, and <sup>tBu</sup>PCPIr, the bulk of the phosphine substituents successively increases, and the  $\Delta\Delta G^{\ddagger}$  between the transition states leading to the two *ortho* isomers correspondingly increases.

Figure 3.3. Transition state interactions leading to (a) the *ortho*, *cis*-isomer and (b) the *ortho*, *trans*-isomer.



Presumably, interconversion of isomers proceeds via reductive coupling to give the C-H coordination complex, whereupon the complex decoordinates or "walks" along the chlorobenzene ring to another C-H bond.<sup>31-34</sup> Direct interconversion of isomers via rotation of the aryl ring about the Ir-ipsoC bond was calculated to be too high in energy to be viable ( $\Delta G^{\ddagger}_{ortho, cis \rightarrow ortho, trans} = 41.3$  kcal/mol).

Coordination of chlorobenzene to 3-coordinate PCPIr through the Cl atom was also calculated for <sup>tBu</sup>PCPIr. While the energy of Cl coordination is calculated to be lower than the formation of the C-H coordination complex ( $\Delta G = 6.5$  kcal/mol vs. 11 kcal/mol), oxidative addition is still considerably more favorable than Cl coordination, which suggests that Cl coordination may be kinetically relevant but not thermodynamically relevant. Experimental confirmation of this complex is likely to be difficult, since the Cl coordination complex is expected to be transient and would lack the diagnostic hydride signal that accompanies C-H oxidative addition.

#### 3.3.2 Fluorobenzene Activation

The oxidative addition of fluorobenzene to <sup>H</sup>PCPIr, <sup>Me</sup>PCPIr, and <sup>tBu</sup>PCPIr was calculated (Table 3.4). Again, the C-H coordination complex leading to the *ortho*, *trans*-isomer was not located for any of the three metal-ligand complexes calculated. Additionally, the transition state for the *ortho*, *trans*-isomer in the <sup>tBu</sup>PCPIr complex was not located.

		∆G (kc	al/mol)		
R = tBu	R4PCPIr + FPh	C-H σ-complex	TS (barrier to OA)	aryl hydride	barrier to RE
2-FPh-Ftrans	0.0	not located	18.5	0.4	18.1
2-FPh-Fcis	0.0	11.3	17.5	3.3	14.2
3-FPh-Ftrans	0.0	9.8	17.1	4.5	12.6
3-FPh-Fcis	0.0	10.7	17.6	4.3	13.3
4-FPh	0.0	10.7	18.0	5.0	13.0
R = Me	R4PCPIr + FPh	C-H σ-complex	TS (barrier to OA)	aryl hydride	barrier to RE
2-FPh-Ftrans	0.0	not located	4.6	-8.0	12.6
2-FPh-Fcis	0.0	3.4	1.6	-6.4	8.0
3-FPh-Ftrans	0.0	1.6	2.3	-3.6	5.9
3-FPh-Fcis	0.0	3.6	2.8	-3.6	6.4
4-FPh	0.0	3.8	4.8	-2.7	7.5
R = H	R4PCPIr + FPh	C-H σ-complex	TS (barrier to OA)	aryl hydride	barrier to RE
2-FPh-Ftrans	0.0	0.1	not located	-4.8	#VALUE!
2-FPh-Fcis	0.0	0.1	0.9	-3.7	4.6
3-FPh-Ftrans	0.0	0.3	1.8	-1.1	2.9

**Table 3.4**. Calculated  $\Delta G$  (kcal/mol) for the various complexes in the reaction coordinate for the oxidative addition of fluorobenzene to <sup>tBu</sup>PCPIr, <sup>Me</sup>PCPIr, and <sup>H</sup>PCPIr.

The general trends for the thermodynamics of fluorobenzene oxidative addition proceed in exactly the same fashion as seen for chlorobenzene. *Ortho* addition is more favorable than *meta*, which is more favorable than *para* addition. The *ortho*, *trans* isomers were calculated to be lower in energy than the *ortho*, *cis* isomers.

1.6

2.6

-1.0

0.4

2.6

2.2

0.4

0.7

3-FPh-Fcis

4-FPh

0.0

0.0

Kinetically, the barriers for fluorobenzene oxidative addition also correlate with the extent of steric interactions at the transition state. The fluorine atom is smaller than chlorine ( $R_{F,van} der Waals = 1.47$  Å vs.  $R_{Cl, van} der Waals = 1.75$  Å) and thus suffers fewer unfavorable steric interactions with the phosphine substituents of the pincer ligand than the corresponding chlorobenzene complexes.<sup>30</sup> As such, the kinetic barriers for the oxidative addition of fluorobenzene were smaller than for the analogous chlorobenzene oxidative addition, but remain in the same relative order. This result serves to further

emphasize the role of steric interactions in determining the kinetic barrier to oxidative addition.

#### 3.3.3 Comparison to Experimental Data

The calculated results were found to be in complete agreement with experimental data.<sup>19</sup> Experiments performed by David Laviska in our lab demonstrated that the *ortho* isomers were thermodynamically more favorable than the *meta* and *para* isomers. Of the *ortho* isomers, the less favorable kinetic product (*ortho*,*trans*) was found to be the more favorable thermodynamic product. Assignment of the isomers was confirmed through a series of NMR experiments, which showed possible metal-halide interactions for the *ortho*,*trans*-isomers.

#### 3.4 Summary

The oxidative addition of chlorobenzene and fluorobenzene to <sup>H</sup>PCPIr, <sup>Me</sup>PCPIr, and <sup>tBu</sup>PCPIr were calculated. The thermodynamics of addition were found to favor activation of the C-H bond proximal to the electron-withdrawing Cl atom ( $\Delta G$ : *para* addition > *meta* addition > *ortho* addition). The kinetic barriers were also calculated and correlated with the relative steric interactions in the transition state leading to the oxidative addition product. Both calculations for thermodynamic and kinetic preferences were in accord with experimental results.

#### References

- (1) Neufeldt, S. R.; Sanford, M. S. Acc. Chem. Res. 2012, 45, 936.
- (2) Shul'pin, G. B. Org. Biomol. Chem. 2010, 8, 4217.
- (3) Murai, S.; Kakiuchi, F.; Sekine, S.; Tanaka, Y.; Kamatani, A.; Sonoda, M.; Chatani, N. *Nature* **1993**, *366*, 529.
- (4) Ben-Ari, E.; Gandelman, M.; Rozenberg, H.; Shimon, L. J. W.; Milstein, D. J. *Am. Chem. Soc.* **2003**, *125*, 4714.
- (5) Fan, L.; Parkin, S.; Ozerov, O. V. J. Am. Chem. Soc. 2005, 127, 16772.
- (6) Ben-Ari, E.; Cohen, R.; Gandelman, M.; Shimon, L. J. W.; Martin, J. M. L.; Milstein, D. *Organometallics* **2006**, *25*, 3190.
- (7) Tanabe, T.; Brennessel, W. W.; Clot, E.; Eisenstein, O.; Jones, W. D. *Dalton Trans.* **2010**, *39*, 10495.
- (8) Bennett, J. L.; Wolczanski, P. T. J. Am. Chem. Soc. 1997, 119, 10696.
- (9) Bergman, R. G. Science **1984**, 223, 902.
- (10) Bryndza, H. E.; Fong, L. K.; Paciello, R. A.; Tam, W.; Bercaw, J. E. J. Am. Chem. Soc. 1987, 109, 1444.
- (11) Krogh-Jespersen, K.; Czerw, M.; Kanzelberger, M.; Goldman, A. S. J. Chem. Inf. Comput. Sci. 2001, 41, 56.
- (12) Krogh-Jespersen, K.; Czerw, M.; Summa, N.; Renkema, K. B.; Achord, P. D.; Goldman, A. S. *J. Am. Chem. Soc.* **2002**, *124*, 11404.
- (13) Zhang, X.; Kanzelberger, M.; Emge, T. J.; Goldman, A. S. J. Am. Chem. Soc. **2004**, *126*, 13192.
- (14) Liu, F.; Pak, E. B.; Singh, B.; Jensen, C. M.; Goldman, A. S. J. Am. Chem. Soc. 1999, 121, 4086.
- (15) Kanzelberger, M.; Singh, B.; Czerw, M.; Krogh-Jespersen, K.; Goldman, A. S. J. Am. Chem. Soc. 2000, 122, 11017.
- (16) Zhang, X.; Emge, T. J.; Ghosh, R.; Goldman, A. S. J. Am. Chem. Soc. 2005, 127, 8250.
- (17) Choi, J.; Choliy, Y.; Zhang, X.; Emge, T. J.; Krogh-Jespersen, K.; Goldman, A. S. *J. Am. Chem. Soc.* **2009**, *131*, 15627.
- (18) Choi, J.; Wang, D. Y.; Kundu, S.; Choliy, Y.; Emge, T. J.; Krogh-Jespersen, K.; Goldman, A. S. *Science* **2011**, *332*, 1545.
- (19) Laviska, D. Rutgers, The State University of New Jersey, New Brunswick, NJ. Unpublished work, 2012.
- (20) Parr, R. G.; Yang, W. *Density-Functional Theory of Atoms and Molecules*; University Press: Oxford, 1989.
- (21) Gaussian 03, Revision C.02, Frisch, M. J.; Trucks, G. W.; Schlegel, H. B.;
  Scuseria, G. E.; Rob, M. A.; Cheeseman, J. R.; Jr., J. A. M.; Vreven, T.; Kudin, K. N.; Burant, J. C.; J. M. Millam; Iyengar, S. S.; Tomasi, J.; , V. B.; , B. M.; M. Cossi; G. Scalmani; N. Rega; G. A. Petersson; H. Nakatsuji; M. Hada; M. Ehara; K. Toyota; Fukuda, R.; , J. H.; , M. I.; T. Nakajima; Y. Honda; O. Kitao; Nakai, H.; , M. K.; X. Li, J. E. K.; H. P. Hratchian; J. B. Cross; V. Bakken; C. Adamo; J. Jaramillo; R. Gomperts; R. E. Stratmann; O. Yazyev; A. J. Austin; R. Cammi; C. Pomelli, J.; W. Ochterski; P. Y. Ayala; K. Morokuma; G. A. Voth; P. Salvador; J. J. Dannenberg; Zakrzewski, V. G.; , S. D.; A. D. Daniels; M. C. Strain; O. Farkas;

D. K. Malick; A. D. Rabuck; K. Raghavachari; J. B. Foresman; J. V. Ortiz; Q. Cui; A. G. Baboul; S. Clifford; J. Cioslowski; B. B. Stefanov; G. Liu; A. Liashenko; P. Piskorz; I. Komaromi; R. L. Martin; D. J. Fox; T. Keith; Al-Laham, M. A.; , C. Y. P.; A. Nanayakkara; M. Challacombe; P. M. W. Gill; B. Johnson; W. Chen; M. W. Wong; C. Gonzalez; Pople, J. A., Gaussian, Inc., Wallingford, CT.

- (22) Gaussian 09, Revision A.02, Frisch, M. J.; Trucks, G. W.; Schlegel, H. B.; Scuseria, G. E.; Robb, M. A.; Cheeseman, J. R.; Scalmani, G.; Barone, V.; Mennucci, B.; Petersson, G. A.; Nakatsuji, H.; Caricato, M.; Li, X.; Hratchian, H. P.; Izmaylov, A. F.; Bloino, J.; Zheng, G.; Sonnenberg, J. L.; Hada, M.; Ehara, M.; Toyota, K.; Fukuda, R.; Hasegawa, J.; Ishida, M.; Nakajima, T.; Honda, Y.; Kitao, O.; Nakai, H.; Vreven, T.; Montgomery, Jr., J. A.; Peralta, J. E.; Ogliaro, F.; Bearpark, M.; Heyd, J. J.; Brothers, E.; Kudin, K. N.; Staroverov, V. N.; Kobayashi, R.; Normand, J.; Raghavachari, K.; Rendell, A.; Burant, J. C.; Iyengar, S. S.; Tomasi, J.; Cossi, M.; Rega, N.; Millam, N. J.; Klene, M.; Knox, J. E.; Cross, J. B.; Bakken, V.; Adamo, C.; Jaramillo, J.; Gomperts, R.; Stratmann, R. E.; Yazyev, O.; Austin, A. J.; Cammi, R.; Pomelli, C.; Ochterski, J. W.; Martin, R. L.; Morokuma, K.; Zakrzewski, V. G.; Voth, G. A.; Salvador, P.; Dannenberg, J. J.; Dapprich, S.; Daniels, A. D.; Farkas, Ö.; Foresman, J. B.; Ortiz, J. V.; Cioslowski, J.; Fox, D. J., Gaussian, Inc., Wallingford, CT.
- (23) Perdew, J. P.; Burke, K.; Ernzerhof, M. Phys. Rev. Lett. 1996, 77, 3865.
- (24) Dolg, M.; Wedig, U.; Stoll, H.; Preuss, H. J. Chem. Phys. 1987, 86, 866.
- (25) Ditchfield, R.; Hehre, W. J.; Pople, J. A. J. Chem. Phys. 1971, 54, 724.
- (26) Hariharan, P. C.; Pople, J. A. Mol. Phys. 1974, 27, 209.
- (27) Krishnan, R.; Binkley, J. S.; Seeger, R.; Pople, J. A. J. Chem. Phys. 1980, 72, 650.
- (28) Evans, M. E.; Burke, C. L.; Yaibuathes, S.; Clot, E.; Eisenstein, O.; Jones, W. D. J. Am. Chem. Soc. 2009, 131, 13464.
- (29) Clot, E.; Besora, M.; Maseras, F.; Megret, C.; Eisenstein, O.; Oelckers, B.; Perutz, R. N. Chem. Commun. 2003, 490.
- (30) Bondi, A. J. Phys. Chem. **1964**, 68, 441.
- (31) Strawser, D.; Karton, A.; Zenkina, O. V.; Iron, M. A.; Shimon, L. J. W.; Martin, J. M. L.; van der Boom, M. E. *J. Am. Chem. Soc.* **2005**, *127*, 9322.
- (32) Harman, W. D. Coord. Chem. Rev. 2004, 248, 853.
- (33) Northcutt, T. O.; Wick, D. D.; Vetter, A. J.; Jones, W. D. J. Am. Chem. Soc. 2001, 123, 7257.
- (34) Carbo, J. J.; Eisenstein, O.; Higgitt, C. L.; Klahn, A. H.; Maseras, F.; Oelckers, B.; Perutz, R. N. J. Chem. Soc., Dalton Trans. 2001, 1452.

## Chapter 4

## Investigation of Bonding and Through-Space Effects on CO Stretching Frequencies

#### Abstract

Calculations were performed to investigate the contributions of backbonding and through-space effects on the CO stretching frequency. CO stretching frequencies for a series of Ni(CO)<sub>3</sub>L complexes were calculated using DFT calculations. NBO analysis of these complexes yielded parameters that reflected backbonding to the  $\pi^*$  orbitals of the CO ligands and polarization of the  $\pi^b$  orbitals of the CO ligands. We found that a linear two-variable model fit the data better than the traditional one-variable model of solely back-bonding effects. A close look at rotamers of anisolyl phosphine complexes highlighted the dominant effect of polarization in some cases. In the course of our investigation, we also calculated the effects of a point charge on free CO, which was consistent with previous work.

#### 4.1 Introduction

Carbon monoxide C-O stretching frequencies ( $v_{CO}$ ) have long been used to assess the electronic environment of a metal center in organometallic complexes. The perturbation of C-O stretching frequencies has been traditionally rationalized as a result of M $\rightarrow$ CO  $\pi$ -back-bonding, where more electron-rich metal centers exhibit a greater extent of  $\pi$ -back-bonding and thus induce a decrease in  $v_{CO}$ .<sup>1</sup> As such,  $v_{CO}$  has been used as a parameter in the determination of several ligand electronic series.<sup>2-5</sup> Tolman parameterized the electronic effects of phosphorus ligands using the frequency of the A<sub>1</sub> carbonyl vibrational mode of Ni(CO)<sub>3</sub>L in CH<sub>2</sub>Cl<sub>2</sub>,<sup>2</sup> and this approach has been extended to different ligand types and complexes.<sup>3,4</sup>

While Tolman defined the electronic effect as a change in molecular properties by through-bond interactions, several groups have investigated the perturbation of  $v_{CO}$  by through-space effects.<sup>6-12</sup> Notably, their work has shown that placement of bound CO in an electrostatic field induces a change in  $v_{CO}$ .<sup>10</sup> These through-space effects can be most easily understood by looking at the nature of the bonding orbitals in free CO. The bonding orbitals of free CO are polarized toward the oxygen atom. Introduction of a positive charge near the carbon atom would oppose this polarization, increase the covalency of the molecule, and increase the value of the stretching force constant.<sup>6</sup> When depicted as valence bond structures (Figure 4.1), a positive charge near the C-terminus should preferentially stabilize the triply-bonded resonance form (I).

Figure 4.1. Resonance forms of CO.

$$: \overset{\circledast}{\mathsf{C}} = \overset{\circledast}{\mathsf{O}}: \checkmark \rightarrow : \mathsf{C} = \overset{\circledast}{\mathsf{O}}:$$

In the course of our group's work, we have found that some of our organometallic complexes exhibit C-O stretching frequencies that are not easily explained by traditional  $M \rightarrow CO \pi$ -back-bonding rationales.<sup>13</sup> Instead, we have found that a consideration of through-space effects gives a better *prima facie* explanation.<sup>13</sup> Herein, we report calculations of several organometallic carbonyl complexes that aim to separate and quantify through-space and through-bond factors that influence  $v_{CO}$ . The data suggests that while through-bond effects ( $M \rightarrow CO \pi$ -back-bonding) are significant, through-space effects (electrostatic effects) are also significant and, in some cases, the dominant effects governing the perturbation of  $v_{CO}$ .

#### 4.2 Computational Methods

#### 4.2.1 Computational Details

All calculations used DFT methodology<sup>14</sup> as implemented in the Gaussian03, version C.02, series of computer programs.<sup>15</sup> We have made use of the Perdew–Burke– Ernzerhof exchange-correlation functional (PBE).<sup>16</sup> The relativistic, small-core ECP and corresponding basis sets of Dolg et al. (SDD model) were used for all metal atoms (Ir, Ni, Co, V, Cr).<sup>17</sup> We used split valence plus polarization and diffuse functions 6-31+G(d) basis sets<sup>18-21</sup> for all carbonyls and all atoms directly bonded to the metal center in organometallic complexes. Split valence plus polarization function 6-31G(d) basis sets were used for all remaining atoms. All complexes underwent initial geometry optimization and subsequent frequency calculation. To increase numerical accuracy, the complexes were then subjected again to geometry optimization using the analytical force constants obtained from the initial frequency calculation. Electronic population analysis (natural bond order analysis<sup>22,23</sup>) and a final frequency calculation were performed on the re-optimized geometries.

#### 4.2.2 Natural Bond Orbital Analysis

In order to evaluate the separate contributions of through-space and through-bond effects, we utilized Natural Bond Orbital (NBO) analysis as implemented in the NBO 3.1 software package.<sup>22,23</sup> Very generally speaking, NBO analysis proceeds by initially generating a basis set of atom-localized orbitals. This basis set of atomic orbitals is then used to generate localized electron-pair orbitals for Lewis-like bonding, antibonding, and lone-pair orbitals. In this manner, NBO analysis allows the determination of orbital occupancy (population analysis) as well as the determination of the percentage contribution of each atomic orbital that constitutes the natural bond orbital.

When using this computational model in the context of evaluating perturbations to the CO molecule, through-bond effects (namely  $\pi$ -back-bonding) should principally influence the occupancy of the carbonyl  $\pi^*$  orbitals. Complexes with greater  $\pi$ -backbonding should see higher orbital occupancy values for the carbonyl  $\pi^*$  orbitals. Through-space effects (electrostatic effects) should induce changes in polarization and should be reflected in the atomic orbital composition of the bonding carbonyl  $\pi^b$  orbitals. Thus, in systems where through-space effects are the only factor, the orbital occupancies of the carbonyl  $\pi^b$  and  $\pi^*$  orbitals should remain constant (e.g. essentially 2.0 and 0.0, respectively, for free CO) while the percent-composition of the  $\pi^b$  orbitals coming from their constituent carbon and oxygen p-orbitals should change.

#### 4.3 Calculation of Selected Tolman-type Complexes

A series of vibrational frequency and NBO calculations were performed on complexes of the general form Ni(CO)<sub>3</sub>L, where L is a two-electron donor ligand. These pseudo-tetrahedral complexes include many of the phosphine ligands used by Tolman in his review<sup>2</sup> and also include both anionic and cationic ligands. The calculated stretching frequencies (Table 4.1) show similar trends when compared to literature stretching frequencies, though the absolute values are about 25 cm<sup>-1</sup> smaller. A similar computational method applied by Crabtree *et al.* shows a similar general trend, and their calculated values also deviate slightly from experimental values.<sup>3</sup>

	Literature	Calculated	$\pi^*$ Orbital	$\%$ of $\pi^{\rm D}$ on the	-	Literature	Calculated	$\pi^*$ Orbital	% of $\pi^{\rm D}$ on the
Ligand	<sup>v</sup> co	vco	Occupancy	Carbon Atom	Ligand	Vco	vco	Occupancy	Carbon Atom
	(cm <sup>-1</sup> )	(cm <sup>-1</sup> )	(backbonding effects)	(polarization effects)		(cm <sup>-1</sup> )	(cm <sup>-1</sup> )	(backbonding effects)	(polarization effects)
PtBu <sub>3</sub>	2056.1	2031 4	0.19989	25.68	SIMe (NHC)		2032.8	0.19923	25.65
PEt <sub>3</sub>	2061.7	2039.7	0.20176	25.47	SIPr (NHC)	2052.2 <sup>°</sup>	2028.8	0.21020	25.23
P(NMe <sub>2</sub> ) <sub>3</sub>	2061.9	2039.9	0.19880	25.52	SIMes (NHC)	2051.5	2029.3	0.20839	25.25
PMe <sub>3</sub>	2064.1	2043.8	0.19868	25.50	BIMe (NHC)		2034.5	0.19782	25.68
PPh <sub>3</sub>	2068.9	2044.2	0.19626	25.53	OH2		2065.7	0.17358	25.88
P(vinyl) <sub>3</sub>	2069.5	2046.8	0.19379	25.57	OMe <sub>2</sub>		2061.5	0.17461	25.82
NMe <sub>3</sub>	2067.6 <sup>b</sup>	2047.1	0.18905	25.63	0Ph <sub>2</sub>		2065.9	0.16267	26.04
PHMe <sub>2</sub>	2069.6	2049.5	0.19482	25.55	FMe		2076.4	0.16031	26.01
PPh <sub>2</sub> OMe	2072.0	2046.1	0.19607	25.60	FPh		2079.5	0.15562	26.10
NH <sub>3</sub>	2073.3 <sup>b</sup>	2052.4	0.18926	25.64	CIMe		2068.2	0.17159	25.83
PH <sub>2</sub> Me	2075.3	2055.7	0.18952	25.62	CIPh		2068.1	0.17007	25.62
PH <sub>2</sub> Ph	2077.0	2055.9	0.18909	25.64	MeCN		2058.2	0.17683	25.54
P(OMe) <sub>3</sub>	2079.5	2054.3	0.19414	25.74	MeNC		2058.4	0.18033	25.56
PCIPh2	2080.7	2055.6	0.18533	25.68	pyridine		2048.5	0.18526	25.62
PMe <sub>2</sub> CF <sub>3</sub>	2080.9	2056.9	0.18826	25.65	pyridine with o -F		2051.5	0.18000	25.67
PH <sup>3</sup>	2083.2	2062.7	0.18341	25.69	pyridine with m-F		2050.5	0.18326	25.67
PCI <sub>2</sub> Ph	2092.1	2068.5	0.17435	25.85	pyridine with p -F		2049.7	0.18437	25.65
PH2F	2090.9	2071.4	0.17795	25.84	pyridine with bis-m-F		2052.4	0.18305	25.68
PHF2	2100.8	2079.5	0.17310	25.96	PbicycloB		2044.5	0.19656	25.52
PCI3	2097 (2107 <sup>b</sup> )	2081.6	0.16111	26.05	PadamantB3		2047.6	0.18346	25.73
PF3	2110.8	2088.4	0.16970	26.04	NbicycloB		2045.0	0.18888	25.63
PHPh2	2073.3	2049.6	0.19253	25.58	CbicycloNH		1996.4	0.22376	25.10
PPh <sub>2</sub> (anisolyl), o -OMe toward Ni	2066.1	2039.1	0.19685	25.41	PbicydoBH-		2004.0	0.22469	24.98
PPh₂(anisolyl), <i>o</i> -OMe away from Ni	2066.1	2040.6	0.19681	25.52	NbicycloBH-		2009.0	0.21719	24.78
P(anisolyl) <sub>3</sub> , <i>o</i> -OMe, 0 OMe toward Ni	2058.3	2037.5	0.19928	25.47	CbicycloN-		1954.3	0.25060	24.55
P(anisolyl) <sub>3</sub> , <i>o</i> -OMe, 1 OMe toward Ni	2058.3	2035.4	0.19935	25.36	Br <sup>-</sup>		2006.8	0.21874	24.45
P(anisolyl) <sub>3</sub> , o-OMe, 2 OMe toward Ni	2058.3	2032.7	0.19829	25.24	cl <sup>-</sup>		2006.4	0.21785	24.38
P(anisolyl) <sub>3</sub> , o-OMe, 3 OMe toward Ni	2058.3	2028.6	0.19683	25.10	Ŀ		1993.5	0.22550	24.25
P(anisolyl) <sub>3</sub> , <i>m</i> -OMe, 0 OMe toward Ni		2043.2	0.19619	25.50	Η		1964.5	0.26307	24 59
P(anisolyl) <sub>3</sub> , <i>m</i> -OMe, 1 OMe toward Ni		2041.9	0.19746	25.49	Ph <sup>-</sup>		1972 4	0.23386	24.80
P(anisolyl) <sub>3</sub> , <i>m</i> -OMe, 2 OMe toward Ni		2040.1	0.19850	25.50	Me		1957.7	0.25189	24.52
P(anisolyl) <sub>3</sub> , <i>m</i> -OMe, 3 OMe toward Ni		2038.2	0.19960	25.50	NMe <sub>2</sub>		1969.5	0.24254	24.35
P(anisolyl) <sub>3</sub> with <i>p</i> -OMe	2066.1	20387	0.19926	25.46	MeO <sup>-</sup>		1983.0	0.23423	24.26
IMe (NHC)		2031.8	0.20922	25.33	OH <sup>T</sup>		1982.3	0.23660	24.22
IPr (NHC)	2051.5 <sup>°</sup>	2027.8	0.20955	25.18	CN <sup>-</sup>		2006.6	0.22098	24.70
ICy (NHC)	2049.6	2027.2	0.20711	25.43	Ni(CO) <sub>4</sub> with 1 heavy CO		2087.0	0.16257	25.94
IMes (NHC)	2050.7	2028.5	0.20904	25.24	Co(CO)4 With 1 heavy CO		1955.2	0.27322	24.08
$\ensuremath{^{\text{b}}}\xspace$ contractions of the twist of the transformation of transformation	5.								

**Table 4.1.** Computational Results for Ni(CO) $_3L$  Complexes: Calculated andExperimental  $v_{CO}$  Values and NBO Analysis.

A plot of calculated  $v_{CO}$  vs. average  $\pi^*$  orbital occupancy (Figure 4.2) does show a strong linear correlation between  $v_{CO}$  and average  $\pi^*$  orbital occupancy. This plot, consistent with the Tolman-type backbonding explanation for perturbations to the CO stretching frequency, should be linear with a negative slope; as backbonding increases, the CO stretching frequency should decrease. The coefficient of determination for a least-squares one-variable linear regression ( $R^2 = 0.947$ ) indicates a strong correlation between backbonding and CO stretching frequency.

**Figure 4.2.** Plot of calculated  $v_{CO}$  vs. average CO  $\pi^*$  orbital occupancy for Ni(CO)<sub>3</sub>L Complexes.





Similarly, the CO frequency is also expected to correlate with orbital polarization. Polarization was quantified by percent-composition of the CO  $\pi^{b}$  orbitals coming from their constituent carbon atom. A plot of calculated  $v_{CO}$  vs. this measure of polarization (Figure 4.3) does show a linear correlation between  $v_{CO}$  and average  $\pi^{*}$  orbital occupancy. While the coefficient of determination for a least-squares one-variable linear regression ( $R^{2} = 0.861$ ) is not high, it does indicate some correlation between polarization and CO stretching frequency.

**Figure 4.3.** Plot of calculated  $v_{CO}$  vs. average percentage of CO  $\pi^{b}$  orbital from the C atom for Ni(CO)<sub>3</sub>L Complexes.





Phosphine ligands bearing anisolyl substituents highlight the contribution of through-space effects (Table 4.2). These ligands can exist in different rotamer conformations, where the methoxy moieties are oriented proximal or distal to the carbonyl groups. Within both the anisolyldiphenylphosphine and trianisolylphosphine series, the extent of back-bonding is relatively constant in both cases. By contrast, the polarization changes depending on the proximal or distal orientation of the methoxy moieties to the carbonyl groups. Given more methoxy moieties oriented farther away from the carbonyl groups, the extent of polarization toward the carbon increases, and  $v_{CO}$  similarly increases.

Ligand	Calculated v <sub>co</sub> (cm <sup>-1</sup> )	π* Orbital Occupancy (back-bonding effects)	% of π <sup>b</sup> on the Carbon Atom (polarization effects)
PPh <sub>2</sub> (anisolyl), <i>o</i> -OMe away from Ni	2040.6	0.19681	25.52
$PPh_2(anisolyl)$ , <i>o</i> -OMe toward Ni	2039.1	0.19685	25.41
P(anisolyl) <sub>3</sub> , <i>o</i> -OMe, 0 OMe toward <b>N</b> i	2037.5	0.19928	25.47
P(anisolyl) <sub>3</sub> , <i>o</i> -OMe, 1 OMe toward Ni	2035.4	0.19935	25.36
P(anisolyl) <sub>3</sub> , <i>o</i> -OMe, 2 OMe toward Ni	2032.7	0.19829	25.24
P(anisolyl) <sub>3</sub> , <i>o</i> -OMe, 3 OMe toward Ni	2028.6	0.19683	25.10

Table 4.2. NBO data for Selected Ni(CO)<sub>3</sub>L Complexes.

Instead of separate treatment of polarization and backbonding parameters with calculated CO stretching frequencies, a two-variable least-squares fit was performed in which both parameters were simultaneously fit to the data set comprised of calculated CO stretching frequencies (Figure 4.4). The coefficient of determination for the regression plane of best-fit (Figure 4.4,  $R^2 = 0.956$ ) suggests that the two-variable fit is better than either of the one-variable fits.



**Figure 4.4**. Two-variable linear regression fit of calculated CO stretching frequency to polarization and backbonding parameters for Ni(CO)<sub>3</sub>L Complexes.

 $v_{CO} = (-1001.0\pm80.8 \text{ cm}^{-1}) \text{ x} (\text{avg } \pi^* \text{ orbital occupancy on CO}) + (15.1\pm3.9 \text{ cm}^{-1}) \text{ x}$ (average percentage of the CO  $\pi^b$  orbitals lying on the C atom) + (1850.4±113.3 cm^{-1})

Because the polarization and the backbonding parameters refer to widely different properties, it was unclear from inspection of the regression equation what the relative contributions of polarization and backbonding were on the calculated CO stretching frequencies over the set of ligands used. To provide a sense of relative contributions, a standard deviation was calculated for both polarization and backbonding effects by first multiplying the sets of polarization and backbonding parameters by the regression coefficients for the respective parameters. The resulting two sets of values correspond to the frequency contribution of each effect for each ligand on the calculated CO stretching frequency. The standard deviation of each set was then calculated, which reflected the relative significance of polarization and backbonding effects on the perturbations to the CO stretching frequency. The standard deviation of backbonding effects was calculated to be 23.8 cm<sup>-1</sup>, while the standard deviation of polarization effects was calculated to be 7.5 cm<sup>-1</sup>, suggesting that backbonding effects are approximately 3 times more significant for the perturbations to CO stretching frequency over the set of calculated ligands.

Although both backbonding and polarization parameters independently correlate with CO stretching frequency, the perturbations to CO stretching frequency may not be dependent on both variables. If polarization correlates with backbonding, one or the other may be the underlying factor responsible for perturbations to CO stretching frequency. A plot of polarization vs. backbonding (Figure 4.5) does show a rough correlation between the two parameters. While there are a few points that lie off the bestfit line, two quadrants of the graph are clearly under-populated. Ligands that exhibit high backbonding and high polarization of the CO  $\pi^{b}$  orbital toward the C atom (upper right quadrant) and ligands that exhibit low backbonding and low polarization of the CO  $\pi^{b}$  orbital toward the C atom (lower left quadrant) are not represented or underrepresented. In contrast, ligands are generally aligned along the line connecting the quadrant representing high backbonding and low polarization of the CO  $\pi^{b}$  orbital toward the C atom (upper left) and its opposite quadrant (lower right). While this arises from the specific ligand set, it is unsurprising as ligands that .polarize the CO toward the C atom are usually relatively more electronegative, which decreases the propensity for that particular ligand-metal complex to exhibit backbonding. Future investigations to identify such ligands that lie in the under-represented regimes would further help to support the two-variable regression and more strongly establish independent correlation of the two parameters with CO stretching frequency.

**Figure 4.5.** Plot of average percentage of CO  $\pi^{b}$  orbital from the C atom vs. average CO  $\pi^{*}$  orbital occupancy for Ni(CO)<sub>3</sub>L Complexes.



Plot of Ligand Parameters (polarization vs. back-bonding)

#### 4.4 Perturbation of CO with a Point Charge

The stretching frequencies of free CO in the presence of a point charge were also computed. Assuming that the geometry of the M-(CO)<sub>3</sub> fragment is relatively constant across all complexes, through-space perturbations to  $v_{CO}$  can be roughly approximated as the effect of a point charge at a fixed distance away from the CO; different overall perturbations are approximated as different point charges. Several series of data were generated for different metal-point charge distances by varying the magnitude of the charges at a particular distance (Figure 4.6). Moving left to right in Figure 4.6 corresponds to a greater polarization of the  $\pi^{b}$  orbitals toward the carbon atom, and variation of  $v_{CO}$  with the change in polarization clearly shows that stretching frequency is perturbed by the presence of a point charge, which is consistent with previous work from our group.<sup>6</sup> While the plot is roughly linear across the range of polarization for Tolman's ligands (25.13% to 25.82%), deviations from linearity are observed at high polarization away from free CO. The plots change in the curvature as the distance of the point charge from the CO changes (i.e. d(q-C) = 2.2 Å vs. d(q-C) = 1000 Å). This result is not unexpected, since the carbon and oxygen atom lie at different distances from the point charge and thus the relative electrostatic field felt by them scales differently as well.

**Figure 4.6.** Plot of calculated  $v_{CO}$  vs. the average percentage of the CO  $\pi^{b}$  orbitals lying on the carbon atom for free CO in the presence of a point charge.



### 4.5 Summary

Calculations were performed to investigate the contributions of through-space effects on CO stretching frequency. We found that a linear two-variable model fit the data better than the traditional one-variable model of solely back-bonding effects. A close look at the data gave a few examples that highlighted the dominant effect of polarization in some cases. In the course of our investigation, we also calculated the effects of a point charge on free CO, which was consistent with previous work.

#### References

- Collman, J. P.; Hegedus, L. S.; Norton, J. R.; Finke, R. G. *Principles and Applications of Organotransition Metal Chemistry*; University Science Books: Mill Valley, CA, 1987.
- (2) Tolman, C. A. Chem. Rev. 1977, 77, 313.
- (3) Perrin, L.; Clot, E.; Eisenstein, O.; Loch, J.; Crabtree, R. H. *Inorg. Chem.* **2001**, *40*, 5806.
- (4) Cooney, K. D.; Cundari, T. R.; Hoffman, N. W.; Pittard, K. A.; Temple, M. D.; Zhao, Y. J. Am. Chem. Soc. 2003, 125, 4318.
- (5) Dorta, R.; Stevens, E. D.; Scott, N. M.; Costabile, C.; Cavallo, L.; Hoff, C. D.; Nolan, S. P. J. Am. Chem. Soc. 2005, 127, 2485.
- (6) Goldman, A. S.; Krogh-Jespersen, K. J. Am. Chem. Soc. 1996, 118, 12159.
- (7) Phillips, G. N., Jr.; Teodoro, M. L.; Li, T.; Smith, B.; Olson, J. S. J. Phys. Chem. B 1999, 103, 8817.
- (8) Park, E. S.; Andrews, S. S.; Hu, R. B.; Boxer, S. G. J. Phys. Chem. B 1999, 103, 9813.
- (9) Lamberti, C.; Bordiga, S.; Geobaldo, F.; Zecchina, A.; Otero Arean, C. J. Chem. Phys. **1995**, 103, 3158.
- (10) Kushkuley, B.; Stavrov, S. S. Biophys. J. 1996, 70, 1214.
- (11) Augspurger, J. D.; Dykstra, C. E.; Oldfield, E. J. Am. Chem. Soc. 1991, 113, 2447.
- (12) Laberge, M.; Vanderkooi, J. M.; Sharp, K. A. J. Phys. Chem. 1996, 100, 10793.
- (13) Zhu, K.; Achord, P. D.; Zhang, X.; Krogh-Jespersen, K.; Goldman, A. S. J. Am. Chem. Soc. 2004, 126, 13044.
- (14) Parr, R. G.; Yang, W. *Density-functional Theory of Atoms and Molecules*; Oxford University Press: Oxford, U.K., 1989.
- Gaussian 03, Revision C.02, Frisch, M. J. T., G. W.; Schlegel, H. B.; Scuseria, G. (15)E.; Robb, M. A.; Cheeseman, J. R.; Montgomery, Jr., J. A.; Vreven, T.; Kudin, K. N.; Burant, J. C.; Millam, J. M.; Iyengar, S. S.; Tomasi, J.; Barone, V.; Mennucci, B.; Cossi, M.; Scalmani, G.; Rega, N.; Petersson, G. A.; Nakatsuji, H.; Hada, M.; Ehara, M.; Toyota, K.; Fukuda, R.; Hasegawa, J.; Ishida, M.; Nakajima, T.; Honda, Y.; Kitao, O.; Nakai, H.; Klene, M.; Li, X.; Knox, J. E.; Hratchian, H. P.; Cross, J. B.; Bakken, V.; Adamo, C.; Jaramillo, J.; Gomperts, R.; Stratmann, R. E.; Yazyev, O.; Austin, A. J.; Cammi, R.; Pomelli, C.; Ochterski, J. W.; Ayala, P. Y.; Morokuma, K.; Voth, G. A.; Salvador, P.; Dannenberg, J. J.; Zakrzewski, V. G.; Dapprich, S.; Daniels, A. D.; Strain, M. C.; Farkas, O.; Malick, D. K.; Rabuck, A. D.; Raghavachari, K.; Foresman, J. B.; Ortiz, J. V.; Cui, Q.; Baboul, A. G.; Clifford, S.; Cioslowski, J.; Stefanov, B. B.; Liu, G.; Liashenko, A.; Piskorz, P.; Komaromi, I.; Martin, R. L.; Fox, D. J.; Keith, T.; Al-Laham, M. A.; Peng, C. Y.; Nanayakkara, A.; Challacombe, M.; Gill, P. M. W.; Johnson, B.; Chen, W.; Wong, M. W.; Gonzalez, C.; and Pople, J. A., Gaussian, Inc., Wallingford, CT.
- (16) Perdew, J. P.; Burke, K.; Ernzerhof, M. Phys. Rev. Lett. 1996, 77, 3865.
- (17) Dolg, M.; Wedig, U.; Stoll, H.; Preuss, H. J. Chem. Phys. 1987, 86, 866.
- (18) Ditchfield, R.; Hehre, W. J.; Pople, J. A. J. Chem. Phys. 1971, 54, 724.

- (19) Hariharan, P. C.; Pople, J. A. Mol. Phys. 1974, 27, 209.
- (20) Krishnan, R.; Binkley, J. S.; Seeger, R.; Pople, J. A. J. Chem. Phys. 1980, 72, 650.
- (21) Clark, T.; Chandrasekhar, J.; Spitznagel, G. W.; Schleyer, P. v. R. J. Comput. Chem. 1983, 4, 294.
- (22) Reed, A. E.; Curtiss, L. A.; Weinhold, F. Chem. Rev. 1988, 88, 899.
- (23) Carpenter, J. E.; Weinhold, F. Theochem 1988, 46, 41.

## Chapter 5

# Kinetic Study of (<sup>tBu</sup>PCP)Ir and (<sup>tBu</sup>POCOP)Ir Complexes: Olefin Coordination and Isomerization

#### Abstract

Olefin isomerization is particularly relevant to the selectivity of alkane dehydrogenation and alkane metathesis reactions. In our investigations to better understand the mechanism of olefin isomerization, the reaction kinetics of olefin coordination to (<sup>tBu</sup>PCP)Ir and (<sup>tBu</sup>POCOP)Ir were quantified and discussed in the context of olefin isomerization. Additionally, the (<sup>tBu</sup>POCOP)Ir-catalyzed olefin isomerization of 1-octene was investigated and found to show an apparent 2<sup>nd</sup> order dependence on 1-octene concentration.

#### **5.1 Introduction**

Examples of transition metal-catalyzed olefin isomerization are widespread and play a key role in numerous important chemical processes.<sup>1-9</sup> The mechanism most commonly accepted for olefin isomerization involves initial addition of a metal-H bond across the olefin double bond. If the resulting metal alkyl has a  $\beta$ -carbon atom that is inequivalent to the carbon to which the hydride was added,  $\beta$ -elimination of hydrogen from that carbon will result in olefin isomerization (Scheme 5.1a). An alternative pathway, less commonly proposed and less well studied, involves the intermediacy of a  $\pi$ -allyl complex (Scheme 5.1b) through which olefin isomerization occurs via a formal 1,3-hydride shift.<sup>1,2,10-14</sup> If a coordinatively unsaturated metal complex bears a metalbound hydride, the hydride addition pathway is believed to be the preferred mechanism for olefin isomerization.<sup>1</sup>

Scheme 5.1. Schematic illustration of the two mechanism classes proposed for olefin isomerization: (a) "hydride" mechanism, (b)  $\pi$ -allyl mechanism.



Olefin isomerization is particularly relevant to the selectivity of alkane dehydrogenation and alkane metathesis reactions. Iridium pincer catalysts, including  $(^{R}PCP)Ir (^{R}PCP = \kappa^{3} - C_{6}H_{3} - 2,6 - (CH_{2}PR_{2})_{2})$  and  $(^{R}POCOP)Ir (^{R}POCOP = \kappa^{3} - C_{6}H_{3} - 2,6 - (CH_{2}PR_{2})_{2})$  $(OPR_2)_2$ ) have been found to effect alkane dehydrogenation. Of these, (<sup>tBu</sup>PCP)Ir and (<sup>iPr</sup>PCP)Ir complexes have been found to dehydrogenate *n*-alkanes with kinetic selectivity to form  $\alpha$ -olefins.<sup>15</sup> These alkane dehydrogenation catalysts, when coupled with a Schrock-type olefin metathesis catalyst, can vield alkane metathesis (Scheme 5.2). The alkane metathesis system might be expected to selectively produce ethane and  $C_{2n-2}$ species as shown in the example in Scheme 2 (metathesis of *n*-hexane to yield ethane and *n*-decane); in fact, AM of *n*-hexane generates a range of  $C_2$  to  $C_{15}$  *n*-alkanes.<sup>16,17</sup> Significantly, the (<sup>tBu</sup>PCP)Ir system gives moderate selectivity for the formation of *n*decane (ca. 50 mol% of the  $C_n$  products where n>6), whereas the (<sup>tBu</sup>POCOP)Ir system gives an essentially stochastic distribution of n-alkanes.<sup>16,17</sup> The most obvious explanation for the initial formation of  $C_{3.5}$  and  $C_{7.9}$  products from *n*-hexane is based upon isomerization/metathesis of the olefin intermediates prior to hydrogenation. A significant degree of isomerization of 1-alkenes has indeed been previously reported when pincer-Ir complexes are used for either acceptorless or transfer-dehydrogenation of *n*-alkanes.<sup>18</sup> In consideration of these issues, we initiated a study of pincer-iridium catalyzed olefin isomerization with an emphasis on the comparison between the (<sup>tBu</sup>PCP)Ir and (<sup>tBu</sup>POCOP)Ir systems.<sup>19</sup>

**Scheme 5.2.** Tandem catalytic alkane metathesis with pincer-ligated iridium complexes and olefin metathesis catalysts (shown for reaction of *n*-hexane)



A kinetic study of the isomerization of 1-octene catalyzed by (<sup>HBu</sup>PCP)IrH<sub>2</sub> (1) was previously conducted by Soumik Biswas.<sup>19</sup> Addition of 1-octene (100 mM) to 1 (5 mM) in either n-octane or p-xylene results in apparently quantitative conversion to (<sup>HBu</sup>PCP)Ir(1-octene) within 10 min at 25 °C. The turnover frequency (k) for eq 1 in either solvent is 4.2 x 10<sup>-3</sup> s<sup>-1</sup> at 125 °C (rate = k[(<sup>HBu</sup>PCP)Ir],  $\Delta G^{\neq} \sim 27.9$  kcal/mol), suggesting that a "hydride" mechanism for olefin isomerization is not operative.<sup>20</sup> A zero-order dependence on [1-octene] is observed (Figure 5.1), which is consistent with an intramolecular  $\pi$ -allyl mechanism in which (<sup>HBu</sup>PCP)Ir(1-octene) is the major resting state (as observed by NMR spectroscopy), and the rate-determining step lies within segments (a), (b), or (c) of the cycle indicated in Scheme 5.3. (We define these as "segments" of the cycle since each of them, and particularly (a) or (b), could involve more than one elementary reaction step).


Figure 5.1. Catalytic isomerization of 1-octene to internal octenes by 1 (5 mM) at 125 °C

in *n*-octane solvent.



Scheme 5.3. Schematic  $\pi$  -allyl pathway for 1-alkene isomerization with dissociative olefin exchange (detailed mechanism not specified).



These kinetic experiments were repeated with ( $^{IBu}POCOP$ )IrH<sub>2</sub> (**2**) as the catalyst precursor. As was the case with ( $^{IBu}PCP$ )IrH<sub>2</sub>, ( $^{IBu}POCOP$ )Ir(1-octene) was formed rapidly and apparently quantitatively, and remained the major species in solution during the catalysis, as determined by *in situ* <sup>31</sup>P NMR spectroscopy. While no significant

difference in isomerization rates was found using *n*-octane as solvent vs. *p*-xylene, the reaction kinetics showed a clear positive dependence on 1-octene concentration. This result is discussed below in greater detail.

In this chapter, I will discuss my contributions to the study published as reference 20, with regards to quantifying the kinetics of olefin coordination to (<sup>tBu</sup>PCP)Ir and (<sup>tBu</sup>POCOP)Ir as well as discuss olefin isomerization by (<sup>tBu</sup>POCOP)Ir.

## 5.2 Results and Discussion

## 5.2.1 Kinetics of Olefin Dissociation from (pincer) Ir(alkene) Complexes

Consistent with Scheme 5.3, loss of alkene from (<sup>IBu</sup>PCP)Ir(alkene) is rapid and dissociative. Exchange spectroscopy (EXSY) experiments reveal that at 25 °C the rate of 1-hexene dissociation is 0.51 s<sup>-1</sup> and independent of the concentration of added 1-hexene. An Eyring plot based on rates measured at temperatures between 20 °C and 46 °C affords activation parameters  $\Delta H^{\neq} = 21.4 \pm 0.8$  kcal/mol and  $\Delta S^{\neq} = 12 \pm 3$  eu for loss of 1-hexene (Figure 5.2). Dissociation of *trans*-2-hexene, as would be expected of a dissociative process with a bulkier ligand, is even faster; at 4 °C the rate is 3.1 s<sup>-1</sup>, independent of the concentration of added *trans*-2-hexene, with activation parameters determined to be  $\Delta H^{\neq}$ = 15.4 ± 0.4 kcal/mol and  $\Delta S^{\neq} = -0.7 \pm 1.5$  eu (from rates measured at temperatures ranging from -17 °C to 6 °C, Figure 5.3). Extrapolation to 125 °C yields a value (2 × 10<sup>4</sup> s<sup>-1</sup>) that is orders of magnitude greater than the overall rate of isomerization. Thus, the rate determining step of the cycle indicated in Scheme 5.3 lies within segments (a) or (b).



**Figure 5.2.** Eyring plot for the rate of dissociative 1-hexene exchange of  $(^{tBu}PCP)Ir(1-hexene)$  with free 1-hexene in *p*-xylene- $d_{10}$  in the temperature interval 20 °C to 46 °C.

**Figure 5.3.** Eyring plot for the rate of dissociative *trans*-2-hexene exchange of  $({}^{tBu}PCP)Ir(trans$ -2-hexene) with free *trans*-2-hexene in mesitylene- $d_{12}$  in the temperature interval -17 °C to 6 °C.



Reaction kinetics for self-exchange for (<sup>Bu</sup>POCOP)Ir(olefin) were also determined by EXSY spectroscopy. Similar to loss of alkene from (<sup>tBu</sup>PCP)Ir(alkene), loss of alkene from (<sup>tBu</sup>POCOP)Ir(alkene) is rapid and dissociative. From Eyring plots shown in Figures 5.4 and 5.5, the activation enthalpy for (<sup>tBu</sup>POCOP)Ir(olefin) was found to be 29.4  $\pm$  1.2 and 26.0  $\pm$  1.2 kcal/mol for 1-hexene and *trans*-2-hexene, respectively, while the respective activation entropies were found to be 15  $\pm$  3 and 17  $\pm$  3 eu (from rates measured at temperatures between 107 °C and 126 °C, and 57 °C to 79 °C, respectively). These olefin dissociation rates for (<sup>tBu</sup>POCOP)Ir(olefin) are in fact much slower than those of the corresponding (<sup>tBu</sup>PCP)Ir complexes, as would be expected from the decrease in steric demand of the arrangement of the tBu substituents in the <sup>tBu</sup>POCOP ligand relative to the <sup>tBu</sup>PCP ligand. This decrease in unfavorable steric repulsions for (<sup>tBu</sup>POCOP)Ir results in an increase in olefin binding affinity. Nevertheless, extrapolation to 125 °C for *trans*-2-hexene self-exchange (experimentally determined at increments from 57 °C to 79 °C) affords a rate of 2.1 x  $10^2$  s<sup>-1</sup>, which is still much faster than the reported rate of isomerization of (<sup>tBu</sup>POCOP)Ir(*trans*-2-hexene) to (<sup>tBu</sup>POCOP)Ir(1-hexene),<sup>20</sup> suggesting that olefin exchange is a fast step in (<sup>tBu</sup>POCOP)Ir-catalyzed olefin isomerization.

**Figure 5.4.** Eyring plot for the rate of dissociative 1-hexene exchange of  $({}^{tBu}POCOP)Ir(1-hexene)$  with free 1-hexene in *p*-xylene- $d_{10}$  in the temperature interval 107 °C to 126 °C.



**Figure 5.5.** Eyring plot for the rate of dissociative *trans*-2-hexene exchange of (<sup>tBu</sup>POCOP)Ir(*trans*-2-hexene) with free *trans*-2-hexene in *p*-xylene- $d_{10}$  in the temperature interval 57 °C to 79 °C.



# 5.2.2 (<sup>tBu</sup>POCOP)Ir-Catalyzed Olefin Isomerization of 1-Octene

The ( $^{IBu}POCOP$ )Ir-catalyzed olefin isomerization of 1-octene to internal octenes was investigated analogously to ( $^{IBu}PCP$ )Ir. In contrast with the ( $^{IBu}PCP$ )Ir-catalyzed reaction, the reaction kinetics of ( $^{IBu}POCOP$ )Ir-catalyzed olefin isomerization show a clear positive dependence on 1-octene concentration (Figure 5.6). Given that the 1octene complex is observed to be the major species in solution, this is not easily reconciled with a hydride-insertion mechanism, since the steady-state concentration of ( $^{IBu}POCOP$ )IrH<sub>2</sub> should be *inverse-second-order* in [1-octene] (and first-order in *n*octane) (eq 2). This positive dependence on [1-octene] is, however, also inconsistent with a  $\pi$ -allyl mechanism operating with ( $^{IBu}POCOP$ )Ir(1-octene) as the resting state, unless the rate-determining step involves an additional equivalent of 1-octene. While this would be the case if loss of 2-alkene (step c in Scheme 5.3) were reversible and the backreaction with 2-octene were fast relative to addition of 1-alkene, these conditions seem highly unlikely given the lesser bulkiness and initially higher concentration of 1-alkenes.

**Figure 5.6.** Catalytic isomerization of 1-octene to internal octenes by (<sup>tBu</sup>POCOP)IrH<sub>2</sub> (5 mM) at 125 °C.



 $[(^{tBu}POCOP)IrH_2] = K_{eq-2} \bullet [n \text{-}octane] \bullet [(^{tBu}POCOP)Ir(1 \text{-}octene)] / [1 \text{-}octene]^2$ 

An alternative explanation for the positive dependence of rate on [1-alkene] is that formation of (<sup>tBu</sup>POCOP)Ir(2-alkene) is reversible, and that displacement of coordinated 2-alkene by 1-alkene proceeds through an *associative* rate-determining step; such a pathway is indicated in Scheme 5.4. However, as noted above, the rate of *trans*-2-hexene self exchange is much faster than isomerization of (<sup>tBu</sup>POCOP)Ir(trans-2-alkene) to (<sup>tBu</sup>POCOP)Ir(1-alkene), which argues against such a mechanism.

**Scheme 5.4.** Alternative  $\pi$ -allyl pathway for 1-alkene isomerization, involving associative displacement of 2-alkene by 1-alkene.



Closer examination of the reaction kinetics appears to show that the reaction is 2<sup>nd</sup> order in [1-octene] (Figure 5.7). Given that the resting state of the catalyst is the (<sup>tBu</sup>POCOP)Ir(1-octene) complex, this suggests that *two* additional equivalents of 1-octene bind to the complex in the rate determining step. Speculatively, one could envision a rate determining transition state in which the phosphinite arms of the pincer ligand decoordinate and are replaced by olefin ligands, resulting in the active olefin isomerization catalyst. However, we currently have no data to support or refute such a hypothesis.



Figure 5.7.  $2^{nd}$  order plot of the catalytic isomerization of 1-octene to internal octenes by 2 (5 mM) at 125 °C.

## 5.3 Summary

The reaction kinetics of olefin coordination to (<sup>tBu</sup>PCP)Ir and (<sup>tBu</sup>POCOP)Ir were quantified and were found to be rapid relative to olefin isomerization. The kinetics of olefin exchange was inversely related to the steric bulk of the olefin and pincer ligand; less bulky olefins and ligands resulted in slower exchange rates. Accordingly, 1-hexene underwent olefin exchange more slowly than *trans*-2-hexene, and (<sup>tBu</sup>POCOP)Ir-bound olefins underwent olefin exchange more slowly than (<sup>tBu</sup>PCP)Ir-bound olefins. The (<sup>tBu</sup>POCOP)Ir-catalyzed olefin isomerization of 1-octene was found to show an apparent 2<sup>nd</sup> order dependence on 1-octene concentration.

#### 5.4 Experimental

## 5.4.1 General Considerations.

All manipulations were carried out using standard Schlenk, high-vacuum and glovebox techniques. Anhydrous p-xylene- $d_{10}$  was dried over Na/K and collected by vacuum transfer. 1-hexene and trans-2-hexene were degassed and used without further purification. Complexes  $1^{21}$  and  $2^{22}$  were synthesized as previously reported.

## 5.4.2 EXSY Experiments.

10.0 mg of a pincer-iridium dihydride complex (17 µmol) and 10.0 µL of olefin were dissolved in 0.7 mL of *p*-xylene- $d_{10}$ . Conversion to the pincer-iridium olefin complex was complete after 30 min at 25 °C for most pincer-ligand-olefin combinations. Formation of the (<sup>tBu</sup>PCP)Ir(*trans*-2-hexene) complex required heating at 50 °C for 30 min to effect complete conversion to the olefin complex, and the rapid rate of olefin exchange for this complex necessitated a change in solvent to mesitylene- $d_{12}$ . After formation of the olefin complex, a vinyl proton of the bound olefin complex was selectively irradiated in a 1D EXSY experiment, and magnetization transfer to free olefin was observed at different mixing times. The rate of olefin dissociation was determined from a plot of magnetization transfer vs. mixing time.<sup>23</sup> NMR spectroscopy acquisition temperatures were determined by measuring an external ethylene glycol or methanol reference sample. These 1D EXSY experiments were performed at different temperatures, and an Eyring plot yielded activation energy parameters  $\Delta H^{\ddagger}$  and  $\Delta S^{\ddagger}$ . Olefin exchange was confirmed to proceed via a dissociative mechanism by repeating the EXSY experiment with a much larger excess of free olefin (4 to 10-fold excess) and observing negligible change in the rate.

## 5.4.3 Catalytic 1-Octene Isomerization.

3.0 mg of a pincer-iridium dihydride complex (5.1 mM, ( $^{tBu}PCP$ )IrH<sub>2</sub> or ( $^{tBu}POCOP$ )IrH<sub>2</sub>) was dissolved with 1-octene (17.0 µL, 108 mM) and mesitylene (5.0 µL, 36 mM, internal standard) in a volume of *n*-octane or *p*-xylene solvent such that the total volume of the reaction solution was 1 mL. The reaction solution was transferred into an air-tight, septa-sealed vial. An initial aliquot was taken before heating the reaction to 125 °C. Reaction data was obtained via GC analysis of aliquots taken at various times during the reaction.

## References

- Herrmann, W. A.; Prinz, M. In *Applied Homogeneous Catalysis with* Organometallic Compounds (2nd Edition); Cornils, B., Herrmann, W. A., Eds.; Wiley-VCH Verlag GmbH: Weinheim, Germany, 2002; 3, pp 1119.
- (2) Crabtree, R. H. In *The Organometallic Chemistry of the Transition Metals*; 5th ed.; John Wiley & Sons: Hoboken, NJ, 2009, pp 229.
- (3) Mol, J. C. J. Mol. Catal. A Chem. 2004, 213, 39.
- (4) van Leeuwen, P. W. N. M. In *Homogeneous Catalysis: Understanding the Art*; Kluwer Academic Publishers: Dordrecht, 2004, pp 101.
- (5) Schmidt, B. Eur. J. Org. Chem. 2004, 1865.
- (6) Otsuka, S.; Tani, K. Transition Met. Org. Synth. (2nd Ed.) 2004, 1, 199.
- (7) Seayad, A.; Ahmed, M.; Klein, H.; Jackstell, R.; Gross, T.; Beller, M. *Science* **2002**, *297*, 1676.
- (8) Scarso, A.; Colladon, M.; Sgarbossa, P.; Santo, C.; Michelin, R. A.; Strukul, G. *Organometallics* **2010**, *29*, 1487.
- (9) Casey, C. P.; Cyr, C. R. J. Am. Chem. Soc. 1973, 95, 2240.
- (10) Manuel, T. A. J. Org. Chem. 1962, 27, 3941.
- (11) Alper, H.; LePort, P. C.; Wolfe, S. J. Am. Chem. Soc. 1969, 91, 7553.
- (12) Cowherd, F. G.; Von Rosenberg, J. L. J. Am. Chem. Soc. 1969, 91, 2157.
- (13) Misono, M.; Grabowski, W.; Yoneda, Y. J. Catal. 1977, 49, 363.
- (14) Casey, C. P.; Cyr, C. R. J. Am. Chem. Soc. 1973, 95, 2248.
- (15) Liu, F.; Pak, E. B.; Singh, B.; Jensen, C. M.; Goldman, A. S. J. Am. Chem. Soc. 1999, 121, 4086.
- (16) Goldman, A. S.; Roy, A. H.; Huang, Z.; Ahuja, R.; Schinski, W.; Brookhart, M. *Science* **2006**, *312*, 257.
- (17) Bailey, B. C.; Schrock, R. R.; Kundu, S.; Goldman, A. S.; Huang, Z.; Brookhart, M. Organometallics **2009**, *28*, 355.
- (18) Choi, J.; MacArthur, A. H. R.; Brookhart, M.; Goldman, A. S. Chem. Rev. 2011, 111, 1761.
- (19) Biswas, S. Regioselectivity in Catalytic Transfer Dehydrogenation and Mechanism of 1-Alkene Isomerization. Ph. D. Dissertation, Rutgers, The State University of New Jersey, 2010.
- (20) Biswas, S.; Huang, Z.; Choliy, Y.; Wang, D. Y.; Brookhart, M.; Krogh-Jespersen, K.; Goldman, A. S. J. Am. Chem. Soc. **2012**, 134, 13276.
- (21) Gupta, M.; Hagen, C.; Flesher, R. J.; Kaska, W. C.; Jensen, C. M. Chem. Commun. 1996, 2083.
- (22) Göttker-Schnetmann, I.; White, P. S.; Brookhart, M. Organometallics 2004, 23, 1766.
- (23) Perrin, C. L.; Dwyer, T. J. Chem. Rev. 1990, 90, 935.

# Chapter 6

# Investigations of the (<sup>tBu3Me</sup>PCP)Ir(CO) Complex: Synthesis and Oxidative Addition of H<sub>2</sub>

## Abstract

A revised synthesis scheme for the synthesis of the <sup>tBu3Me</sup>PCP ligand was developed. The revised synthesis scheme was found to be amenable for the synthesis of a variety of unsymmetrical pincer ligands. With the <sup>tBu3Me</sup>PCP ligand, addition of H<sub>2</sub> to <sup>tBu3Me</sup>PCPIr(CO) and <sup>tBu4</sup>PCPIr(CO) was investigated. Addition of H<sub>2</sub> was not observed to add to <sup>tBu4</sup>PCPIr(CO). However, when H<sub>2</sub> was added to <sup>tBu3Me</sup>PCPIr(CO), three isomers were observed and assigned using an isotopic labeling experiment and NMR spectroscopy. The kinetic products were the two possible isomers of *cis*-H<sub>2</sub> addition. The thermodynamic product was the result of *trans*-H<sub>2</sub> addition.

## **6.1 Introduction**

Oxidative addition of small molecules has been a longstanding fundamental transformation of organometallic chemistry. Both steric and electronic factors have been investigated regarding the favorability of small molecule addition.<sup>1-8</sup> The simplest small molecule to undergo oxidative addition, H<sub>2</sub>, has been reported from the beginnings of organometallic chemistry.<sup>9,10</sup> While the effects of sterics on the reactivity and selectivity of oxidative addition of larger molecules to metal centers has been explored, investigations into the oxidative addition of H<sub>2</sub> have generally focused on electronic effects, given the small size of H<sub>2</sub>.<sup>11-13</sup> There are comparatively fewer investigations of steric effects on the oxidative addition reaction of H<sub>2</sub>.<sup>14,15</sup> Selective catalytic reactions that involve the oxidative addition of hydrogen such as asymmetric hydrogenation have been achieved, but most strategies utilize bulky ligand frameworks to bias the coordination and reactivity of the much larger substrate being hydrogenated rather than control the addition of H<sub>2</sub>.<sup>13</sup> As such, steric considerations for the oxidative addition reaction of H<sub>2</sub> tend to be scant.

Our group has reported the synthesis of several PCP pincer Ir complexes, which are identical save for the substituents on the phosphine atoms.<sup>16</sup> Our previous studies have examined the reactivity of these complexes for catalytic transfer dehydrogenation, and have found that a very large difference in reactivity is observed between tBu3MePCPIrH<sub>2</sub> and tBu4PCPIrH<sub>2</sub>. It appears that replacing one sterically large tBu group for a methyl group results in drastically improved activity.

In this chapter, I detail a very large change in reactivity for the oxidative addition of H<sub>2</sub> when changing the ligand from  ${}^{tBu4}PCPIrH_2$  to  ${}^{tBu3Me}PCPIrH_2$ . In addition, the synthesis route for <sup>tBu3Me</sup>PCPIrH<sub>2</sub> was revised; the new route was found to be useful for the generation of a variety of unsymmetrical PCP pincer ligands.

## 6.2 Revised Synthesis of <sup>tBu3Me</sup>PCPIrH<sub>2</sub>

<sup>tBu3Me</sup>PCPIr(CO) is obtained from <sup>tBu3Me</sup>PCPIrH<sub>2</sub> by reaction with CO. However, the previously reported synthesis route of <sup>tBu3Me</sup>PCPIrH<sub>2</sub> by our group is somewhat inelegant.<sup>16</sup> The reported route uses *m*-xylylene dibromide as the starting point and requires desymmetrization of the two benzyl bromide arms as the phosphine groups are introduced successively. The desymmetrization and initial mono-phosphonation of *m*xylylene dibromide was achieved by utilizing a very large excess of the dibromide (5 fold excess of dibromide vs. phosphine). Separation of the desired mono-phosphine product from the excess starting material required multiple cycles of reprecipitation by the addition of diethyl ether to an acetone solution of the reaction mixture. Addition of the second phosphine group, completion of the ligand, and metallation all proceeded analogously to the well-established procedure for <sup>tBu4</sup>PCPIrHCl synthesis.

Similarly, Jensen *et al.* reported a synthesis scheme to generate unsymmetrical diphosphine pincer ligands by initial desymmetrization of the resourcinol starting material to (2-chloromethyl)benzyl alcohol.<sup>17</sup> Initial phosphonation was achieved by reaction of the benzylic chloride with diphenylphosphine. Completion of the ligand was achieved following BH<sub>3</sub>-protection of the phosphine, conversion of the remaining benzylic alcohol to a chloride, addition of the second phosphine group, and deprotection of the phosphine groups. Although Jensen utilized this synthesis scheme to generate PCP

pincer ligands with a diphenylphosphine moiety, this scheme can be easily adapted to the synthesis of unsymmetrical alkyl-phosphine pincer ligands such as the <sup>tBu3Me</sup>PCP ligand.

In contrast to the previous synthesis of the <sup>tBu3Me</sup>PCP ligand and Jensen's unsymmetrical diphosphine PCP ligand synthesis, the revised synthesis scheme (Scheme 6.1) utilizes methyl (3-bromomethyl)benzoate as the starting point, which obviates the need for desymmetrization. Synthesis of the <sup>tBu3Me</sup>PCP ligand began with the phosphonation of methyl (3-bromomethyl)benzoate with Li<sup>1</sup>Bu<sub>2</sub>P·BH<sub>3</sub> to generate compound **2**. The BH<sub>3</sub>-protected phosphine is easily synthesized and has the advantage of being stable under ambient conditions.<sup>18,19</sup> The ester moiety of compound **2** was reduced with diisobutylaluminum hydride to yield benzylic alcohol **3** and then converted to benzylic bromide **4** via reaction with PBr<sub>3</sub>. The second phosphonation reaction was achieved using another lithiated BH<sub>3</sub>-protected phosphine, Li<sup>1</sup>BuMeP·BH<sub>3</sub>, to yield BH<sub>3</sub>protected ligand **5**. Deprotection of the phosphines under conditions reported by Jensen afforded the desired <sup>tBu3Me</sup>PCP pincer ligand **6**.<sup>17</sup> Metallation of the ligand onto Ir to yield hydrido chloride complex **7** and subsequent reduction to dihydride complex **8** followed previously reported procedures for <sup>(Bu3Me</sup>PCPIrH<sub>2</sub>.



Scheme 6.1. Revised Synthesis Scheme for <sup>tBu3Me</sup>PCPIrH<sub>2</sub> (8).

While the original aim of this revised synthesis scheme was to obtain the <sup>tBu3Me</sup>PCP ligand, this scheme can be easily modified to generate a large variety of unsymmetrical PCP-type pincer ligands through the installation of different phosphine groups during the second phosphination step. In addition, this scheme is scalable; compound **4** was easily synthesized in multigram quantities and can serve as a precursor for various unsymmetrical or chiral pincer ligands. While our group has not yet undertaken an extensive investigation of the ligand variants, this synthesis scheme was successfully utilized by our group in the synthesis of <sup>tBu2</sup>PCP<sup>iPr2</sup>Ir complexes from compound **4**.<sup>20</sup>

## 6.3 Addition of H<sub>2</sub> to <sup>tBu3Me</sup>PCPIrCO

With the <sup>tBu4</sup>PCPIr(CO) and <sup>tBu3Me</sup>PCPIr(CO) complexes in hand, we turned to investigating the addition of H<sub>2</sub> to these complexes. (Milstein *et al.* reported the analogous addition of H<sub>2</sub> to <sup>iPr4</sup>PCPIrCO.<sup>21</sup>) Addition of H<sub>2</sub> to <sup>tBu4</sup>PCPIrCO does not show any reaction after several weeks at elevated temperature (40 °C). By contrast, addition of H<sub>2</sub> to <sup>IBu3Me</sup>PCPIrCO at room temperature immediately displays signals in the hydride region of the <sup>1</sup>H NMR spectrum, corresponding to 6-coordinate dihydride complexes. The three isomers of H<sub>2</sub> addition are possible and are shown in Figure 6.1: *cis* addition of H<sub>2</sub> to the bis-<sup>1</sup>Bu side, *cis* addition of H<sub>2</sub> to the <sup>1</sup>BuMe side, and *trans* addition of H<sub>2</sub>. After 8 h at room temperature, six hydride signals were observed in the <sup>1</sup>H NMR spectrum corresponding to the three isomers. The assignment of the observed isomers was achieved through the use of <sup>13</sup>C-labeled CO in the formation of t<sup>IBu3Me</sup>PCPIr(<sup>13</sup>CO). Two hydride signals display large <sup>1</sup>H-<sup>13</sup>C coupling constants (J<sub>HC</sub> = 38 Hz and 39.5 Hz) for one of the hydride signals, consistent with a relative *trans* arrangement between the hydride and CO group (Figure 6.2). The remaining hydride signals could be assigned to the *trans* addition isomer, while the remaining hydride signals could be assigned to the *trans* addition isomer, while the remaining hydride signals correspond to *cis* addition of H<sub>2</sub> to t<sup>IBu3Me</sup>PCPIr(CO).

**Figure 6.1**. Isomers of <sup>tBu3Me</sup>PCPIr(CO)(H)<sub>2</sub> complexes.







*cis* addition to bis-<sup>t</sup>Bu side



trans addition



**Figure 6.2.** <sup>1</sup>H NMR spectrum of H<sub>2</sub> addition to  $^{\text{tBu3Me}}\text{PCPIr(CO)}$  (top, green) and  $^{\text{tBu3Me}}\text{PCPIr}(^{13}\text{CO})$  (bottom, red).

Low temperature experiments clearly showed that the *cis* isomers are the kinetic products of H<sub>2</sub> addition. After 3 h at 0 °C, only the *cis* isomers were observed. The assignment of the *cis* isomers was performed using 1D NOESY experiments for the hydride signals *trans* to the CO group (Figure 6.3). Selective excitation of the hydride H<sub>a</sub> at  $\delta = -10.6$  ppm resulted in a NOE correlation with two <sup>t</sup>Bu groups as well as hydride H<sub>d</sub>. By contrast, hydride H<sub>b</sub> at  $\delta = -10.9$  ppm has a NOE correlation signal with only one <sup>t</sup>Bu group. Thus the kinetically preferred isomer was determined to be *cis* addition of the H<sub>2</sub> to the bis-<sup>t</sup>Bu side.



**Figure 6.3.** 1D NOESY spectra of  ${}^{tBu3Me}PCPIr(CO)$  with the selective excitation of hydride (**a**) H<sub>a</sub> and (**b**) H<sub>b</sub>.

The integration ratio of the hydride signals for the *cis* isomers was found to be 1.9 at low temperature (<-30 °C) at early reaction times, and the same ratio is observed as the reaction progresses and as the reaction is allowed to warm to room temperature. This suggests that the kinetic selectivity and thermodynamic selectivity for the two *cis* isomers are coincidentally the same ( $\Delta\Delta G^{\ddagger} = \Delta\Delta G$ ).

The *cis* isomers convert completely to the *trans* isomer at room temperature after 17 d. Upon heating to 120 °C in vacuo overnight, the *trans* isomer does undergo partial conversion back to the 4-coordinate complex (22 % conversion by integration of the  $^{31}$ P NMR signals).

## 6.4 Discussion

It is remarkable that a slight change in substitution on the ligand results in a drastic change in reactivity for the addition of H<sub>2</sub>. With <sup>1</sup>Bu groups exclusively on the ligand for <sup>1Bu4</sup>PCPIr(CO), there is effectively no H<sub>2</sub> addition, presumably because the <sup>1</sup>Bu groups are a large steric presence which inhibits access to the metal center. However, it is not the mere presence of two <sup>1</sup>Bu on one side of the complex that prevents H<sub>2</sub> addition. Given that H<sub>2</sub> addition to *both* sides of the <sup>1Bu3Me</sup>PCPIr(CO) complex is kinetically facile, and given that addition of H<sub>2</sub> to the side with bis-<sup>1</sup>Bu groups is the kinetically (and thermodynamically) favored isomer, it is clear that the kinetic picture is more complicated than simply the presence of sterically large groups on one side of the complex. It is likely that the lone methyl group in <sup>1Bu3Me</sup>PCPIr(CO) allows the ligand's phosphine substitutents to reorganize in the direction of the less-hindered quadrant to relieve steric strain.

## 6.5 Summary

A revised synthesis scheme for the synthesis of the  $^{tBu3Me}PCP$  ligand was developed. The revised synthesis scheme was found to be amenable for the synthesis of a variety of unsymmetrical pincer ligands. With the  $^{tBu3Me}PCP$  ligand, addition of H<sub>2</sub> to  $^{tBu3Me}PCPIr(CO)$  and  $^{tBu4}PCPIr(CO)$  was investigated. Addition of H<sub>2</sub> was not observed to add to  $^{tBu4}PCPIr(CO)$ . However, when H<sub>2</sub> was added to  $^{tBu3Me}PCPIr(CO)$ , three isomers were observed and assigned using an isotopic labeling experiment and NMR spectroscopy. The kinetic products were the two possible isomers of *cis*-H<sub>2</sub> addition. The thermodynamic product resulted from *trans*-H<sub>2</sub> addition.

## 6.6 Experimental

All reactions, recrystallizations, and routine manipulations were conducted under argon using an argon-filled glove box or by using standard Schlenk techniques. Reagent grade solvents were used and dried according to established methods, then degassed with argon. All NMR solvents were dried using Na/K alloy, vacuum-transferred, and stored in an argon-filled glove box. Column chromatography was performed as described by Still *et al.* using Silicycle 40-63  $\mu$ m 60 Å silica gel (SilaFlash P60).<sup>22</sup> TLC analysis of reaction mixtures was performed on Silicycle silica gel 60 Å F<sub>254</sub> TLC plates and visualized using p-anisaldehyde, KMnO<sub>4</sub>, and 254 nm UV light. <sup>1</sup>H, <sup>31</sup>P{<sup>1</sup>H}, and <sup>13</sup>C{<sup>1</sup>H} NMR spectra were obtained on 400-MHz or 500-MHz Varian Spectrometers. <sup>1</sup>H and <sup>13</sup>C NMR chemical shifts are reported in ppm downfield from tetramethylsilane and were referenced to residual protiated (<sup>1</sup>H) or deuterated solvent (<sup>13</sup>C). <sup>31</sup>P{<sup>1</sup>H} NMR chemical shifts are referenced to an external standard, Me<sub>3</sub>P in mesitylene- $d_{12}$  solvent ( $\delta$  - 62.4 ppm), in a capillary tube.

X-ray diffraction data were collected on a Bruker Smart APEX CCD diffractometer with graphite monochromatized Mo K $\alpha$  radiation ( $\lambda = 0.71073$ Å) at 100 K. Crystals were immersed in Paratone oil, placed on a glass needle, and examined at 100 K. The data were corrected for Lorenz effects, polarization, and absorption, the latter by a multiscan (SADABS) method.<sup>23</sup> The structures were solved by direct methods (SHELXS86).<sup>24</sup> All non-hydrogen atoms were refined (SHELXL97)<sup>25</sup> based upon F<sub>obs</sub>.<sup>24</sup> All hydrogen atom coordinates were calculated with idealized geometries (SHELXL97). Scattering factors (f<sub>o</sub>, f', f") are as described in SHELXL97.<sup>25</sup>

Synthesis and characterization of <sup>tBu4</sup>PCPIrCO and compounds **7**, **8**, and **9** matched previously reported procedures.<sup>16,26</sup> Compound **6** was prepared from compound **5** using the deprotection procedure detailed by Jensen *et al.*,<sup>17</sup> and its spectral data was previously reported by our group.<sup>16</sup> <sup>t</sup>BuMePH was prepared as previously reported by our group.<sup>16</sup> <sup>t</sup>BuMePH was prepared as previously reported by Higham *et al.*<sup>19</sup>

Representative procedure for the formation of lithiated phosphine-boranes ( $Li^{t}Bu_{2}P \cdot BH_{3}$ ,  $Li^{t}BuMeP \cdot BH_{3}$ ). A solution of  ${}^{t}Bu_{2}PH \cdot BH_{3}$  (2.2000 g, 13.7461 mmol) in 20 mL of THF was cooled to 0 °C. A solution of  ${}^{n}BuLi$  in hexanes (2.5 M, 5.5 mL, 13.75 mmol) was added via syringe. The reaction solution was allowed to warm to room temperature and stir for 1 h.

Compound 2. Methyl 3-(bromomethyl)benzoate (3.00 g, 13.10 mmol) was dissolved in 10 mL of THF and cooled to 0 °C. This solution was transferred via cannula to a flask charged with a solution of  $Li^{t}Bu_{2}P \cdot BH_{3}$  (13.75 mmol) in 20 mL of THF that was prepared separately and cooled to 0 °C. After 90 min, 20 mL of water was added. The aqueous layer was extracted with dichloromethane (3 x 10 mL), and the combined organic layers were dried over magnesium sulfate, filtered, and concentrated. Column chromatography (20:1 hexanes: ethyl acetate) afforded the product as a white solid (3.66 g, 11.88 mmol) in 91 % yield. Crystals suitable for single-crystal X-ray analysis were obtained from the slow evaporation of a solution of compound 2 in pentane. An ORTEP diagram of this complex is shown in Figure 6.4. X-ray crystallography refinement parameters and various x-ray crystal structure parameters are listed in Tables 6.1 - 6.6. <sup>31</sup>P{<sup>1</sup>H} NMR (CDCl<sub>3</sub>, 500 MHz):  $\delta$  47.58 (q). <sup>1</sup>H NMR (CDCl<sub>3</sub>, 500 MHz):  $\delta$  8.03 (d, J = 1.5 Hz, 1H), 7.89 (d, J = 8.8 Hz, 1H), 7.76 (d, J = 7.3 Hz, 1H), 7.37 (t, J = 7.7 Hz, 1H), 3.91 (s, 3H), 3.18 (d, J = 12.1 Hz, 2H), 1.26 (d, 12.1 Hz, 18H), 0.54 (br q, J = 91.3 Hz, 3H). <sup>13</sup>C{<sup>1</sup>H} NMR (CDCl<sub>3</sub>, 500 MHz):  $\delta$  167.12 (s), 135.50 (d, J = 3.3 Hz), 135.41 (d, J = 2.8 Hz), 131.63 (d, J = 4.7 Hz), 130.10 (s), 128.34 (d, J = 1.9 Hz), 128.06 (d, J = 2.3 Hz), 52.32 (s), 33.04 (d, J = 24.7 Hz), 28.42 (d, J = 1.4 Hz), 25.98 (d, J = 24.7 Hz).

**Compound 3.** A flask was charged with compound **2** (1.5077 g, 4.8919 mmol) and 20 mL of toluene and cooled to 0 °C. This solution was transferred via cannula to a flask containing a solution of di-isobutyl aluminum hydride (12.5 mL, 1.0 M in cyclohexane, 12.5 mmol) that was diluted with 20 mL of toluene and cooled to 0 °C. The reaction solution was allowed to slowly warm to room temparature and stir for 6 h. Diethyl ether

(40 mL) was added, and the reaction solution was cooled to 0 °C. Water (0.5 mL), a 15% sodium hydroxide solution (0.5 mL), and a second portion of water (1.25 mL) were added successively. The reaction mixture was then allowed to warm to room temperature and stir for an additional 15 min. The reaction mixture was dried over magnesium sulfate, filtered, and concentrated. Column chromatography (20:1 hexanes:ethyl acetate) afforded the product as a white solid (1.2059 g, 4.3038 mmol) in 88% yield. Crystals suitable for single-crystal X-ray analysis were obtained from the slow evaporation of a solution of compound 3 in ethyl acetate. Compound 3 was also obtained in a one-pot reaction via the addition of di-isobutyl aluminum hydride directly to the reaction solution of compound 2 without isolation and purification of compound 2 (80 % yield from the methyl 3-(bromomethyl)benzoate starting material). An ORTEP diagram of this complex is shown in Figure 6.5. X-ray crystallography refinement parameters and various x-ray crystal structure parameters are listed in Tables 6.7 – 6.12.  ${}^{31}P{}^{1}H{}$  NMR (CDCl<sub>3</sub>, 500 **MHz**):  $\delta$  47.11 (q). <sup>1</sup>**H NMR (CDCl<sub>3</sub>, 500 MHz)**:  $\delta$  7.45 (d, J = 1.1 Hz, 1H), 7.36 (dd, J = 1.3 Hz, J = 7.5 Hz, 1H), 7.26 (t, J = 7.5 Hz, 1H), 7.20 (dd, J = 0.7 Hz, J = 7.7 Hz, 1H), 4.66 (s, 2H), 3.14 (d, J = 12.1 Hz, 2H), 1.83 (s, 1H), 1.25 (d, J = 12.5 Hz, 18H), 0.53 (br q, J = 92.6 Hz, 3H). <sup>13</sup>C{<sup>1</sup>H} NMR (CDCl<sub>3</sub>, 500 MHz):  $\delta$  140.90 (d, J=1.9 Hz), 135.12 (d, J=3.7 Hz), 129.98 (d, J=4.2 Hz), 129.43 (d, J=4.2 Hz), 128.39 (d, J=1.9 Hz), 125.33 (d, J=1.8 Hz), 65.29 (s), 32.96 (d, J=25.2 Hz), 28.41 (d, J=1.4 Hz), 26.02 (d, J=24.7 Hz).

**Compound 4.** Compound **3** (1.2059 g, 4.3038 mmol) was dissolved in 40 mL of chloroform and cooled to 0 °C. Phosphorus tribromide (0.42 mL, 4.4686 mmol) was added, and the reaction mixture was allowed to stir for 15 min. Water (1 mL) was added.

The reaction mixture was then dried over magnesium sulfate, filtered, and concentrated. Column chromatography (20:1 hexanes:ethyl acetate) afforded the product as a white solid (1.2312 g, 3.5885 mmol) in 83% yield. Crystals suitable for single-crystal X-ray analysis were obtained from the slow evaporation of a solution of compound **2** in chloroform. An ORTEP diagram of this complex is shown in Figure 6.6. X-ray crystallography refinement parameters and various x-ray crystal structure parameters are listed in Tables 6.13 – 6.18. <sup>31</sup>P NMR (CDCl<sub>3</sub>, **500** MHz):  $\delta$  47.29 (q). <sup>1</sup>H NMR (CDCl<sub>3</sub>, **500** MHz):  $\delta$  7.46 (s, 1H), 7.36 (m, 1H), 7.19–7.24 (m, 2H), 4.45 (s, 2H), 3.11 (d, J = 12.1 Hz, 2H), 1.22 (d, J = 12.5 Hz, 18H), 0.52 (br q, J = 93.1, 3H). <sup>13</sup>C NMR (CDCl<sub>3</sub>, **500** MHz):  $\delta$  137.73 (d, J = 1.9 Hz), 135.50 (d, J = 2.8 Hz), 131.52 (d, J = 4.2 Hz), 130.86 (d, J = 3.7 Hz), 128.62 (d, 1.4 Hz), 127.39 (d, J = 1.9 Hz), 33.64 (s), 32.96 (d, J = 24.66 Hz), 28.38 (d, J = 0.9 Hz), 25.99 (d, J = 24.19 Hz).

**Compound 5.** Compound **4** (0.7305 g, 2.1291 mmol) was dissolved in 10 mL of THF and cooled to 0 °C. This solution was transferred via cannula to a flask that was charged with a solution of compound Li<sup>t</sup>BuMeP·BH<sub>3</sub> (1.15 mmol) in 10 mL of THF and cooled to 0 °C. The reaction was stirred at 0 °C for 1 h and then warmed to room temperature. An aqueous solution of hydrochloric acid (1 M, 10 mL) was added. The aqueous layer was extracted with diethyl ether (3 x 10 mL) and ethyl acetate (10 mL). The combined organic layers were dried over magnesium sulfate, filtered, and concentrated. Column chromatography (40:1 hexanes:ethyl acetate) afforded the product as a white solid (0.4972 g, 1.3079 mmol) in 61% yield. <sup>31</sup>P NMR (CDCl<sub>3</sub>, 500 MHz):  $\delta$  7.30–7.35 (m, 2H), 7.20

(t, J = 7.5 Hz, 1H), 7.14 (bd, J = 7.7 Hz, 1H), 3.11 (d, J = 12.5 Hz, 2H), 3.02 (dd, J = 8.8 Hz, J = 13.6 Hz, 1H), 2.93 (t, J = 13.9 Hz, 1H), 1.26 (d, J = 12.1 Hz, 9H), 1.23 (d, J = 12.5 Hz, 9H), 1.19 (d, J = 13.6 Hz, 9H), 1.05 (d, J = 9.5 Hz, 3H), 0–1.0 (br m, 6H). <sup>13</sup>C NMR (CDCl<sub>3</sub>, 500 MHz):  $\delta$  135.14 (dd, J = 2.6 Hz, J = 3.5 Hz), 133.17 (dd, J = 1.6 Hz, J = 5.4 Hz), 132.13 (t, J = 4.0 Hz), 129.31 (dd, J = 2.8 Hz, J = 4.2 Hz), 128.65 (dd, J = 2.1 Hz, J = 4.0 Hz), 128.27 (dd, J = 1.6 Hz, J = 2.4 Hz), 33.05 (d, J = 16.3 Hz), 32.85 (d, J = 17.2 Hz), 29.00 (d, J = 28.4 Hz), 28.4 (d, J = 0.9 Hz), 28.35 (d, J = 0.9 Hz), 27.92 (d, J = 31.6 Hz), 26.00 (d, J = 24.7 Hz), 25.31 (d, J = 1.9 Hz), 4.87 (d, J = 34.9 Hz).

Representative procedure for the addition of  $H_2$  to PCPIr(CO) complexes. PCPIr(CO) (20 µmol) was dissolved in 0.7 mL of toluene- $d_8$  in a J Young NMR tube. The reaction solution was degassed and cooled to -78 °C, and 1 atm of  $H_2$  was added. Signals corresponding to the formation of *cis* isomers were first observed to appear upon heating the reaction to -10 °C. After 15 h at 0 °C, only *cis* isomers were observed by NMR spectroscopy. Heating the reaction mixture to 50 °C for 2 d resulted in complete conversion of both *cis* isomers to the *trans*  $H_2$  addition isomer.

Selected NMR spectral data for <sup>tBu3Me</sup>PCPIr(CO)(H<sub>2</sub>) complexes. <sup>31</sup>P{<sup>1</sup>H} NMR (toluene-d<sub>8</sub>, 400 MHz):  $\delta$  31.8 (*cis* addition to <sup>t</sup>BuMe side isomer, d, J<sub>PP</sub> = 285.5 Hz), 33.7 (*cis* addition to bis-<sup>t</sup>Bu side isomer, d, J<sub>PP</sub> = 283.0 Hz), 38.4 (*trans* addition isomer, d, J<sub>PP</sub> = 288.1 Hz), 73.6 (*cis* addition to bis-<sup>t</sup>Bu side isomer, d, J<sub>PP</sub> = 283.0 Hz), 75.4 (*cis* addition to <sup>t</sup>BuMe side isomer, d, J<sub>PP</sub> = 285.5 Hz), 76.4 (*trans* addition isomer, d, J<sub>PP</sub> = 288.1 Hz). <sup>1</sup>H NMR (toluene-d<sub>8</sub>, 400 MHz):  $\delta$  -9.46 (*trans* addition isomer, m, J<sub>HP1</sub> = 12.8 Hz,  $J_{HP2} = 16.8$  Hz,  $J_{HH} = 5.5$  Hz), -9.58 (*trans* addition isomer, dt,  $J_{HP1} = J_{HP2} = 14.0$  Hz,  $J_{HH} = 5.5$  Hz), -10.57 (*cis* addition to bis-<sup>t</sup>Bu side isomer– hydride *trans* to CO, dt,  $J_{HP1} = J_{HP2} = 17.0$  Hz,  $J_{HH} = 3.1$  Hz), -10.86 (*cis* addition to <sup>t</sup>BuMe side isomer– hydride *trans* to CO, dt,  $J_{HP1} = J_{HP2} = 12.4$  Hz,  $J_{HH} = 3.4$  Hz), -11.63 (*cis* addition to <sup>t</sup>BuMe side isomer– hydride side isomer– hydride *cis* to CO, dt,  $J_{HP1} = J_{HP2} = 12.4$  Hz,  $J_{HH} = 3.4$  Hz), -11.63 (*cis* addition to <sup>t</sup>BuMe side isomer– hydride cis to CO, dt,  $J_{HP1} = J_{HP2} = 12.4$  Hz,  $J_{HH} = 3.4$  Hz), -11.74 (*cis* addition to bis-<sup>t</sup>Bu side isomer– hydride *cis* to CO, dt,  $J_{HP1} = J_{HP2} = 14.5$  Hz,  $J_{HH} = 3.1$  Hz).





 Table 6.1. Crystal data and structure refinement for 2.

Identification code	dyw2-139		
Empirical formula	C17 H30 B O2 P		
Formula weight	308.19		
Temperature	100(2) K		
Wavelength	0.71073 Å		
Crystal system	Monoclinic		
Space group	P2(1)/n		
Unit cell dimensions	a = 14.5265(11) Å	$\alpha = 90^{\circ}$ .	
	b = 6.8077(5) Å	$\beta = 99.530(1)^{\circ}$ .	
	c = 18.7924(14)  Å	$\gamma = 90^{\circ}$ .	
Volume	1832.8(2) Å <sup>3</sup>		
Z	4		
Density (calculated)	1.117 Mg/m <sup>3</sup>		
Absorption coefficient	0.152 mm <sup>-1</sup>		
F(000)	672		
Crystal size	0.40 x 0.25 x 0.22 mm <sup>3</sup>		
Theta range for data collection	1.94 to 32.03°.		
Index ranges	-20<=h<=21, -10<=k<=10, -27<=1<=28		
Reflections collected	28942		
Independent reflections	6340 [R(int) = 0.0248]		
Completeness to theta = $32.03^{\circ}$	99.4 %		
Absorption correction	Semi-empirical from equivalents		
Max. and min. transmission	0.9674 and 0.9417		
Refinement method	Full-matrix least-squares on F <sup>2</sup>		
Data / restraints / parameters	6340 / 502 / 241		
Goodness-of-fit on F <sup>2</sup>	1.006		
Final R indices [I>2sigma(I)]	R1 = 0.0470, wR2 = 0.1178		
R indices (all data)	R1 = 0.0571, wR2 = 0.1247		
Largest diff. peak and hole	0.630 and -0.210 e.Å <sup>-3</sup>		

**Table 6.2.** Atomic coordinates  $(x10^4)$  and equivalent isotropic displacement parameters  $(\text{\AA}^2 x \ 10^3)$  for **2**. U(eq) is defined as one third of the trace of the orthogonalized U<sup>ij</sup> tensor.

	Х	У	Z	U(eq)
P(1)	9568(1)	3810(1)	2553(1)	20(1)
B(1)	9131(1)	1196(2)	2351(1)	46(1)
O(1)	8765(1)	9842(2)	-41(1)	46(1)
O(2)	7588(1)	8852(2)	-892(1)	50(1)
C(1)	8718(1)	5698(2)	2182(1)	24(1)
C(2)	8268(1)	5562(2)	1399(1)	24(1)
C(3)	8398(1)	7060(2)	922(1)	25(1)
C(4)	7937(1)	7034(2)	208(1)	29(1)
C(5)	7337(1)	5492(2)	-32(1)	38(1)
C(6)	7211(1)	3996(2)	436(1)	42(1)
C(7)	7669(1)	4025(2)	1150(1)	34(1)
C(8)	8057(1)	8651(2)	-302(1)	35(1)
C(9A)	10663(2)	4127(5)	2166(2)	26(1)
C(10A)	10431(8)	4040(13)	1332(3)	32(1)
C(11A)	11159(2)	6061(5)	2376(1)	37(1)
C(12A)	11324(2)	2398(6)	2424(1)	45(1)
C(13A)	9706(3)	4436(5)	3537(1)	24(1)
C(14A)	9853(2)	6639(4)	3708(1)	31(1)
C(15A)	10518(2)	3284(6)	3972(1)	34(1)
C(16A)	8799(4)	3807(12)	3802(6)	35(1)
C(9B)	10635(3)	4515(9)	2175(4)	26(1)
C(10B)	10390(15)	3850(20)	1380(6)	32(1)
C(11B)	10853(3)	6716(6)	2209(2)	30(1)
C(12B)	11509(3)	3385(8)	2522(2)	31(1)
C(13B)	9742(6)	4070(9)	3560(2)	24(1)
C(14B)	10168(4)	6042(7)	3820(2)	32(1)
C(15B)	10361(3)	2377(9)	3901(2)	36(1)
C(16B)	8775(7)	3850(20)	3785(11)	35(1)
C(17)	8946(2)	11434(3)	-509(1)	58(1)

P(1)-C(1)	1.8375(10)	C(5)-C(6)	1.378(2)
P(1)-C(9A)	1.867(2)	C(6)-C(7)	1.3966(17)
P(1)-C(9B)	1.871(3)	C(9A)-C(11A)	1.521(4)
P(1)-C(13B)	1.876(3)	C(9A)-C(12A)	1.545(4)
P(1)-C(13A)	1.876(2)	C(9A)-C(10A)	1.549(4)
P(1)-B(1)	1.9057(14)	C(13A)-C(15A)	1.536(4)
O(1)-C(8)	1.3376(18)	C(13A)-C(16A)	1.544(4)
O(1)-C(17)	1.4467(18)	C(13A)-C(14A)	1.541(3)
O(2)-C(8)	1.2097(15)	C(9B)-C(11B)	1.531(6)
C(1)-C(2)	1.5104(13)	C(9B)-C(12B)	1.535(6)
C(2)-C(7)	1.3910(15)	C(9B)-C(10B)	1.546(6)
C(2)-C(3)	1.3919(15)	C(13B)-C(14B)	1.525(6)
C(3)-C(4)	1.3971(14)	C(13B)-C(15B)	1.535(6)
C(4)-C(5)	1.3898(18)	C(13B)-C(16B)	1.539(6)
C(4)-C(8)	1.4882(18)		
		C(7)-C(2)-C(1)	121.30(10)
C(1)-P(1)-C(9A)	109.49(12)	C(3)-C(2)-C(1)	120.00(9)
C(1)-P(1)-C(9B)	102.9(2)	C(2)-C(3)-C(4)	121.02(10)
C(9A)-P(1)-C(9B)	8.2(3)	C(5)-C(4)-C(3)	119.75(11)
C(1)-P(1)-C(13B)	106.6(2)	C(5)-C(4)-C(8)	118.95(11)
C(9A)-P(1)-C(13B)	113.6(3)	C(3)-C(4)-C(8)	121.29(11)
C(9B)-P(1)-C(13B)	112.1(3)	C(6)-C(5)-C(4)	119.58(11)
C(1)-P(1)-C(13A)	99.78(12)	C(5)-C(6)-C(7)	120.68(12)
C(9A)-P(1)-C(13A)	113.47(19)	C(2)-C(7)-C(6)	120.41(12)
C(9B)-P(1)-C(13A)	110.8(3)	O(2)-C(8)-O(1)	123.39(13)
C(13B)-P(1)-C(13A)	7.9(3)	O(2)-C(8)-C(4)	124.39(14)
C(1)-P(1)-B(1)	113.42(6)	O(1)-C(8)-C(4)	112.20(10)
C(9A)-P(1)-B(1)	107.90(14)	C(11A)-C(9A)-C(12A)	109.6(3)
C(9B)-P(1)-B(1)	115.7(2)	C(11A)-C(9A)-C(10A)	108.0(4)
C(13B)-P(1)-B(1)	105.9(2)	C(12A)-C(9A)-C(10A)	108.0(4)
C(13A)-P(1)-B(1)	112.72(13)	C(11A)-C(9A)-P(1)	113.4(2)
C(8)-O(1)-C(17)	115.89(12)	C(12A)-C(9A)-P(1)	108.34(19)
C(2)-C(1)-P(1)	118.78(7)	C(10A)-C(9A)-P(1)	109.4(6)
		C(15A)-C(13A)-C(16A)	107.9(4)
C(7)-C(2)-C(3)	118.56(10)	C(15A)-C(13A)-C(14A)	108.6(3)

Table 6.3. Bond lengths  $(\text{\AA})$  and angles  $(^{\circ})$  for 2.

C(16A)-C(13A)-C(14A)	107.4(4)	C(10B)-C(9B)-P(1)	102.7(10)
C(15A)-C(13A)-P(1)	110.9(2)	C(14B)-C(13B)-C(15B)	110.4(5)
C(16A)-C(13A)-P(1)	107.3(5)	C(14B)-C(13B)-C(16B)	109.7(7)
C(14A)-C(13A)-P(1)	114.54(18)	C(15B)-C(13B)-C(16B)	108.1(7)
C(11B)-C(9B)-C(12B)	108.8(4)	C(14B)-C(13B)-P(1)	112.8(4)
C(11B)-C(9B)-C(10B)	109.9(7)	C(15B)-C(13B)-P(1)	108.8(3)
C(12B)-C(9B)-C(10B)	107.8(7)	C(16B)-C(13B)-P(1)	106.9(9)
C(11B)-C(9B)-P(1)	114.5(3)		
C(12B)-C(9B)-P(1)	112.9(4)		

_	U <sup>11</sup>	U <sup>22</sup>	U <sup>33</sup>	U <sup>23</sup>	U <sup>13</sup>	U <sup>12</sup>
P(1)	19(1)	19(1)	21(1)	0(1)	-1(1)	0(1)
B(1)	47(1)	21(1)	63(1)	1(1)	-16(1)	-2(1)
O(1)	58(1)	41(1)	39(1)	15(1)	2(1)	4(1)
O(2)	47(1)	75(1)	28(1)	13(1)	2(1)	22(1)
C(1)	23(1)	26(1)	22(1)	-2(1)	-4(1)	5(1)
C(2)	18(1)	29(1)	22(1)	-2(1)	-3(1)	3(1)
C(3)	20(1)	29(1)	24(1)	-1(1)	-1(1)	6(1)
C(4)	24(1)	38(1)	23(1)	0(1)	1(1)	12(1)
C(5)	31(1)	55(1)	24(1)	-7(1)	-7(1)	5(1)
C(6)	35(1)	51(1)	35(1)	-8(1)	-11(1)	-11(1)
C(7)	28(1)	39(1)	31(1)	-1(1)	-5(1)	-8(1)
C(8)	35(1)	46(1)	26(1)	4(1)	6(1)	20(1)
C(9A)	20(1)	36(1)	23(1)	-4(1)	2(1)	6(1)
C(10A)	30(1)	42(2)	22(1)	0(1)	6(1)	7(1)
C(11A)	26(1)	55(2)	30(1)	2(1)	5(1)	-14(1)
C(12A)	32(1)	63(2)	38(1)	8(1)	6(1)	25(1)
C(13A)	22(1)	30(1)	19(1)	8(1)	0(1)	-1(1)
C(14A)	35(1)	35(1)	21(1)	-6(1)	1(1)	-2(1)
C(15A)	34(1)	43(2)	25(1)	10(1)	-1(1)	6(1)
C(16A)	29(1)	47(1)	30(1)	8(1)	8(1)	-3(1)
C(9B)	20(1)	36(1)	23(1)	-4(1)	2(1)	6(1)
C(10B)	30(1)	42(2)	22(1)	0(1)	6(1)	7(1)
C(11B)	26(2)	32(2)	33(2)	7(1)	9(1)	-1(1)
C(12B)	22(1)	37(2)	34(2)	2(2)	1(1)	9(1)
C(13B)	22(1)	30(1)	19(1)	8(1)	0(1)	-1(1)
C(14B)	42(2)	33(2)	20(2)	-2(1)	1(1)	-5(2)
C(15B)	43(2)	34(2)	27(2)	12(2)	-3(1)	12(2)
C(16B)	29(1)	47(1)	30(1)	8(1)	8(1)	-3(1)
C(17)	76(1)	47(1)	52(1)	24(1)	15(1)	10(1)

**Table 6.4.** Anisotropic displacement parameters ( $Å^2 x \ 10^3$ ) for **2**. The anisotropic displacement factor exponent takes the form:  $-2\pi^2 [h^2 a^{*2} U^{11} + ... + 2 h k a^* b^* U^{12}]$ 

	Х	У	Z	U(eq)
H(1B1)	8504	1060	2473	70
H(2B1)	9555	269	2639	70
H(3B1)	9110	915	1837	70
H(1A)	8213	5700	2476	29
H(1B)	9034	6989	2253	29
H(3)	8807	8116	1083	30
H(5)	7017	5471	-516	46
H(6)	6807	2934	271	51
H(7)	7572	2989	1468	40
H(10A)	10112	2801	1184	47
H(10B)	11010	4121	1130	47
H(10C)	10023	5144	1154	47
H(11A)	11291	6171	2903	56
H(11B)	10759	7155	2177	56
H(11C)	11746	6104	2184	56
H(12A)	11000	1153	2297	67
H(12B)	11513	2472	2949	67
H(12C)	11878	2474	2191	67
H(14A)	9345	7396	3427	46
H(14B)	10450	7062	3581	46
H(14C)	9858	6860	4224	46
H(15A)	10427	1876	3877	52
H(15B)	10540	3537	4488	52
H(15C)	11107	3703	3829	52
H(16A)	8262	4415	3498	53
H(16B)	8822	4231	4303	53
H(16C)	8738	2374	3774	53
H(10D)	9858	4612	1136	47
H(10E)	10229	2450	1362	47
H(10F)	10929	4062	1137	47
H(11D)	10297	7454	1994	45
H(11E)	11364	6989	1941	45

**Table 6.5.** Hydrogen coordinates  $(x10^4)$  and isotropic displacement parameters  $(\text{\AA}^2 x 10^3)$  for **2**.

H(11F)	11039	7115	2713	45	
H(12D)	11626	3633	3043	47	
H(12E)	12047	3825	2311	47	
H(12F)	11412	1975	2434	47	
H(14D)	10739	6261	3617	48	
H(14E)	10318	6038	4348	48	
H(14F)	9720	7095	3663	48	
H(15D)	10077	1120	3731	53	
H(15E)	10420	2446	4428	53	
H(15F)	10980	2483	3762	53	
H(16D)	8427	2804	3501	53	
H(16E)	8432	5092	3698	53	
H(16F)	8852	3523	4299	53	
H(17A)	8385	12250	-625	86	
H(17B)	9110	10894	-954	86	
H(17C)	9463	12235	-264	86	
C(9A)-P(1)-C(1)-C(2)	70.72(17)	C(13B)-P(1)-C(9A)-C(10A)	-175.8(4)		
--------------------------	-------------	---------------------------	-----------		
C(9B)-P(1)-C(1)-C(2)	75.9(3)	C(13A)-P(1)-C(9A)-C(10A)	-167.2(4)		
C(13B)-P(1)-C(1)-C(2)	-166.0(3)	B(1)-P(1)-C(9A)-C(10A)	67.1(4)		
C(13A)-P(1)-C(1)-C(2)	-169.97(16)	C(1)-P(1)-C(13A)-C(15A)	-170.3(2)		
B(1)-P(1)-C(1)-C(2)	-49.85(12)	C(9A)-P(1)-C(13A)-C(15A)	-53.9(3)		
P(1)-C(1)-C(2)-C(7)	65.92(13)	C(9B)-P(1)-C(13A)-C(15A)	-62.4(3)		
P(1)-C(1)-C(2)-C(3)	-118.48(10)	C(13B)-P(1)-C(13A)-C(15A)	39(3)		
C(7)-C(2)-C(3)-C(4)	0.26(15)	B(1)-P(1)-C(13A)-C(15A)	69.1(3)		
C(1)-C(2)-C(3)-C(4)	-175.46(9)	C(1)-P(1)-C(13A)-C(16A)	72.1(4)		
C(2)-C(3)-C(4)-C(5)	0.01(16)	C(9A)-P(1)-C(13A)-C(16A)	-171.5(4)		
C(2)-C(3)-C(4)-C(8)	178.69(10)	C(9B)-P(1)-C(13A)-C(16A)	-180.0(4)		
C(3)-C(4)-C(5)-C(6)	-0.53(18)	C(13B)-P(1)-C(13A)-C(16A)	-79(3)		
C(8)-C(4)-C(5)-C(6)	-179.24(12)	B(1)-P(1)-C(13A)-C(16A)	-48.5(4)		
C(4)-C(5)-C(6)-C(7)	0.8(2)	C(1)-P(1)-C(13A)-C(14A)	-47.0(3)		
C(3)-C(2)-C(7)-C(6)	-0.02(18)	C(9A)-P(1)-C(13A)-C(14A)	69.3(3)		
C(1)-C(2)-C(7)-C(6)	175.65(11)	C(9B)-P(1)-C(13A)-C(14A)	60.9(4)		
C(5)-C(6)-C(7)-C(2)	-0.5(2)	C(13B)-P(1)-C(13A)-C(14A)	162(3)		
C(17)-O(1)-C(8)-O(2)	0.39(19)	B(1)-P(1)-C(13A)-C(14A)	-167.7(2)		
C(17)-O(1)-C(8)-C(4)	178.66(12)	C(1)-P(1)-C(9B)-C(11B)	43.8(5)		
C(5)-C(4)-C(8)-O(2)	9.03(18)	C(9A)-P(1)-C(9B)-C(11B)	-172(3)		
C(3)-C(4)-C(8)-O(2)	-169.67(11)	C(13B)-P(1)-C(9B)-C(11B)	-70.4(5)		
C(5)-C(4)-C(8)-O(1)	-169.22(11)	C(13A)-P(1)-C(9B)-C(11B)	-62.1(5)		
C(3)-C(4)-C(8)-O(1)	12.09(15)	B(1)-P(1)-C(9B)-C(11B)	168.1(4)		
C(1)-P(1)-C(9A)-C(11A)	63.9(3)	C(1)-P(1)-C(9B)-C(12B)	169.0(4)		
C(9B)-P(1)-C(9A)-C(11A)	26(2)	C(9A)-P(1)-C(9B)-C(12B)	-47(2)		
C(13B)-P(1)-C(9A)-C(11A)	-55.1(3)	C(13B)-P(1)-C(9B)-C(12B)	54.8(5)		
C(13A)-P(1)-C(9A)-C(11A)	-46.5(3)	C(13A)-P(1)-C(9B)-C(12B)	63.1(5)		
B(1)-P(1)-C(9A)-C(11A)	-172.2(2)	B(1)-P(1)-C(9B)-C(12B)	-66.8(5)		
C(1)-P(1)-C(9A)-C(12A)	-174.2(2)	C(1)-P(1)-C(9B)-C(10B)	-75.2(7)		
C(9B)-P(1)-C(9A)-C(12A)	148(3)	C(9A)-P(1)-C(9B)-C(10B)	69(3)		
C(13B)-P(1)-C(9A)-C(12A)	66.7(3)	C(13B)-P(1)-C(9B)-C(10B)	170.6(7)		
C(13A)-P(1)-C(9A)-C(12A)	75.3(3)	C(13A)-P(1)-C(9B)-C(10B)	178.9(6)		
B(1)-P(1)-C(9A)-C(12A)	-50.4(3)	B(1)-P(1)-C(9B)-C(10B)	49.0(7)		
C(1)-P(1)-C(9A)-C(10A)	-56.7(4)	C(1)-P(1)-C(13B)-C(14B)	-63.5(5)		
C(9B)-P(1)-C(9A)-C(10A)	-94(3)	C(9A)-P(1)-C(13B)-C(14B)	57.2(5)		

C(9B)-P(1)-C(13B)-C(14B)	48.4(5)	B(1)-P(1)-C(13B)-C(15B)	52.5(5)
C(13A)-P(1)-C(13B)-C(14B)	-34(2)	C(1)-P(1)-C(13B)-C(16B)	57.1(6)
B(1)-P(1)-C(13B)-C(14B)	175.4(4)	C(9A)-P(1)-C(13B)-C(16B)	177.8(6)
C(1)-P(1)-C(13B)-C(15B)	173.6(4)	C(9B)-P(1)-C(13B)-C(16B)	169.0(6)
C(9A)-P(1)-C(13B)-C(15B)	-65.7(5)	C(13A)-P(1)-C(13B)-C(16B)	87(3)
C(9B)-P(1)-C(13B)-C(15B)	-74.5(5)	B(1)-P(1)-C(13B)-C(16B)	-64.0(6)
C(13A)-P(1)-C(13B)-C(15B)	-156(3)		



**Figure 6.5.** ORTEP diagram of compound **3**.

 Table 6.7. Crystal data and structure refinement for 3.

Identification code	dyw153	dyw153		
Empirical formula	C16 H30 B O P			
Formula weight	280.18			
Temperature	100(2) K			
Wavelength	0.71073 Å			
Crystal system	Monoclinic			
Space group	P2(1)/n			
Unit cell dimensions	a = 8.119(5) Å	<i>α</i> = 90°.		
	b = 13.889(9) Å	$\beta = 99.253(11)^{\circ}.$		
	c = 15.007(10)  Å	$\gamma = 90^{\circ}$ .		
Volume	1670.2(18) Å <sup>3</sup>			
Ζ	4			
Density (calculated)	1.114 Mg/m <sup>3</sup>			
Absorption coefficient	0.156 mm <sup>-1</sup>			
F(000)	616			
Crystal size	0.31 x 0.27 x 0.21 mm <sup>3</sup>			
Theta range for data collection	2.01 to 31.00°.			
Index ranges	-11<=h<=11, -20<=k<=	20, -21<=1<=21		
Reflections collected	20000			
Independent reflections	5320 [R(int) = 0.0218]			
Completeness to theta = $31.00^{\circ}$	99.9 %			
Absorption correction	None			
Max. and min. transmission	0.9680 and 0.9532			
Refinement method	Full-matrix least-squares	s on F <sup>2</sup>		
Data / restraints / parameters	5320 / 0 / 188			
Goodness-of-fit on F <sup>2</sup>	1.014			
Final R indices [I>2sigma(I)]	R1 = 0.0454, wR2 = 0.1	152		
R indices (all data)	R1 = 0.0506, wR2 = 0.1	192		
Largest diff. peak and hole	0.641 and -0.516 e.Å <sup>-3</sup>	0.641 and -0.516 e.Å <sup>-3</sup>		

**Table 6.8.** Atomic coordinates  $(x10^4)$  and equivalent isotropic displacement parameters  $(\text{\AA}^2 x \ 10^3)$  for **3**. U(eq) is defined as one third of the trace of the orthogonalized U<sup>ij</sup> tensor.

	Х	У	Z	U(eq)
P(1)	768(1)	9696(1)	2481(1)	16(1)
O(1)	6012(1)	8385(1)	-902(1)	27(1)
C(1)	3499(1)	9059(1)	661(1)	19(1)
C(2)	1858(1)	9099(1)	834(1)	18(1)
C(3)	757(2)	8371(1)	497(1)	24(1)
C(4)	1297(2)	7610(1)	19(1)	28(1)
C(5)	2943(2)	7562(1)	-124(1)	26(1)
C(6)	4059(1)	8292(1)	191(1)	21(1)
C(7)	1265(2)	9938(1)	1343(1)	21(1)
C(8)	5845(2)	8253(1)	23(1)	28(1)
C(9)	2782(1)	9801(1)	3279(1)	21(1)
C(10)	3951(2)	9032(1)	3024(1)	45(1)
C(11)	3622(2)	10776(1)	3248(2)	60(1)
C(12)	2518(2)	9601(2)	4246(1)	57(1)
C(13)	-739(1)	10681(1)	2630(1)	22(1)
C(14)	-109(2)	11686(1)	2430(1)	42(1)
C(15)	-1187(2)	10664(1)	3583(1)	38(1)
C(16)	-2340(2)	10478(1)	1960(1)	36(1)
B(1)	-233(2)	8454(1)	2613(1)	29(1)

1.8484(15)	C(10)-H(10A)	0.9800
1.8705(15)	C(10)-H(10B)	0.9800
1.8719(14)	C(10)-H(10C)	0.9800
1.9313(17)	C(11)-H(11A)	0.9800
1.4271(17)	C(11)-H(11B)	0.9800
0.8400	C(11)-H(11C)	0.9800
1.3924(17)	C(12)-H(12A)	0.9800
1.3987(17)	C(12)-H(12B)	0.9800
0.9500	C(12)-H(12C)	0.9800
1.3905(16)	C(13)-C(15)	1.532(2)
1.5126(17)	C(13)-C(14)	1.533(2)
1.3871(18)	C(13)-C(16)	1.5359(19)
0.9500	C(14)-H(14A)	0.9800
1.389(2)	C(14)-H(14B)	0.9800
0.9500	C(14)-H(14C)	0.9800
1.3908(18)	C(15)-H(15A)	0.9800
0.9500	C(15)-H(15B)	0.9800
1.5128(19)	C(15)-H(15C)	0.9800
0.9900	C(16)-H(16A)	0.9800
0.9900	C(16)-H(16B)	0.9800
0.9900	C(16)-H(16C)	0.9800
0.9900	B(1)-H(1B)	1.09(2)
1.5186(19)	B(1)-H(2B)	1.127(19)
1.520(2)	B(1)-H(3B)	1.12(2)
1.526(2)		
	1.8484(15) 1.8705(15) 1.8705(15) 1.8719(14) 1.9313(17) 1.4271(17) 0.8400 1.3924(17) 1.3987(17) 0.9500 1.3905(16) 1.5126(17) 1.3871(18) 0.9500 1.389(2) 0.9500 1.3908(18) 0.9500 1.5128(19) 0.9900 0.9900 0.9900 0.9900 1.5186(19) 1.520(2) 1.526(2)	1.8484(15) $C(10)-H(10A)$ $1.8705(15)$ $C(10)-H(10B)$ $1.8719(14)$ $C(10)-H(10C)$ $1.9313(17)$ $C(11)-H(11A)$ $1.4271(17)$ $C(11)-H(11B)$ $0.8400$ $C(11)-H(11C)$ $1.3924(17)$ $C(12)-H(12A)$ $1.3924(17)$ $C(12)-H(12B)$ $0.9500$ $C(12)-H(12C)$ $1.3987(17)$ $C(13)-C(15)$ $1.5126(17)$ $C(13)-C(14)$ $1.3871(18)$ $C(13)-C(16)$ $0.9500$ $C(14)-H(14A)$ $1.389(2)$ $C(14)-H(14B)$ $0.9500$ $C(15)-H(15A)$ $0.9500$ $C(15)-H(15A)$ $0.9500$ $C(15)-H(15B)$ $1.5128(19)$ $C(15)-H(16A)$ $0.9900$ $C(16)-H(16B)$ $0.9900$ $C(16)-H(16B)$ $0.9900$ $C(16)-H(16B)$ $0.9900$ $B(1)-H(1B)$ $1.5186(19)$ $B(1)-H(2B)$ $1.520(2)$ $B(1)-H(3B)$

Table 6.9. Bond lengths  $(\text{\AA})$  and angles  $(^{\circ})$  for 3.

C(7)-P(1)-C(9)	106.19(7)	C(10)-C(9)-C(11)	108.57(15)
C(7)-P(1)-C(13)	102.82(6)	C(10)-C(9)-C(12)	107.45(13)
C(9)-P(1)-C(13)	112.82(6)	C(11)-C(9)-C(12)	108.87(15)
C(7)-P(1)-B(1)	114.48(6)	C(10)-C(9)-P(1)	107.52(10)
C(9)-P(1)-B(1)	110.08(7)	C(11)-C(9)-P(1)	113.62(9)
C(13)-P(1)-B(1)	110.29(8)	C(12)-C(9)-P(1)	110.60(10)
C(8)-O(1)-H(1O)	109.5	C(9)-C(10)-H(10A)	109.5
C(6)-C(1)-C(2)	121.55(10)	C(9)-C(10)-H(10B)	109.5
C(6)-C(1)-H(1)	119.2	H(10A)-C(10)-H(10B)	109.5
C(2)-C(1)-H(1)	119.2	C(9)-C(10)-H(10C)	109.5
C(3)-C(2)-C(1)	118.60(11)	H(10A)-C(10)-H(10C)	109.5
C(3)-C(2)-C(7)	120.34(11)	H(10B)-C(10)-H(10C)	109.5
C(1)-C(2)-C(7)	121.01(10)	C(9)-C(11)-H(11A)	109.5
C(4)-C(3)-C(2)	120.29(11)	C(9)-C(11)-H(11B)	109.5
C(4)-C(3)-H(3)	119.9	H(11A)-C(11)-H(11B)	109.5
C(2)-C(3)-H(3)	119.9	C(9)-C(11)-H(11C)	109.5
C(3)-C(4)-C(5)	120.53(11)	H(11A)-C(11)-H(11C)	109.5
C(3)-C(4)-H(4)	119.7	H(11B)-C(11)-H(11C)	109.5
C(5)-C(4)-H(4)	119.7	C(9)-C(12)-H(12A)	109.5
C(4)-C(5)-C(6)	120.19(11)	C(9)-C(12)-H(12B)	109.5
C(4)-C(5)-H(5)	119.9	H(12A)-C(12)-H(12B)	109.5
C(6)-C(5)-H(5)	119.9	C(9)-C(12)-H(12C)	109.5
C(5)-C(6)-C(1)	118.79(11)	H(12A)-C(12)-H(12C)	109.5
C(5)-C(6)-C(8)	120.42(11)	H(12B)-C(12)-H(12C)	109.5
C(1)-C(6)-C(8)	120.78(11)	C(15)-C(13)-C(14)	109.48(12)
C(2)-C(7)-P(1)	117.49(8)	C(15)-C(13)-C(16)	107.78(11)
C(2)-C(7)-H(7A)	107.9	C(14)-C(13)-C(16)	108.23(12)
P(1)-C(7)-H(7A)	107.9	C(15)-C(13)-P(1)	110.97(9)
C(2)-C(7)-H(7B)	107.9	C(14)-C(13)-P(1)	113.39(10)
P(1)-C(7)-H(7B)	107.9	C(16)-C(13)-P(1)	106.78(10)
H(7A)-C(7)-H(7B)	107.2	C(13)-C(14)-H(14A)	109.5
O(1)-C(8)-C(6)	113.64(10)	C(13)-C(14)-H(14B)	109.5
O(1)-C(8)-H(8A)	108.8	H(14A)-C(14)-H(14B)	109.5
C(6)-C(8)-H(8A)	108.8	C(13)-C(14)-H(14C)	109.5
O(1)-C(8)-H(8B)	108.8	H(14A)-C(14)-H(14C)	109.5
C(6)-C(8)-H(8B)	108.8	H(14B)-C(14)-H(14C)	109.5
H(8A)-C(8)-H(8B)	107.7	C(13)-C(15)-H(15A)	109.5

C(13)-C(15)-H(15B)	109.5	H(16A)-C(16)-H(16C)	109.5
H(15A)-C(15)-H(15B)	109.5	H(16B)-C(16)-H(16C)	109.5
C(13)-C(15)-H(15C)	109.5	P(1)-B(1)-H(1B)	108.0(10)
H(15A)-C(15)-H(15C)	109.5	P(1)-B(1)-H(2B)	104.8(10)
H(15B)-C(15)-H(15C)	109.5	H(1B)-B(1)-H(2B)	110.8(14)
C(13)-C(16)-H(16A)	109.5	P(1)-B(1)-H(3B)	109.0(10)
C(13)-C(16)-H(16B)	109.5	H(1B)-B(1)-H(3B)	112.3(14)
H(16A)-C(16)-H(16B)	109.5	H(2B)-B(1)-H(3B)	111.5(14)
C(13)-C(16)-H(16C)	109.5		

	U <sup>11</sup>	U <sup>22</sup>	U <sup>33</sup>	U <sup>23</sup>	U <sup>13</sup>	U <sup>12</sup>
	10(1)	17(1)	12(1)	2(1)	5(1)	0(1)
P(1)	19(1)	17(1)	13(1)	2(1)	5(1)	0(1)
O(1)	31(1)	27(1)	28(1)	-2(1)	14(1)	-6(1)
C(1)	20(1)	22(1)	14(1)	1(1)	3(1)	-3(1)
C(2)	21(1)	20(1)	13(1)	1(1)	4(1)	-1(1)
C(3)	20(1)	31(1)	20(1)	-4(1)	5(1)	-6(1)
C(4)	31(1)	30(1)	25(1)	-10(1)	10(1)	-12(1)
C(5)	34(1)	24(1)	22(1)	-5(1)	12(1)	-3(1)
C(6)	21(1)	25(1)	16(1)	2(1)	5(1)	1(1)
C(7)	28(1)	20(1)	16(1)	3(1)	8(1)	2(1)
C(8)	22(1)	37(1)	24(1)	0(1)	7(1)	3(1)
C(9)	20(1)	26(1)	18(1)	2(1)	3(1)	4(1)
C(10)	39(1)	60(1)	32(1)	-11(1)	-6(1)	30(1)
C(11)	40(1)	40(1)	87(2)	16(1)	-31(1)	-14(1)
C(12)	30(1)	124(2)	16(1)	6(1)	2(1)	11(1)
C(13)	19(1)	25(1)	21(1)	0(1)	4(1)	4(1)
C(14)	37(1)	22(1)	71(1)	5(1)	14(1)	8(1)
C(15)	39(1)	53(1)	23(1)	-4(1)	10(1)	18(1)
C(16)	20(1)	59(1)	28(1)	-3(1)	2(1)	6(1)
B(1)	41(1)	23(1)	24(1)	2(1)	11(1)	-9(1)

**Table 6.10.** Anisotropic displacement parameters (Å<sup>2</sup> x 10<sup>3</sup>) for **3**. The anisotropic displacement factor exponent takes the form:  $-2\pi^2$  [h<sup>2</sup> a<sup>\*2</sup> U<sup>11</sup> + ... + 2 h k a<sup>\*</sup> b<sup>\*</sup> U<sup>12</sup>]

H(10) $5620$ $7904$ $-1202$ $41$ H(1) $4249$ $9566$ $868$ $23$ H(3) $-370$ $8395$ $593$ $28$ H(4) $534$ $7118$ $-212$ $34$ H(5) $3308$ $7030$ $-439$ $31$ H(7A) $254$ $10207$ $969$ $25$ H(7B) $2136$ $10443$ $1400$ $25$ H(8A) $6497$ $8758$ $389$ $33$ H(8B) $6330$ $7622$ $229$ $33$ H(10A) $4177$ $9153$ $2412$ $68$ H(10B) $3428$ $8398$ $3045$ $68$ H(10C) $5000$ $9047$ $3450$ $68$ H(11A) $4720$ $10761$ $3631$ $90$ H(11B) $2931$ $11273$ $3469$ $90$ H(12A) $1880$ $9004$ $4262$ $85$ H(12B) $1901$ $10136$ $4462$ $85$ H(12A) $1880$ $9034$ $4262$ $85$ H(14A) $-1017$ $12153$ $2417$ $64$ H(14A) $-276$ $11681$ $1844$ $64$ H(14A) $-2711$ $10854$ $4021$ $56$ H(15A) $-2714$ $11114$ $3616$ $56$ H(16B) $-2091$ $10492$ $1343$		Х	У	Z	U(eq)
H(1)4249956686823H(3).370.8395.593.28H(4).534.7118.212.34H(5).3308.7030.439.31H(7A).254.10207.969.25H(7B).2136.10443.400.25H(8A).6497.8758.389.33H(8B).6330.7622.229.33H(10A).4177.9153.2412.68H(10B).3428.8398.3045.68H(10C).5000.9047.3450.68H(11A).4720.10761.3631.90H(11B).2931.11273.3469.90H(12A).1880.9004.4262.85H(12B).9011.0136.4462.85H(12A).1017.2153.2417.64H(14A).1017.2153.2417.64H(14A).1017.2153.2417.64H(14A).1017.2153.2417.64H(14A).1017.2153.2417.64H(14A).276.1681.844.64H(14C).818.1867.2903.64H(15A).211.0854.4021.56H(15A).2104.1114.3616.56H(16A).2775.9842.2085.54H(16B).2091.10492.1343.54H(16B).2091 <td< td=""><td>H(1O)</td><td>5620</td><td>7904</td><td>-1202</td><td>41</td></td<>	H(1O)	5620	7904	-1202	41
H(3)-370839559328H(4)5347118-21234H(5)33087030-43931H(7A)2541020796925H(7B)213610443140025H(8A)6497875838933H(8B)6330762222933H(10A)41779153241268H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(11C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12B)190110136446285H(14A)-101712153241764H(14A)-101712153241764H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15A)-21110854402156H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540	H(1)	4249	9566	868	23
H(4)5347118-21234H(5)33087030-43931H(7A)2541020796925H(7B)213610443140025H(8A)6497875838933H(8B)6330762222933H(10A)41779153241268H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(12A)18809004426285H(12B)190110136446285H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43	H(3)	-370	8395	593	28
H(5)33087030-43931H(7A)2541020796925H(7B)213610443140025H(8A)6497875838933H(8B)6330762222933H(10A)41779153241268H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(12A)18809004426285H(12B)190110136446285H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43	H(4)	534	7118	-212	34
H(7A)2541020796925H(7B)213610443140025H(8A)6497875838933H(8B)6330762222933H(10A)41779153241268H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14A)-101712153241764H(14A)-27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43	H(5)	3308	7030	-439	31
H(7B)213610443140025H(8A)6497875838933H(8B)6330762222933H(10A)41779153241268H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(12C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16B)-209110492134354H(16B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(7A)	254	10207	969	25
H(8A)6497875838933H(8B)6330762222933H(10A)41779153241268H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(12C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(16B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43	H(7B)	2136	10443	1400	25
H(8B)6330762222933H(10A)41779153241268H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(11C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12B)190110136446285H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43	H(8A)	6497	8758	389	33
H(10A)41779153241268H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(11C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12B)190110136446285H(14A)-101712153241764H(14B)27611681184464H(14B)27611681184464H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(16C)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(8B)	6330	7622	229	33
H(10B)34288398304568H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(11C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(10A)	4177	9153	2412	68
H(10C)50009047345068H(11A)472010761363190H(11B)293111273346990H(11C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8382(14)2095(13)43	H(10B)	3428	8398	3045	68
H(11A)472010761363190H(11B)293111273346990H(11C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16B)-209110971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(10C)	5000	9047	3450	68
H(11B)293111273346990H(11C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16B)-209110971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(11A)	4720	10761	3631	90
H(11C)375710923262690H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8382(14)3095(13)43	H(11B)	2931	11273	3469	90
H(12A)18809004426285H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(11C)	3757	10923	2626	90
H(12B)190110136446285H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(12A)	1880	9004	4262	85
H(12C)36049534463585H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(12B)	1901	10136	4462	85
H(14A)-101712153241764H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(12C)	3604	9534	4635	85
H(14B)27611681184464H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43	H(14A)	-1017	12153	2417	64
H(14C)81811867290364H(15A)-21110854402156H(15B)-153210012372356H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(14B)	276	11681	1844	64
H(15A)-21110854402156H(15B)-153210012372356H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(14C)	818	11867	2903	64
H(15B)-153210012372356H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(15A)	-211	10854	4021	56
H(15C)-210411114361656H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(15B)	-1532	10012	3723	56
H(16A)-27759842208554H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(15C)	-2104	11114	3616	56
H(16B)-209110492134354H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(16A)	-2775	9842	2085	54
H(16C)-317610971202654H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(16B)	-2091	10492	1343	54
H(1B)700(20)7900(15)2542(12)43H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(16C)	-3176	10971	2026	54
H(2B)-540(20)8455(14)3318(13)43H(3B)-1380(20)8382(14)2095(13)43	H(1B)	700(20)	7900(15)	2542(12)	43
H(3B) -1380(20) 8382(14) 2095(13) 43	H(2B)	-540(20)	8455(14)	3318(13)	43
	H(3B)	-1380(20)	8382(14)	2095(13)	43

**Table 6.11.** Hydrogen coordinates  $(x10^4)$  and isotropic displacement parameters  $(\text{\AA}^2 x 10^3)$  for **3**.

C(6)-C(1)-C(2)-C(3)	-2.46(16)	C(13)-P(1)-C(9)-C(10)	-174.07(10)
C(6)-C(1)-C(2)-C(7)	-179.82(10)	B(1)-P(1)-C(9)-C(10)	62.26(12)
C(1)-C(2)-C(3)-C(4)	1.68(17)	C(7)-P(1)-C(9)-C(11)	57.99(14)
C(7)-C(2)-C(3)-C(4)	179.06(11)	C(13)-P(1)-C(9)-C(11)	-53.90(15)
C(2)-C(3)-C(4)-C(5)	0.3(2)	B(1)-P(1)-C(9)-C(11)	-177.56(13)
C(3)-C(4)-C(5)-C(6)	-1.6(2)	C(7)-P(1)-C(9)-C(12)	-179.23(12)
C(4)-C(5)-C(6)-C(1)	0.87(18)	C(13)-P(1)-C(9)-C(12)	68.88(13)
C(4)-C(5)-C(6)-C(8)	-178.91(12)	B(1)-P(1)-C(9)-C(12)	-54.78(13)
C(2)-C(1)-C(6)-C(5)	1.19(17)	C(7)-P(1)-C(13)-C(15)	-174.70(10)
C(2)-C(1)-C(6)-C(8)	-179.03(10)	C(9)-P(1)-C(13)-C(15)	-60.75(11)
C(3)-C(2)-C(7)-P(1)	70.84(13)	B(1)-P(1)-C(13)-C(15)	62.80(11)
C(1)-C(2)-C(7)-P(1)	-111.85(11)	C(7)-P(1)-C(13)-C(14)	-51.01(12)
C(9)-P(1)-C(7)-C(2)	85.78(10)	C(9)-P(1)-C(13)-C(14)	62.94(12)
C(13)-P(1)-C(7)-C(2)	-155.52(9)	B(1)-P(1)-C(13)-C(14)	-173.51(11)
B(1)-P(1)-C(7)-C(2)	-35.89(12)	C(7)-P(1)-C(13)-C(16)	68.10(10)
C(5)-C(6)-C(8)-O(1)	69.06(16)	C(9)-P(1)-C(13)-C(16)	-177.95(9)
C(1)-C(6)-C(8)-O(1)	-110.71(13)	B(1)-P(1)-C(13)-C(16)	-54.40(11)
C(7)-P(1)-C(9)-C(10)	-62.18(12)		

Table 6.12.Torsion angles (°) for 3.





 Table 6.13. Crystal data and structure refinement for 4.

Identification code	dyw1149	dyw1149		
Empirical formula	C16 H29 B Br P			
Formula weight	343.08			
Temperature	100(2) K			
Wavelength	0.71073 Å			
Crystal system	Triclinic			
Space group	P-1			
Unit cell dimensions	a = 7.9157(6) Å	$\alpha = 81.703(1)^{\circ}.$		
	b = 8.5216(7) Å	$\beta = 79.170(1)^{\circ}$ .		
	c = 14.9218(12)  Å	$\gamma = 62.750(1)^{\circ}.$		
Volume	876.94(12) Å <sup>3</sup>			
Z	2			
Density (calculated)	1.299 Mg/m <sup>3</sup>			
Absorption coefficient	2.422 mm <sup>-1</sup>			
F(000)	360			
Crystal size	0.44 x 0.20 x 0.20 mm <sup>3</sup>			
Theta range for data collection	2.69 to 30.03°.			
Index ranges	-11<=h<=11, -12<=k<=	11, -20<=1<=21		
Reflections collected	10125			
Independent reflections	5040 [R(int) = 0.0193]			
Completeness to theta = $30.03^{\circ}$	98.6 %			
Absorption correction	Semi-empirical from equ	iivalents		
Max. and min. transmission	0.444 and 0.333			
Refinement method	Full-matrix least-squares	on F <sup>2</sup>		
Data / restraints / parameters	5040 / 0 / 288			
Goodness-of-fit on F <sup>2</sup>	1.004			
Final R indices [I>2sigma(I)]	R1 = 0.0314, wR2 = 0.07	790		
R indices (all data)	R1 = 0.0360, wR2 = 0.03	816		
Largest diff. peak and hole	0.929 and -0.253 e.Å <sup>-3</sup>	0.929 and -0.253 e.Å <sup>-3</sup>		

**Table 6.14.** Atomic coordinates  $(x10^4)$  and equivalent isotropic displacement parameters  $(\text{\AA}^2 x \ 10^3)$  for **4**. U(eq) is defined as one third of the trace of the orthogonalized U<sup>ij</sup> tensor.

	Х	У	Ζ	U(eq)
Br(1)	-3676(1)	3992(1)	-6348(1)	23(1)
P(1)	-128(1)	2694(1)	-2140(1)	13(1)
B(1)	-1213(3)	1089(3)	-2190(1)	20(1)
C(1)	576(2)	3058(2)	-4121(1)	14(1)
C(2)	-865(2)	2857(2)	-4432(1)	15(1)
C(3)	-564(2)	2170(2)	-5269(1)	15(1)
C(4)	1214(2)	1666(2)	-5815(1)	17(1)
C(5)	2648(2)	1891(2)	-5524(1)	18(1)
C(6)	2331(2)	2593(2)	-4686(1)	16(1)
C(7)	-2137(2)	1942(2)	-5565(1)	18(1)
C(8)	178(2)	3909(2)	-3236(1)	16(1)
C(9)	-1782(2)	4543(2)	-1392(1)	16(1)
C(10)	-3790(2)	5206(2)	-1667(1)	24(1)
C(11)	-1863(3)	3850(2)	-387(1)	22(1)
C(12)	-1289(3)	6110(2)	-1508(1)	21(1)
C(13)	2295(2)	1453(2)	-1759(1)	16(1)
C(14)	3689(2)	433(2)	-2570(1)	23(1)
C(15)	2240(3)	95(2)	-970(1)	23(1)
C(16)	3061(3)	2635(2)	-1471(1)	22(1)

Br(1)-C(7)	1.9744(16)	C(9)-C(11)	1.532(2)
P(1)-C(8)	1.8437(16) C(9)-C(12)		1.534(2)
P(1)-C(13)	1.8699(16)	C(9)-C(10)	1.539(2)
P(1)-C(9)	1.8734(15)	C(10)-H(10A)	0.97(2)
P(1)-B(1)	1.9344(18)	C(10)-H(10B)	0.93(3)
B(1)-H(1A)	1.07(2)	C(10)-H(10C)	0.95(2)
B(1)-H(1B)	1.14(2)	C(11)-H(11A)	0.95(2)
B(1)-H(1C)	1.11(2)	C(11)-H(11B)	0.97(3)
C(1)-C(2)	1.395(2)	C(11)-H(11C)	0.97(3)
C(1)-C(6)	1.397(2)	C(12)-H(12A)	0.97(2)
C(1)-C(8)	1.510(2)	C(12)-H(12B)	0.97(2)
C(2)-C(3)	1.390(2)	C(12)-H(12C)	0.91(3)
C(2)-H(2)	0.94(2)	C(13)-C(16)	1.533(2)
C(3)-C(4)	1.394(2)	C(13)-C(14)	1.537(2)
C(3)-C(7)	1.495(2)	C(13)-C(15)	1.537(2)
C(4)-C(5)	1.386(2)	C(14)-H(14A)	0.96(2)
C(4)-H(4)	0.90(2)	C(14)-H(14B)	0.98(3)
C(5)-C(6)	1.394(2)	C(14)-H(14C)	0.92(2)
C(5)-H(5)	0.92(2)	C(15)-H(15A)	0.99(2)
C(6)-H(6)	0.90(2)	C(15)-H(15B)	0.95(3)
C(7)-H(7A)	0.95(2)	C(15)-H(15C)	0.97(2)
C(7)-H(7B)	0.96(2)	C(16)-H(16A)	0.95(3)
C(8)-H(8A)	0.94(2)	C(16)-H(16B)	0.95(2)
C(8)-H(8B)	0.93(2)	C(16)-H(16C)	0.97(2)
C(8)-P(1)-C(13)	106.80(7)	H(1A)-B(1)-H(1C)	115.7(17)
C(8)-P(1)-C(9)	101.81(7)	H(1B)-B(1)-H(1C)	110.7(17)
C(13)-P(1)-C(9)	111.91(7)	C(2)-C(1)-C(6)	118.17(14)
C(8)-P(1)-B(1)	114.64(8)	C(2)-C(1)-C(8)	120.47(13)
C(13)-P(1)-B(1)	110.17(8)	C(6)-C(1)-C(8)	121.13(13)
C(9)-P(1)-B(1)	111.23(7)	C(3)-C(2)-C(1)	121.48(14)
P(1)-B(1)-H(1A)	107.9(13)	C(3)-C(2)-H(2)	118.9(13)
P(1)-B(1)-H(1B)	104.7(12)	C(1)-C(2)-H(2)	119.6(13)
H(1A)-B(1)-H(1B)	109.7(18)	C(2)-C(3)-C(4)	119.64(14)
P(1)-B(1)-H(1C)	107.5(11)	C(2)-C(3)-C(7)	119.58(14)

Table 6.15. Bond lengths  $(\text{\AA})$  and angles  $(^{\circ})$  for 4.

C(4)-C(3)-C(7)	120.76(14)	C(9)-C(11)-H(11B)	111.5(14)
C(5)-C(4)-C(3)	119.57(14)	H(11A)-C(11)-H(11B)	109.3(19)
C(5)-C(4)-H(4)	117.8(15)	C(9)-C(11)-H(11C)	110.3(15)
C(3)-C(4)-H(4)	122.6(15)	H(11A)-C(11)-H(11C)	107.3(19)
C(4)-C(5)-C(6)	120.44(15)	H(11B)-C(11)-H(11C)	104(2)
C(4)-C(5)-H(5)	120.1(13)	C(9)-C(12)-H(12A)	106.3(14)
C(6)-C(5)-H(5)	119.4(13)	C(9)-C(12)-H(12B)	113.3(14)
C(5)-C(6)-C(1)	120.66(14)	H(12A)-C(12)-H(12B)	107.5(19)
C(5)-C(6)-H(6)	121.1(13)	C(9)-C(12)-H(12C)	112.9(16)
C(1)-C(6)-H(6)	118.2(13)	H(12A)-C(12)-H(12C)	110(2)
C(3)-C(7)-Br(1)	111.92(11)	H(12B)-C(12)-H(12C)	107(2)
C(3)-C(7)-H(7A)	111.1(14)	C(16)-C(13)-C(14)	108.25(14)
Br(1)-C(7)-H(7A)	103.5(14)	C(16)-C(13)-C(15)	109.71(14)
C(3)-C(7)-H(7B)	114.2(14)	C(14)-C(13)-C(15)	107.95(13)
Br(1)-C(7)-H(7B)	100.8(15)	C(16)-C(13)-P(1)	114.07(11)
H(7A)-C(7)-H(7B)	114(2)	C(14)-C(13)-P(1)	107.14(11)
C(1)-C(8)-P(1)	119.51(11)	C(15)-C(13)-P(1)	109.51(11)
C(1)-C(8)-H(8A)	108.6(14)	C(13)-C(14)-H(14A)	111.8(13)
P(1)-C(8)-H(8A)	101.9(14)	C(13)-C(14)-H(14B)	113.6(16)
C(1)-C(8)-H(8B)	110.6(13)	H(14A)-C(14)-H(14B)	107(2)
P(1)-C(8)-H(8B)	105.0(14)	C(13)-C(14)-H(14C)	108.4(15)
H(8A)-C(8)-H(8B)	110.9(19)	H(14A)-C(14)-H(14C)	110(2)
C(11)-C(9)-C(12)	109.93(13)	H(14B)-C(14)-H(14C)	106(2)
C(11)-C(9)-C(10)	107.77(14)	C(13)-C(15)-H(15A)	109.4(14)
C(12)-C(9)-C(10)	108.22(14)	C(13)-C(15)-H(15B)	107.3(15)
C(11)-C(9)-P(1)	110.31(11)	H(15A)-C(15)-H(15B)	110(2)
C(12)-C(9)-P(1)	114.30(11)	C(13)-C(15)-H(15C)	113.3(14)
C(10)-C(9)-P(1)	106.01(11)	H(15A)-C(15)-H(15C)	110.4(19)
C(9)-C(10)-H(10A)	112.5(15)	H(15B)-C(15)-H(15C)	106(2)
C(9)-C(10)-H(10B)	111.3(15)	C(13)-C(16)-H(16A)	108.3(16)
H(10A)-C(10)-H(10B)	109(2)	C(13)-C(16)-H(16B)	115.1(14)
C(9)-C(10)-H(10C)	108.6(14)	H(16A)-C(16)-H(16B)	110(2)
H(10A)-C(10)-H(10C)	107(2)	C(13)-C(16)-H(16C)	112.6(13)
H(10B)-C(10)-H(10C)	108(2)	H(16A)-C(16)-H(16C)	108(2)
C(9)-C(11)-H(11A)	113.8(12)	H(16B)-C(16)-H(16C)	102.3(19)

	U <sup>11</sup>	U <sup>22</sup>	U <sup>33</sup>	U <sup>23</sup>	U <sup>13</sup>	U <sup>12</sup>
Br(1)	22(1)	24(1)	29(1)	7(1)	-13(1)	-13(1)
P(1)	13(1)	14(1)	13(1)	-2(1)	-1(1)	-7(1)
B(1)	23(1)	23(1)	19(1)	-4(1)	0(1)	-16(1)
C(1)	16(1)	14(1)	14(1)	0(1)	-2(1)	-7(1)
C(2)	13(1)	15(1)	15(1)	0(1)	-2(1)	-6(1)
C(3)	18(1)	14(1)	15(1)	2(1)	-6(1)	-8(1)
C(4)	21(1)	16(1)	13(1)	-1(1)	-2(1)	-8(1)
C(5)	17(1)	19(1)	17(1)	-1(1)	0(1)	-8(1)
C(6)	16(1)	20(1)	16(1)	0(1)	-2(1)	-10(1)
C(7)	22(1)	19(1)	18(1)	3(1)	-7(1)	-12(1)
C(8)	18(1)	16(1)	14(1)	-1(1)	-2(1)	-9(1)
C(9)	16(1)	16(1)	15(1)	-4(1)	-1(1)	-6(1)
C(10)	15(1)	25(1)	28(1)	-8(1)	-2(1)	-4(1)
C(11)	26(1)	24(1)	15(1)	-4(1)	0(1)	-10(1)
C(12)	24(1)	17(1)	22(1)	-4(1)	-3(1)	-9(1)
C(13)	15(1)	16(1)	16(1)	-1(1)	-2(1)	-5(1)
C(14)	17(1)	23(1)	21(1)	-3(1)	0(1)	-3(1)
C(15)	24(1)	21(1)	20(1)	4(1)	-4(1)	-7(1)
C(16)	19(1)	26(1)	25(1)	-1(1)	-6(1)	-11(1)

**Table 6.16.** Anisotropic displacement parameters (Å<sup>2</sup> x 10<sup>3</sup>) for **4**. The anisotropic displacement factor exponent takes the form:  $-2\pi^2$  [h<sup>2</sup> a<sup>\*2</sup> U<sup>11</sup> + ... + 2 h k a<sup>\*</sup> b<sup>\*</sup> U<sup>12</sup>]

	Х	У	Z	U(eq)
H(1A)	-2430(30)	1780(30)	-2546(16)	30(6)
H(1B)	-1700(30)	780(30)	-1444(17)	34(6)
H(1C)	-50(30)	-120(30)	-2498(15)	24(5)
H(2)	-2070(30)	3210(30)	-4076(15)	21(5)
H(4)	1460(30)	1230(30)	-6364(16)	24(5)
H(5)	3810(30)	1610(30)	-5892(14)	18(5)
H(6)	3250(30)	2760(30)	-4494(14)	15(5)
H(7A)	-3050(30)	1910(30)	-5055(16)	26(6)
H(7B)	-1700(30)	1000(30)	-5965(17)	34(6)
H(8A)	-1010(30)	4920(30)	-3217(15)	26(5)
H(8B)	1150(30)	4200(30)	-3189(15)	21(5)
H(10A)	-4230(30)	4300(30)	-1567(17)	31(6)
H(10B)	-3820(30)	5640(30)	-2273(18)	34(6)
H(10C)	-4680(30)	6150(30)	-1296(17)	33(6)
H(11A)	-710(30)	3520(30)	-137(14)	17(5)
H(11B)	-2180(30)	2860(30)	-296(16)	30(6)
H(11C)	-2900(40)	4750(30)	-13(18)	38(7)
H(12A)	-2260(30)	6980(30)	-1102(16)	29(6)
H(12B)	-1330(30)	6670(30)	-2121(16)	29(6)
H(12C)	-100(40)	5810(30)	-1362(17)	32(6)
H(14A)	3880(30)	1210(30)	-3064(16)	25(5)
H(14B)	3260(40)	-310(40)	-2823(19)	41(7)
H(14C)	4840(40)	-320(30)	-2371(16)	31(6)
H(15A)	3540(30)	-610(30)	-795(16)	30(6)
H(15B)	1830(30)	-650(30)	-1184(16)	31(6)
H(15C)	1330(30)	630(30)	-443(17)	30(6)
H(16A)	4360(40)	1930(30)	-1389(17)	36(7)
H(16B)	2370(30)	3250(30)	-938(16)	23(5)
H(16C)	2990(30)	3600(30)	-1926(16)	25(5)

**Table 6.17.** Hydrogen coordinates  $(x10^4)$  and isotropic displacement parameters  $(\text{\AA}^2 x 10^3)$  for **4**.

C(6)-C(1)-C(2)-C(3)	1.7(2)	C(13)-P(1)-C(9)-C(11)	-53.19(13)
C(8)-C(1)-C(2)-C(3)	176.27(14)	B(1)-P(1)-C(9)-C(11)	70.52(13)
C(1)-C(2)-C(3)-C(4)	0.0(2)	C(8)-P(1)-C(9)-C(12)	-42.44(13)
C(1)-C(2)-C(3)-C(7)	178.64(14)	C(13)-P(1)-C(9)-C(12)	71.29(13)
C(2)-C(3)-C(4)-C(5)	-1.4(2)	B(1)-P(1)-C(9)-C(12)	-165.00(12)
C(7)-C(3)-C(4)-C(5)	-179.98(15)	C(8)-P(1)-C(9)-C(10)	76.67(12)
C(3)-C(4)-C(5)-C(6)	1.0(2)	C(13)-P(1)-C(9)-C(10)	-169.60(11)
C(4)-C(5)-C(6)-C(1)	0.8(2)	B(1)-P(1)-C(9)-C(10)	-45.89(14)
C(2)-C(1)-C(6)-C(5)	-2.2(2)	C(8)-P(1)-C(13)-C(16)	69.20(13)
C(8)-C(1)-C(6)-C(5)	-176.67(14)	C(9)-P(1)-C(13)-C(16)	-41.41(14)
C(2)-C(3)-C(7)-Br(1)	94.58(15)	B(1)-P(1)-C(13)-C(16)	-165.71(12)
C(4)-C(3)-C(7)-Br(1)	-86.83(16)	C(8)-P(1)-C(13)-C(14)	-50.58(13)
C(2)-C(1)-C(8)-P(1)	72.03(17)	C(9)-P(1)-C(13)-C(14)	-161.19(11)
C(6)-C(1)-C(8)-P(1)	-113.59(15)	B(1)-P(1)-C(13)-C(14)	74.52(13)
C(13)-P(1)-C(8)-C(1)	91.58(13)	C(8)-P(1)-C(13)-C(15)	-167.42(11)
C(9)-P(1)-C(8)-C(1)	-150.94(12)	C(9)-P(1)-C(13)-C(15)	81.98(12)
B(1)-P(1)-C(8)-C(1)	-30.75(15)	B(1)-P(1)-C(13)-C(15)	-42.32(13
C(8)-P(1)-C(9)-C(11)	-166.91(11)		

Table 6.18.Torsion angles (°) for 4.

#### References

- (1) Hatnean, J. A.; Beck, R.; Borrelli, J. D.; Johnson, S. A. Organometallics 2010, 29, 6077.
- (2) Nguyen, D. H.; Modrego, F. J.; Cetina-Casas, J. M.; Gómez-Bautista, D.; Jiménez, M. V.; Castarlenas, R.; Lahoz, F. J.; Oro, L. A.; Pérez-Torrente, J. J. Organometallics 2012, 31, 6395.
- (3) Besora, M.; Gourlaouen, C.; Yates, B.; Maseras, F. *Dalton Trans.* **2011**, *40*, 11089.
- (4) Hostetler, M. J.; Bergman, R. G. J. Am. Chem. Soc. 1992, 114, 787.
- (5) Wilson, M. R.; Liu, H.; Prock, A.; Giering, W. P. *Organometallics* **1993**, *12*, 2044.
- (6) Shaw, B. L.; Stainbank, R. E. *Journal of the Chemical Society, Dalton Transactions* **1972**, 223.
- (7) Ugo, R.; Pasini, A.; Fusi, A.; Cenini, S. J. Amer. Chem. Soc. 1972, 94, 7364.
- (8) Niu, S.; Hall, M. B. Chem. Rev. 2000, 100, 353.
- (9) Collman, J. P.; Sears, C. T., Jr. Inorg. Chem. 1968, 7, 27.
- (10) Chatt, J.; Butter, S. A. Chem. Commun. 1967, 501.
- (11) Johnson, C. E.; Fisher, B. J.; Eisenberg, R. J. Am. Chem. Soc. 1983, 105, 7772.
- (12) Johnson, C. E.; Eisenberg, R. J. Am. Chem. Soc. 1985, 107, 3148.
- (13) Hartwig, J. F. Organotransition Metal Chemistry: From Bonding to Catalysis; University Science Books: Sausalito, CA, 2010.
- (14) Matsubara, T.; Maseras, F.; Koga, N.; Morokuma, K. J. Phys. Chem. **1996**, 100, 2573.
- (15) Safarowic, F. J.; Bierdeman, D. J.; Keister, J. B. J. Am. Chem. Soc. **1996**, 118, 11805.
- Kundu, S.; Choliy, Y.; Zhuo, G.; Ahuja, R.; Emge, T. J.; Warmuth, R.; Brookhart, M.; Krogh-Jespersen, K.; Goldman, A. S. *Organometallics* 2009, 28, 5432.
- (17) Naghipour, A.; Sabounchei, S. J.; Morales-Morales, D.; Hernández-Ortega, S.; Jensen, C. M. J. Organomet. Chem. **2004**, 689, 2494.
- (18) Brunel, J. M.; Faure, B.; Maffei, M. Coord. Chem. Rev. 1998, 178–180, 665.
- (19) Higham, L. J.; Heslop, K.; Pringle, P. G.; Orpen, A. G. J. Organomet. Chem. **2004**, 689, 2975.
- (20) Blessent, M. Rutgers, The State University of New Jersey, New Brunswick, NJ. Unpublished work, 2012.
- (21) Rybtchinski, B.; Ben-David, Y.; Milstein, D. Organometallics 1997, 16, 3786.
- (22) Still, W. C.; Kahn, M.; Mitra, A. J. Org. Chem. 1978, 43, 2923.
- (23) Bruker-AXS. SADABS, Bruker area detector scaling and absorption correction, v2.05, Bruker-AXS Inc., Madison, Wisconsin, 2003; SAINTplus, Bruker area detector data reduction program, v6.45, Bruker-AXS Inc., Madison, Wisconsin, 2003.
- (24) Sheldrick, G. M. SHELX86, Program for the Solution of Crystal Structures, University of Göttingen, Germany, 1986.
- (25) Sheldrick, G. M. S., Program for Crystal Structure Refinement, University of Göttingen, Germany, 1997.
- (26) Lee, D. W.; Jensen, C. M.; Morales-Morales, D. Organometallics 2003, 22, 4744.

## Chapter 7

# Addition of Tropone to (<sup>tBu</sup>PCP)IrH<sub>2</sub>: Formation of a Dimerization Product

#### Abstract

Adducts of tropone with (PCP)IrH<sub>2</sub> (PCP =  $\kappa^3$ -C<sub>6</sub>H<sub>3</sub>-2,6-(CH<sub>2</sub>P<sup>t</sup>Bu<sub>2</sub>)<sub>2</sub>) have been observed. The thermodynamic and kinetic products have been synthesized and characterized. The thermodynamic product was previously found to catalyze the dimerization of tropone to give a fused tricyclic dihydrodicycloheptafuranol. The formation of this tropone dimer was investigated. A possible mechanism for the dimerization reaction is suggested in which tropone, normally an electrophile, acts as a nucleophile upon activation by C-H oxidative addition to the iridium(I) center.

#### 7.1 Introduction

 $\alpha,\beta$ -unsaturated ketones are of great utility in a wide variety of organic transformations, arising from the ability to activate and functionalize the carbonyl moiety and positions  $\alpha$ ,  $\beta$ , and  $\gamma$  to the carbonyl group. Classically,  $\alpha,\beta$ -unsaturated ketones can be prepared via aldol condensation,<sup>1</sup> allylic oxidation,<sup>2-4</sup> and elimination reactions from  $\alpha$ -substituted ketones.<sup>5</sup> These methods often require several steps or may be complicated by low yields and unwanted byproducts. A simple one-step reaction to introduce the double bond into an aliphatic molecular framework would be a desirable synthetic tool.

Direct dehydrogenation of ketones has been one strategy to approach  $\alpha$ , $\beta$ unsaturated ketones. Dehydrogenation of ketone substrates by palladium complexes in both stoichiometric and catalytic amounts have been reported, although efforts at catalytic reactions have been hampered by low conversions and yields.<sup>6-9</sup> The use of substituted iodoxybenzoic acids with Oxone has been reported to effect the formation of  $\alpha$ , $\beta$ -unsaturated ketones from alcohols, presumably via initial alcohol oxidation and subsequent ketone dehydrogenation; however, the formation of  $\alpha$ , $\beta$ -unsaturated ketones by this method is limited to cyclic substrates.<sup>10</sup>

Pincer-ligated iridium complexes, specifically derivatives of "(PCP)Ir" (PCP =  $\kappa^3$ -C<sub>6</sub>H<sub>3</sub>-2,6-(CH<sub>2</sub>PR<sub>2</sub>)<sub>2</sub>; R = t-Bu in this chapter) (1), are the most successful class of catalysts for the dehydrogenation of aliphatic linkages reported to date.<sup>11,12</sup> In addition to the dehydrogenation of alkanes, the catalytic dehydrogenation of THF<sup>13</sup> and amines<sup>14</sup> has been reported with selectivity for the  $\alpha$ , $\beta$ -position. Work performed by Xiawei Zhang, a previous graduate student in our group, had shown that ketone dehydrogenation occurs readily but in general the resulting enones, or products of further dehydrogenation, tend

to strongly inhibit catalysis.<sup>15,16</sup> In the case of a simple acyclic enone, a conjugated metallacyclic species is formed which is remarkably stable. In the particular case of cycloheptanone, a very unusual conjugated [5.2.0] metallabicyclic species is formed, apparently resulting from the triple dehydrogenation (transfer of 6 H atoms) of the ring to give tropone, followed by C-H addition and O-coordination. Accordingly this complex can be generated directly by the reaction of "(PCP)Ir" with tropone. Remarkably, this species acts as a catalyst for the dimerization of tropone to give a fused tricyclic dihydrodicycloheptafuranol, a reaction which to our knowledge is without reported precedent.

#### 7.2 Previous Investigations: Reactions of Ketones with (<sup>tBu</sup>PCP)Ir

Reactions of complex **1** (17 mM) with a hydrogen acceptor (*tert*-butylethylene or norbornene) and various ketones were investigated by Xiawei Zhang.<sup>15,16</sup> Use of acyclic ketone 3-pentanone resulted in the formation of metallacyclic complexes **2** and **3**, whose stability presumably prevents catalytic turnover (Scheme 7.1). Use of cyclic ketones cyclohexanone and 3-methylcyclohexanone resulted in the formation of aryloxy hydride complexes **4** and **6** through successive dehydrogenation events (Scheme 7.2). Catalytic turnover was inhibited due to the formation of the stable aryloxy hydride.







Scheme 7.2. Reaction of cyclohexanone and 3-methylcyclohexanone with 1.<sup>15</sup>

Cycloheptanone was next investigated as a ketone substrate.<sup>15,16</sup> It was hypothesized that the possible dehydrogenation products of cycloheptanone should not form alcohols, and, as a result, cycloheptanone might undergo catalytic dehydrogenation. Treatment of complex **1** (17 mM) with cycloheptanone (51 mM) and NBE (170 mM) at 120 °C for 12 hours and subsequent recrystallization yielded an orange solid. The product was identified as complex **7** through NMR spectroscopy and X-ray crystallography (Scheme 7.3). Addition of CO to the resulting solution yielded the  $\eta^1$ -troponyl iridium CO complex **9**, and dark amber-colored crystals were crystallized from the solution. Surprisingly, the expected CO adduct **9** was found to have co-crystallized with organic compound **10** (Figure 7.1). Compound **10** is a previously unreported dimerization product of tropone.<sup>17</sup> My investigations sought to characterize complexes **7** and **8** and to investigate the formation of novel tropone dimer **10**.

Scheme 7.3. Addition of cycloheptanone to 1.



Figure 7.1. Crystal structures of CO adduct 9 and tropone dimer 10.<sup>15</sup>



#### 7.3 Results and Discussion

Reaction of complex **1** (26 mM) with NBE (39 mM) and a small excess of tropone (32 mM) at room temperature yielded the kinetic product **8** (Scheme 7.3), which was characterized by NMR spectroscopy. Complex **8** was found to be much less stable than the analogous kinetic C-H adduct **2**; it isomerizes to complex **7** within one hour at room temperature. This difference in stability may be attributable to the greater strain in the 4-membered metallacyclic ring as compared to the 5-membered metallacyclic unit in **2**. In analogy to the reaction of pentanone (Scheme 7.1), initial formation of **8**, despite

the greater thermodynamic stability of 7, suggests that coordination of the carbonyl moiety to the iridium atom does not precede C-H bond activation.

When excess tropone was reacted with complex 1 and norbornene, a dark-colored solution was formed in contrast to the orange-colored solution that was obtained from the reaction with only one equivalent of tropone. Upon further investigation, when a large excess of tropone (26 equivalents) was reacted with complex 1 and a small excess of norbornene, tropone was consumed completely to form primarily compound 10. Addition of tropone to a solution containing a catalytic amount of complex 7 also yielded tropone dimer 10, indicating that the tropone adduct 7 is a probable intermediate in the dimerization mechanism. A control experiment in which tropone and norbornene was heated to 120 °C in the absence of complex 1 resulted in observation of exclusively starting materials by gas chromatography and NMR spectroscopy.

A possible mechanism for dimerization is given in Scheme 7.4. Complex 7 has a resonance form that suggests that C32 (Figure 7.1) may be nucleophilic. Nucleophilic attack by complex 7 on free tropone in a 1,8-fashion yields intermediate **11** after proton transfer. The ability of tropone to undergo nucleophilic 1,8-addition reactions is well precedented,<sup>18-20</sup> but to our knowledge, no examples of tropone serving as a nucleophile have been reported in the literature. A formal 1,7-hydride shift would yield iridium alkyl hydride intermediate **12**. Reductive elimination and hemiketal formation would then afford dimerization product **10**. In this reaction mechanism, reductive elimination of the alkyl hydride intermediate **12** should be facile because the metallacycle lacks resonance stabilization.



Scheme 7.4. Possible mechanism for tropone dimerization.

An attempt to observe nucleophilic behavior of complex 7 was performed by the addition of excess 1-iodopropane to a solution of complex 7. While no reaction was observed after 3 h at room temperature, heating to 110 °C resulted in the disappearance of complex 7 by  $^{31}$ P NMR spectroscopy and the appearance of signals consistent with propyl iodide complex  $13^{21}$  and hydrido iodide complex  $14^{.22}$  GC analysis of the reaction solution following the reaction revealed only the presence of tropone. No evidence of the propyl adduct of tropone was observed. A possible mechanism for the observed results is given in Scheme 7.5.





#### 7.4 Summary

The adducts of the reaction of tropone with complex **1**, complexes **7** and **8**, have been synthesized and characterized. The kinetic product, complex **8**, isomerizes to the thermodynamic product, complex **7**, at room temperature. The formation of the novel tropone dimer **10** appears to be catalyzed by complex **7**. A possible mechanism for the dimerization reaction is suggested in which tropone, normally an electrophile, acts as a nucleophile upon activation by C-H oxidative addition to the iridium(I) center. The concept of utilizing C-H addition of a molecule to a transition metal complex to induce nucleophilic character at a remote site of a substrate molecule is a possible area for further investigation.

#### 7.5 Experimental

All reactions, recrystallizations, and routine manipulations were conducted under argon using an argon-filled glove box or by using standard Schlenk techniques. Reagent grade solvents were used and dried according to established methods, then degassed with argon. All NMR solvents were dried, vacuum-transferred, and stored in an argon-filled glove box. <sup>1</sup>H, <sup>31</sup>P{<sup>1</sup>H}, and <sup>13</sup>C{<sup>1</sup>H} NMR spectra were obtained on 300-MHz or 400-MHz Varian Spectrometers. <sup>1</sup>H and <sup>13</sup>C NMR chemical shifts are reported in ppm downfield from tetramethylsilane and were referenced to residual protiated (<sup>1</sup>H) or deuterated solvent (<sup>13</sup>C). <sup>31</sup>P{<sup>1</sup>H} NMR chemical shifts are referenced to an external standard, Me<sub>3</sub>P in mesitylene- $d_{12}$  solvent ( $\delta$  -62.4 ppm), in a capillary tube. GC-MS analyses were carried out with a Varian GC-MS (3900 GC and 2100T MS). Complex **1** was prepared as described in the literature.<sup>12</sup>

(PCP)Ir(H)( $\kappa$ -O,C-OC<sub>7</sub>H<sub>5</sub>) isomers, (troponyl hydride complexes 7 and 8): 15 mg of (<sup>tBu</sup>PCP)IrH<sub>2</sub> (0.026 mmol) was dissolved in 1 mL of *p*-xylene solution containing 3.7 mg of norbornene (0.039 mmol) at room temperature. To the resulting solution was added 3.1  $\mu$ L of tropone (0.032 mmol); after stirring for ca. 1 min, the red solution turned orange. The reaction was monitored by NMR. Complex **8** was formed in 95% yield (<sup>31</sup>P{<sup>1</sup>H} NMR). <sup>31</sup>P NMR (162 MHz, *p*-xylene-*d*<sub>10</sub>):  $\delta$  62.54 (d, *J*<sub>PH</sub> = 12.8 Hz ). <sup>1</sup>H NMR (400 MHz, *p*-xylene-*d*<sub>10</sub>): 1.34 (t, *J*<sub>PH</sub> = 6.4 Hz, 18H, C(CH<sub>3</sub>)<sub>3</sub>), 0.90 (t, *J*<sub>PH</sub> = 6.2 Hz, 18 H, C(CH<sub>3</sub>)<sub>3</sub>), -10.19 (t, *J*<sub>PH</sub> = 18.2 Hz, 1H, Ir-*H*).

Complex **8** isomerized to 7 at room temperature in about one hour. Complex 7:  ${}^{31}P{}^{1}H$ NMR (121.4 MHz, benzene- $d_6$ ):  $\delta$  58.68 (d,  $J_{PH} = 14.0$  Hz).  ${}^{1}H$  NMR (300 MHz, benzene- $d_6$ ):  $\delta$  8.54 (d,  $J_{HH} = 8.4$  Hz, 1H, tropone CH), 7.26 (d,  $J_{HH} = 7.2$  Hz, 2H, PCP, m-H), 7.18 (t,  $J_{HH} = 7.2$  Hz, 1H, PCP, p-H), 6.73–6.40 (m, 4H, tropone C<sub>4</sub>H<sub>4</sub>), 3.60 (d of vt,  $J_{PH} = 3.0$  Hz,  $J_{HH} = 16.8$  Hz, 2H, CH<sub>2</sub>), 3.25 (d of vt,  $J_{PH} = 4.2$  Hz,  $J_{HH} = 16.8$  Hz, 2H, CH<sub>2</sub>), 1.15 (t,  $J_{PH} = 6.4$  Hz, 18H, C(CH<sub>3</sub>)<sub>3</sub>), 1.11 (t,  $J_{PH} = 6.3$  Hz, 18 H, C(CH<sub>3</sub>)<sub>3</sub>), -29.53 (t,  $J_{PH} = 14.4$  Hz, 1H, Ir-H).

5,5-dihydrodicyclohepta[b,d]furan-5a-ol, (10): 15 mg of (PCP)IrH<sub>2</sub> (0.026 mmol), 3.1 µL of tropone (0.032 mmol), and 6.2 mg of norbornene (0.065 mmol) was dissolved in 0.8 mL of p-xylene- $d_{10}$  to yield complex 7 after 3 hours at room temperature. To this reaction mixture, additional tropone (100  $\mu$ L, 1.00 mmol) was added, and the reaction mixture was heated at 80 °C. After 11 hours, no further change in the tropone concentration was observed by NMR spectroscopy. Column chromatography (20:1 $\rightarrow$ 1:1 hexanes:ethyl acetate) afforded the product as a dark red oil (35.6 mg, 32 % isolated yield). Addition of excess D<sub>2</sub>O (25 µL, 1.4 mmol) to a solution of 10 (35.6 mg, 0.168 mmol) in methylene chloride- $d_2$  (0.8 mL) resulted in the disappearance of the singlet at  $\delta$ 9.69 ppm in the <sup>1</sup>H NMR spectrum. <sup>1</sup>H NMR (400 MHz, methylene chloride- $d_2$ ):  $\delta$  9.69 (s, 1H), 7.71 (dd, J = 8.6 Hz, 1.2 Hz, 1H), 7.30 (ddd, J = 12.0 Hz, 7.6 Hz, 1.7 Hz, 1H), 7.23-7.10 (m, 5H), 6.85 (dd, J = 8.1 Hz, 1.2 Hz, 1H), 6.82 (td, J = 7.4 Hz, 1.3 Hz, 1H),3.89 (s, 2H). <sup>13</sup>C NMR (100 MHz, methylene chloride- $d_2$ ):  $\delta$  188.64, 155.91, 154.52, 141.06, 138.72, 138.18, 135.61, 135.00, 130.91, 128.87, 125.80, 120.54, 118.14, 37.18. GC-MS (EI) calculated for  $C_{14}H_{10}O^+$  ( $C_{14}H_{12}O_2 - H_2O$ ): 194.07, found: 194.00.

#### **References:**

- (1) *Modern Aldol Reactions*; Mahrwald, R., Ed.; Wiley-VCH: Berlin, 2004.
- (2) Zhao, Y.; Yeung, Y.-Y. Org. Lett. 2010, 12, 2128.
- (3) Shing, T. K. M.; Yeung, Y.-Y.; Su, P. L. Org. Lett. 2006, 8, 3149.
- (4) Catino, A. J.; Forslund, R. E.; Doyle, M. P. J. Am. Chem. Soc. 2004, 126, 13622.
- (5) Reich, H. J.; Renga, J. M.; Reich, I. L. J. Am. Chem. Soc. 1975, 97, 5434.
- (6) For a recent mini review, see: Muzart, J. Eur. J. Org. Chem. 2010, 3779.
- (7) Tokunaga, M.; Harada, S.; Iwasawa, T.; Obora, Y.; Tsuji, Y. *Tetrahedron Lett.* **2007**, *48*, 6860.
- (8) Shvo, Y.; Arisha, A. H. I. J. Org. Chem. 1998, 63, 5640.
- (9) Theissen, R. J. J. Org. Chem. 1971, 36, 752.
- (10) Uyanik, M.; Akakura, M.; Ishihara, K. J. Am. Chem. Soc. 2009, 131, 251.
- (11) For a recent lead reference, see: Punji, B., Emge, T. J., Goldman, A. S. *Organometallics* **2010**, *29*, 2702.
- (12) Gupta, M.; Hagen, C.; Flesher, R. J.; Kaska, W. C.; Jensen, C. M. Chem. Commun. 1996, 2083.
- (13) Gupta, M.; Kaska, W. C.; Jensen, C. M. Chem. Commun. 1997, 461.
- (14) Zhang, X.; Fried, A.; Knapp, S.; Goldman, A. S. Chem. Commun. 2003, 2060.
- (15) Zhang, X.; Wang, D. Y.; Emge, T. J.; Goldman, A. S. *Inorg. Chim. Acta* **2011**, *369*, 253.
- (16) Zhang, X. Activation and Transformation of Strong Bonds by Pincer-Ligated Iridum Complexes. Ph. D Dissertation, Rutgers, The State University of New Jersey, 2005.
- (17) Tropone photodimerization, to give cycloaddition products very different from 9, has been well studied; for examples, see: (a) Mukai, T., Tezuka, T., Akasaki, Y. J. Am. Chem. Soc. 1966, 88, 5025; (b) Kende, A. S. J. Am. Soc. Chem. 1966, 88, 5026; (c) Tezuka, T., Akasaki, Y., Muaki, T. Tet. Lett. 1967, 1397; (d) Reingold, I. D., Kowalski, J. A., Cummings, G. C., Gleiter, R., Lange, H., Lovell, S., Kahr, B., Aflatooni, K, Burrow, P. D., Gallup, G. A. J. Phys. Org. Chem. 2006, 19, 642.
- (18) Nair, V.; Poonoth, M.; Vellalath, S.; Suresh, E.; Thirumalai, R. J. Org. Chem. **2006**, 71, 8964.
- (19) Kumar, K.; Kapur, A.; Ishar, M. P. S. Org. Lett. 2000, 2, 787.
- (20) Rigby, J. H.; Wilson, J. Z. J. Am. Chem. Soc. 1984, 106, 8217.
- (21) For characterization of the analogous (PCP)Ir(Me)(I) complex, see: Ghosh, R.; Emge, T. J.; Krogh-Jespersen, K.; Goldman, A. S. J. Am. Chem. Soc. 2008, 130, 11317.
- (22) For characterization of the analogous (PCP)Ir(H)(Cl) complex, see: Moulton, C. J.; Shaw, B. L. J. Chem. Soc., Dalton Trans. **1976**, 1020.

## Chapter 8

## Synthesis and Characterization of (<sup>tBu</sup>PCP)Ir(Me)(F)

#### Abstract

Carbon-fluorine bonds are the strongest known single bonds to carbon and as a consequence can prove very hard to cleave. Though vinyl and aryl C-F bonds can undergo oxidative addition to transition metal complexes, this reaction has appeared inoperable with aliphatic substrates. In this chapter, I report the addition of methyl fluoride's C(sp<sup>3</sup>)-F bond to an iridium center via the initial, reversible cleavage of a C-H bond. The methyl fluoride addition product was characterized through NMR spectroscopy, LIFDI mass spectrometry, and elemental analysis.

#### 8.1 Introduction

The oxidative addition of single-bonded substrates to transition metal centers and the microscopic reverse, reductive elimination, comprise an especially useful reaction class (Scheme 8.1). The addition and elimination of bonds to carbon are key steps in the majority of organic reactions catalyzed or promoted by metal complexes. In this context, the addition of carbon-halogen bonds, specifically C-Cl, C-Br and C-I, has been studied and exploited extensively over the past five decades.<sup>1-6</sup> The formation and cleavage of carbon-fluorine bonds is of great interest in medicinal and environmental chemistry;<sup>7-14</sup> for example, approximately 20% of pharmaceuticals and 40% of agrochemicals have C-F bonds.<sup>8</sup> Examples of oxidative addition and reductive elimination of C-F bonds. however, are essentially limited to those with any and vinyl carbon atoms.<sup>7,14-17</sup> By contrast, we reported the oxidative addition of  $C(sp^3)$ -F bonds to a transition metal center.<sup>18</sup> Notably, these reactions do not proceed via direct (3-centered) C-F addition, nor by any of the several pathways that have previously been proposed for carbon-halogen oxidative addition,<sup>1-6</sup> but instead the  $C(sp^3)$ -F additions take place via initial oxidative addition of a  $C(sp^3)$ -H bond.<sup>18</sup>

Scheme 8.1. General reaction of oxidative addition and reductive elimination.

$$L_nM + X-Y \xrightarrow[reductive]{} cxidative addition} L_nM \begin{pmatrix} X \\ Y \end{pmatrix}$$

We recently reported that the pincer-ligated iridium complex (PCP)Ir (PCP =  $\kappa^3$ -C<sub>6</sub>H<sub>3</sub>-2,6-[CH<sub>2</sub>P(*t*-Bu)<sub>2</sub>]<sub>2</sub>) undergoes oxidative addition of C(sp<sup>3</sup>)-O bonds.<sup>19</sup> Our

study was limited to aryloxy-carbon bonds (ArO-C) which have homolytic bond dissociation energies less than 65 kcal/mol (e.g.  $63.8 \pm 1$  kcal/mol for CH<sub>3</sub>–OPh<sup>20</sup>). By contrast, fluorine forms the strongest known bonds to sp<sup>3</sup>-hybridized-carbon; for example the H<sub>3</sub>C-F bond dissociation energy is 110±1 kcal/mol.<sup>20</sup> Nonetheless, because we found that more electronegative OR groups (e.g. R = 3,5-(CF<sub>3</sub>)<sub>2</sub>C<sub>6</sub>H<sub>3</sub>) favored C–OR addition to (PCP)Ir,<sup>19</sup> we considered that C-F addition to the same fragment might be possible as well. As part of the study reported in reference 18, my investigations focused on the synthesis and characterization of the methyl fluoride oxidative addition product.

#### 8.2 Results and Discussion

(PCP)Ir(NBE) (1; NBE = norbornene) acts as a precursor to the active fragment (PCP)Ir.<sup>21</sup> When fluoromethane (1 atm) was added to a *p*-xylene- $d_{10}$  solution of 1, <sup>31</sup>P NMR spectroscopy revealed complete conversion to a single major product (2;  $\delta$  50.4 ppm, doublet, <sup>2</sup> $J_{FP} = 10.5$  Hz; yield by NMR spectroscopy > 95 %) within 20 min at 50 °C (Scheme 8.2). In benzene- $d_6$  solvent, product 2 is characterized by a triplet at  $\delta$  – 28.20 ppm (<sup>2</sup> $J_{PC} = 4.3$  Hz) in the <sup>13</sup>C NMR spectrum and a signal at  $\delta$  1.85 ppm (Ir-*CH*<sub>3</sub>) in the <sup>1</sup>H NMR spectrum, which appears as a broad singlet at room temperature and as a doublet of triplets at 80 °C (<sup>3</sup> $J_{PH} = 4.7$  Hz, <sup>3</sup> $J_{FH} = 3.5$  Hz). When <sup>31</sup>P-decoupling is applied, this <sup>1</sup>H NMR signal appears as a doublet (<sup>3</sup> $J_{FH} = 3.5$  Hz) (Figure 8.1). These <sup>1</sup>H and <sup>13</sup>C NMR signals are characteristic of a methyl group coordinated to iridium at the apical site of a square pyramidal structure.<sup>19,22</sup> Accordingly, the <sup>1</sup>H NMR spectrum indicates a PCP ligand with only C<sub>s</sub> symmetry, while 2D NOESY spectra show that the putative methyl ligand interacts with only one set of phosphino-*t*-butyl groups.<sup>18</sup>

Scheme 8.2. Oxidative addition of CH<sub>3</sub>F to complex 1.



The <sup>19</sup>F NMR spectrum shows a broad signal at  $\delta$  -252.6 ppm (peak width at halfheight,  $W_{1/2} = 28.6$  Hz) at 25 °C; both the broadness and chemical shift are consistent with that of other iridium fluoride complexes, in particular coordinatively unsaturated complexes.<sup>23,24</sup> While the width of the <sup>19</sup>F NMR signal precluded direct observation of the <sup>31</sup>P-<sup>19</sup>F coupling that was observed in the <sup>31</sup>P NMR spectrum (<sup>2</sup>*J*<sub>FP</sub> = 10.5 Hz), application of <sup>31</sup>P-decoupling (<sup>19</sup>F{<sup>31</sup>P} NMR) resulted in a decrease in W<sub>1/2</sub> from 28.6 Hz to 13.9 Hz (Figure 8.2), confirming the existence of <sup>31</sup>P-<sup>19</sup>F coupling. Simulation of the <sup>19</sup>F NMR signal using Spinworks software successfully reproduced it (with and without <sup>31</sup>P-decoupling) based on coupling to two <sup>31</sup>P nuclei with <sup>2</sup>*J*<sub>PF</sub> = 10.5 Hz, coupling to three <sup>1</sup>H nuclei with <sup>3</sup>*J*<sub>HF</sub> = 3.5 Hz, and a linewidth of 10.0 Hz (Figures 8.3 and 8.4).

When the reaction was conducted with <sup>13</sup>C-labeled methyl fluoride, the major signal in the <sup>1</sup>H-decoupled <sup>13</sup>C NMR spectrum (<sup>13</sup>C{<sup>1</sup>H}) was a triplet at  $\delta$  –28.20 ppm (<sup>2</sup>J<sub>PC</sub> = 4.4 Hz); without <sup>1</sup>H decoupling, this signal appeared as a quartet of triplets (<sup>1</sup>J<sub>HC</sub> = 136 Hz, <sup>2</sup>J<sub>PC</sub> = 4.4 Hz) due to coupling with three protons. The <sup>1</sup>H-<sup>13</sup>C coupling of 136 Hz was also manifest in the <sup>1</sup>H NMR signal at  $\delta$  1.85 ppm. The assignment of these signals to an iridium-bound methyl group derived from methyl fluoride is thus confirmed. The <sup>2</sup>J<sub>PC</sub> coupling was also manifest in the <sup>31</sup>P NMR signal, which appeared as a partially resolved doublet of doublets at  $\delta$  50.4 ppm that could be simulated (Spinworks) with coupling constants  ${}^{2}J_{\text{FP}} = 10.5$  Hz and  ${}^{2}J_{\text{PC}} = 4.4$  Hz, and a linewidth of 4.5 Hz (Figure 8.5). The results of elemental analysis and LIFDI (liquid introduction field desorption ionization) mass spectrometry<sup>25</sup> of **2** (Figure 8.6) were fully consistent with the characterization of **2** as (PCP)Ir(CH<sub>3</sub>)F.

We have previously shown that oxidative addition of the methyl–oxygen bond of anisole derivatives (H<sub>3</sub>C–OAr) occurs via initial addition of a methyl C-H bond to (PCP)Ir as evidenced by a significant normal CH<sub>3</sub>/CD<sub>3</sub> kinetic isotope effect (KIE).<sup>19</sup> When **1** was treated with a large excess of 3,5-bis(trifluoromethyl)benzylfluoride and its deuterated (CD<sub>2</sub>F) analogue (at least 5 fold excess of each), a CH<sub>2</sub>F/CD<sub>2</sub>F kinetic isotope effect of 2.7(3) at 60 °C was determined, implying that C–H bond activation is involved in or occurs prior to the product-determining step in the oxidative addition of the C(sp<sup>3</sup>)-F bond.<sup>18</sup> In analogy with the mechanism of C–O bond activation in anisole derivatives, we propose that initial C–H bond activation of methyl fluoride is followed by  $\alpha$ -fluorine migration<sup>26</sup> to generate an intermediate methylidene complex (**3**); iridium-to-methylidene hydride migration then completes the net C-F oxidative addition (Scheme 8.3).<sup>27</sup>




## 8.3 Conclusion

Considering the broader implications of these and previously reported results,<sup>18</sup> microscopic reversibility dictates that the lowest energy pathway for C–F elimination from **2** must proceed via initial  $\alpha$ -H migration from carbon to the metal center. This unusual mechanism may offer insight applicable toward the development of methods for the formation of sp<sup>3</sup>-C–F bonds via C–F elimination from a metal center; such reactions are of substantial current interest in the context of the synthesis of pharmaceutically important compounds.

## 8.4 Experimental

All operations were carried out under an argon atmosphere using standard Schlenk techniques or in an argon-filled dry box. Anhydrous pentane, *p*-xylene, *p*-xylene- $d_{10}$ , methyl t-butyl ether, and C<sub>6</sub>D<sub>6</sub> were dried over Na/K and collected by vacuum transfer. Norbornene (NBE) was purified by sublimation. All NMR experiments were performed using 400-MHz or 500-MHz Varian instruments. The residual <sup>1</sup>H peak of the deuterated solvent was used as a reference for <sup>1</sup>H chemical shifts. <sup>31</sup>P NMR and <sup>19</sup>F NMR chemical shifts were referenced to external H<sub>3</sub>PO<sub>4</sub> and CFCl<sub>3</sub> standards, respectively. Simulations of NMR spectra were performed using the NUMMRIT simulation implementation<sup>28</sup> within the Spinworks software program, version 3.1.7.<sup>29</sup> Elemental analysis data were obtained by the CENTC Elemental Analysis Facility at the University of Rochester using a Perkin Elmer 2400 Series II combustion analyzer. Liquid introduction field desorption ionization mass spectrometry (LIFDI) data were obtained by the High Resolution Mass Spectrometry Facility and the University of California at Riverside using a Waters GCT (2008) high resolution mass spectrometer. To remove moisture, the glass surface of J-Young NMR tubes was treated as described by Hughes *et al.*<sup>30</sup> (PCP)IrH<sub>n</sub> (PCP =  $\kappa^3$ -2,6-(<sup>1</sup>Bu<sub>2</sub>PCH<sub>2</sub>)<sub>2</sub>C<sub>6</sub>H<sub>3</sub>, n = 2 or 4) was prepared as previously described.<sup>31</sup> Fluoromethane was obtained from SynQuest Laboratories and Specialty Gases of America Inc. and was used without further purification.

(PCP)Ir(Me)(F), 2: On a high-vacuum line, gaseous CH<sub>3</sub>F (1 atm) was added to a J-Young tube containing a *p*-xylene solution (0.7 mL) of (PCP)Ir(NBE) (1) which had been generated from (PCP)IrH<sub>n</sub> (15.1 mg, 0.026 mmol) and NBE (7.2 mg, 0.076 mol). After 20 min at 50 °C, <sup>31</sup>P NMR spectroscopy revealed the disappearance of **1** and appearance of one major product, **2**. After the solvent and organic volatiles were removed in vacuo, a dark red solid (**2**) (15.5 mg) was obtained in 97% yield. NMR data for **2**: <sup>31</sup>P{<sup>1</sup>H} NMR (C<sub>6</sub>D<sub>6</sub>, 202 MHz):  $\delta$  50.38 (d, <sup>2</sup>*J*<sub>FP</sub> = 10.5 Hz). <sup>1</sup>H NMR (C<sub>6</sub>D<sub>6</sub>, 500 MHz):  $\delta$  7.03 (d, *J*<sub>HH</sub> = 7.5 Hz, 2H, PCP), 6.87 (t, *J*<sub>HH</sub> = 7.5 Hz, 1H, PCP), 2.99 (d of vt, *J*<sub>HH</sub> = 17.5 Hz, 2H, C*H*<sub>2</sub>P), 2.86 (d of vt, *J*<sub>PH</sub> = 4.1 Hz, *J*<sub>HH</sub>=17.5 Hz, 2H, C*H*<sub>2</sub>P), 1.85 (dt, *J*<sub>PH</sub> = 4.7 Hz, *J*<sub>FH</sub> = 3.5 Hz, 3H, Ir–CH<sub>3</sub>), 1.23 (vt,  $J_{PH} = 6.3$  Hz, 18H, PC(CH<sub>3</sub>)<sub>3</sub>), 1.20 (t,  $J_{PH} = 6.0$  Hz, 18H, PC(CH<sub>3</sub>)<sub>3</sub>). <sup>13</sup>C{<sup>1</sup>H} NMR (C<sub>6</sub>D<sub>6</sub>, 125 MHz):  $\delta$  150.67 (t,  $J_{PC} = 7.6$  Hz, PCP), 145.45 (d,  $J_{FC} = 39.4$  Hz, PCP), 121.74 (t,  $J_{PC} = 7.7$  Hz, PCP), 121.64 (s, PCP), 37.30 (vt, <sup>1</sup> $J_{PC} = 10.4$  Hz, PC(CH<sub>3</sub>)<sub>3</sub>), 34.73 (vt, <sup>1</sup> $J_{PC} = 8.8$  Hz, PC(CH<sub>3</sub>)<sub>3</sub>), 31.91 (vt, <sup>1</sup> $J_{PC} = 12.3$  Hz, CH<sub>2</sub>P), 30.64 (s, PC(CH<sub>3</sub>)<sub>3</sub>), 29.47 (vt, <sup>2</sup> $J_{PC} = 2.3$  Hz, PC(CH<sub>3</sub>)<sub>3</sub>), -28.20 (t, <sup>2</sup> $J_{PC} = 4.4$  Hz, Ir–CH<sub>3</sub>). <sup>19</sup>F NMR (C<sub>6</sub>D<sub>6</sub>, 470 MHz):  $\delta$  -252.6 (app s, Ir–F). LIFDI MS m/z (%): 620 (100) [M<sup>+</sup>], 618 (65) [M<sup>+</sup>]. Elemental analysis, calcd for C<sub>25</sub>H<sub>46</sub>FIrP<sub>2</sub>: C, 48.45; H, 7.48. Found: C, 48.51; 7.43.

**Figure 8.1.** <sup>1</sup>H NMR signal of **2** (Ir- $CH_3$ ) with no decoupling (red) and with <sup>31</sup>P decoupling (black).



1.0-╡dyw-022411-1-178-ccc-deccuple-19FobserveP31decouple 0.9-0.8-0.7 0.6-Normalized Intensity 0.5 0.4 0.3 0.2 0.1 0 -252.60 Chemical Shift (ppm) -252.70 252.65 252.45 2 52 .50 252.55 252.75

**Figure 8.3.** <sup>19</sup>F NMR signal of **2**: simulated with Spinworks software (top) and observed NMR data (bottom). Simulation parameters:  ${}^{2}J_{PF} = 10.5$  Hz;  ${}^{3}J_{HF} = 3.5$  Hz; linewidth = 10.0 Hz.



**Figure 8.2.** <sup>19</sup>F NMR signal of **2** without decoupling (red) and with <sup>31</sup>P decoupling (black).

**Figure 8.4.** <sup>19</sup>F{<sup>31</sup>P} NMR signal of **2**: simulated with Spinworks software (top) and observed NMR data (bottom). Simulation parameters:  ${}^{3}J_{HF} = 3.5$  Hz; linewidth = 10.0 Hz.



**Figure 8.5.** <sup>31</sup>P NMR signal of <sup>13</sup>C-labeled **2**, synthesized from the reaction with <sup>13</sup>CH<sub>3</sub>F, simulated with Spinworks software (top) and observed NMR data (bottom). Simulation parameters:  ${}^{2}J_{FP} = 10.5$  Hz;  ${}^{2}J_{PC} = 4.4$  Hz; linewidth = 4.5 Hz.



**Figure 8.6.** LIFDI-MS spectrum of **2**. The cluster of peaks with M = 618-622 shows the mass distribution expected for (PCP)Ir(CH<sub>3</sub>)F (**2**). The clusters with M = 612-616 and M = 604-608 are consistent with (PCP)Ir(N<sub>2</sub>) and (PCP)Ir(H)(OH) impurities respectively, possibly introduced during sample handling.



## References

- (1) Collman, J. P. Acc. Chem. Res. **1968**, *1*, 136.
- (2) Halpern, J. Acc. Chem. Res. 1970, 3, 386.
- (3) Stille, J. K.; Lau, K. S. Y. Acc. Chem. Res. 1977, 10, 434.
- (4) *Metal-Catalyzed Cross-Coupling Reactions*; de Meijere, A.; Diederich, F., Eds.; Wiley-VCH: Weinheim, 2004.
- (5) Vigalok, A. Chem. Eur. J. 2008, 14, 5102.
- (6) Hartwig, J. F. *Organotransition Metal Chemistry*; University Science Books: Sausalito, CA, 2010.
- (7) Amii, H.; Uneyama, K. Chem. Rev. 2009, 109, 2119.
- (8) Muller, K.; Faeh, C.; Diederich, F. *Science* **2007**, *317*, 1881.
- (9) McCulloch, A. J. Fluor. Chem. 2003, 123, 21.
- (10) Hughes, R. P. Eur. J. Inorg. Chem. 2009, 4591.
- Watson, D. A.; Su, M.; Teverovskiy, G.; Zhang, Y.; Garcia-Fortanet, J.; Kinzel, T.; Buchwald, S. L. *Science* 2009, *325*, 1661.
- (12) Cho, E. J.; Senecal, T. D.; Kinzel, T.; Zhang, Y.; Watson, D. A.; Buchwald, S. L. *Science* **2010**, *328*, 1679.
- (13) Douvris, C.; Ozerov, O. V. *Science* **2008**, *321*, 1188.
- (14) Furuya, T.; Benitez, D.; Tkatchouk, E.; Strom, A. E.; Tang, P.; Goddard, W. A.; Ritter, T. J. Am. Chem. Soc. **2010**, 132, 3793.
- (15) Kiplinger, J. L.; Richmond, T. G.; Osterberg, C. E. Chem. Rev. 1994, 94, 373.
- (16) Torrens, H. Coord. Chem. Rev. 2005, 249, 1957.
- (17) Perutz, R. N.; Braun, T. In *Comprehensive Organometallic Chemistry III* Crabtree, R. H., Mingos, D. M. P., Eds.; Elsevier: Oxford, 2007; 1, pp 725.
- (18) Choi, J.; Wang, D. Y.; Kundu, S.; Choliy, Y.; Emge, T. J.; Krogh-Jespersen, K.; Goldman, A. S. *Science* **2011**, *332*, 1545.
- (19) Choi, J.; Choliy, Y.; Zhang, X.; Emge, T. J.; Krogh-Jespersen, K.; Goldman, A. S. *J. Am. Chem. Soc.* **2009**, *131*, 15627.
- (20) McMillen, D. F.; Golden, D. M. Annu. Rev. Phys. Chem. 1982, 33, 493.
- (21) Kanzelberger, M.; Singh, B.; Czerw, M.; Krogh-Jespersen, K.; Goldman, A. S. J. *Am. Chem. Soc.* **2000**, *122*, 11017.
- (22) Rybtchinski, B.; Vigalok, A.; Bendavid, Y.; Milstein, D. J. Am. Chem. Soc. 1996, 118, 12406.
- (23) Cooper, A. C.; Folting, K.; Huffman, J. C.; Caulton, K. G. *Organometallics* **1997**, *16*, 505.
- (24) Fawcett, J.; Harding, D. A. J.; Hope, E. G. Dalton Trans. 2010, 39, 5827.
- (25) Dransfield, T. A.; Nazir, R.; Perutz, R. N.; Whitwood, A. C. J. Fluor. Chem. **2010**, *131*, 1213.
- (26) Huang, D.; Koren, P. R.; Folting, K.; Davidson, E. R.; Caulton, K. G. J. Am. *Chem. Soc.* **2000**, *122*, 8916.
- (27) Werkema, E. L.; Messines, E.; Perrin, L.; Maron, L.; Eisenstein, O.; Andersen, R. A. J. Am. Chem. Soc. 2005, 127, 7781.
- (28) Quirt, A. R.; Martin, J. S. J. Magn. Reson. 1971, 5, 318.
- (29) K. Marat, Spinworks software program, version 3.1.7, University of Manitoba (2010); available at <u>www.umanitoba.ca/chemistry/nmr/spinworks</u>.

- (30) Hughes, R. P.; Rose, P. R.; Rheingold, A. L. Organometallics 1993, 12, 3109.
- (31) Goldman, A. S.; Ghosh, R. In *Handbook of C-H Transformations Applications in Organic Synthesis*; Dyker, G., Ed.; Wiley-VCH: New York, 2005.